An alternative subtraction scheme using Nagy-Soper dipoles

Von der Fakultät für Mathematik, Informatik und Naturwissenschaften der Rheinisch-Westfälischen Technischen Hochschule Aachen zur Erlangung des akademischen Grades eines Doktors der Naturwissenschaften genehmigte Dissertation

vorgelegt von

M.Sc. in Physics

Cheng Han Chung

aus Zhubei, Taiwan

Referent: Universitätsprofessor Michael Krämer Korreferent: Universitätsprofessor Michal Czakon

Tag der mündlichen Prüfung: 21 Juni 2011

Diese Dissertation ist auf den Internetseiten der Hochschulbibliothek online verfügbar.

Abstract

In this thesis we address an alternative subtraction scheme in high energy colliders at Nextto-Leading Order (NLO) QCD calculations. In particular, we focus on the treatment of real radiation contributions in the soft and collinear limits. After UV-renormalization, the remaining infrared singularities appearing both in the real radiation and in the virtual contributions can be regularized using dipole subtraction method. In this scheme, dipoles are based on the momentum mapping and on the splitting functions derived from an improved parton shower formulation with quantum interference effects. In our new scheme, we employ a slightly altered momentum mapping such that the number of subtraction terms is greatly reduced in comparison with the standard Catani-Seymour scheme. In addition, the new scheme also facilitates the matching of NLO calculations with parton showers using the same splitting functions. We also achieve the complete integrations of the splitting functions over an unresolved one parton phase space, obtaining the correct soft and collinear singularity structures that are necessary to cancel the soft divergences in the virtual contributions. We discuss the general framework setup of the scheme as well as some scattering processes at colliders; we find complete agreement with the results in the widely used Catani-Seymour dipole subtraction scheme.

Contents

1	A brief review of QCD					
	1.1	Introdu	ction	6		
	1.2	Colour	$SU(3)_C$ and quark confinement \ldots \ldots \ldots \ldots \ldots \ldots	8		
	1.3	QCD I	agrangian	9		
	1.4	gauge invariance	13			
	1.5	5 Renormalization				
	1.6	Parton	branching at Next-to-Leading Order	18		
2	Nagy-Soper dipoles					
	2.1	.1 Motivation		22		
	2.2	2.2 General structure of the NLO cross section and subtraction procedure				
		2.2.1	The general subtraction procedure	25		
		2.2.2	Generalization to hadron collisions	28		
		2.2.3	Observable-dependent formulation of the subtraction method	30		
	2.3	3 Scheme setup and momentum mapping				
		2.3.1	Splitting a final state parton	32		
		2.3.2	Splitting an initial state parton	36		
	2.4	ng functions	40			
		2.4.1	Definition of the splitting amplitudes	40		

		2.4.2	Eikonal factor	43
		2.4.3	Collinear splitting functions: final state splittings	47
		2.4.4	Collinear splitting functions: initial state splittings	52
		2.4.5	Soft splitting functions	56
	2.5	5 Integrated splitting functions		
		2.5.1	Collinear integrals: final state splittings	60
		2.5.2	Collinear integrals: initial state splittings	61
		2.5.3	Interference between initial states	64
		2.5.4	Interference between initial and final states	65
		2.5.5	Interference between final (final and initial) states	66
2	Ann	lication		71
3	App			71
	3.1	Single W production		71
		3.1.1	Tree level	71
		3.1.2	Real emission, virtual correction and dipole subtraction	72
	3.2	Dijet production in e^+e^- annihilation $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$		
3.3 Higgs production in gluon-gluon fusion: $gg \rightarrow H$			production in gluon-gluon fusion: $gg o H$	85
		3.3.1	The subprocess $q ar q o g H$	89
		3.3.2	The subprocess $qg ightarrow qH$	91
		3.3.3	The subprocess $gg \to gH$	93
		3.3.4	One-loop virtual corrections	95
		3.3.5	Proof: Nagy-Soper scheme and Catani-Seymour scheme	98
		3.3.6	Proof: Catani-Seymour scheme and literature results	100
	3.4 Higgs decay: $H \rightarrow gg$			101
		3.4.1	The subprocess $H o g q ar q$	102
		3.4.2	The subprocess $H ightarrow ggg$	103

4 Conclusions

5	App	ppendix 1				
	5.1	Useful	mathematical formulae	111		
		5.1.1	Gamma function, Beta function and Hypergeometric function	111		
		5.1.2	Dilogarithm function	112		
		5.1.3	The volume element in <i>d</i> dimensions	113		
	5.2 Integration measures					
		5.2.1	Final state splitting	114		
		5.2.2	Initial state splitting	117		
		5.2.3	Master integrals	119		
		5.2.4	Pole extractions	120		
	5.3	Colour	algebra	120		
	5.4	Phase s	space integration	122		
	5.5	Soft ph	oton radiation	129		
5.6 Collinear photon radiation			ear photon radiation	131		
		5.6.1	Final state radiation	131		
		5.6.2	Initial state radiation	134		
	5.7 One-loop calculations: examples			134		
	5.8 A review of the Standard Model (SM)		w of the Standard Model (SM)	140		
		5.8.1	Abelian gauge theory: Quantum electrodynamics (QED)	140		
		5.8.2	Non-abelian gauge theory	141		
		5.8.3	Electroweak theory	144		
		5.8.4	The Standard Model of particle physics	156		

108

Chapter 1

A brief review of QCD

1.1 Introduction

In elementary particle physics, the fundamental theory of the strong interaction is described by Quantum Chromodynamics (QCD). It describes the interactions between quarks and gluons, and in particular how they bind together to form hadrons (*e.g.* the proton and the neutron). QCD is a quantum field theory within a special class described by non-abelian gauge theory [1] (or sometimes called Yang-Mills gauge theory). It is based on the SU(N) gauge group. Understanding how to use non-abelian gauge theory, combined with the parton model, led to the development of QCD and it is now a very well established theory in the sense that QCD predictions have successfully accounted for all the strong interaction experiments observed at colliders, in particular the phenomena of hadronic jet structure in e^+e^- annihilation, the Drell-Yan process and heavy quark production [100]. QCD has two peculiar properties, which differ from Quantum Electrodynamics (QED)/electroweak interactions and which also reveal its uniqueness.

Asymptotic freedom: this means that at very high energies, the strong force (also called the colour force) of quarks and gluons is so weak that they behave almost as free particles when the quarks or gluons are really close to each other. This phenomenon is called asymptotic freedom and it is due to the fact that the strong running coupling constant α_s(Q²) depends on the energy scale Q; α_s(Q²) becomes weaker as the scale Q increases¹. To check this, one must determine the running of the coupling constant α_s, which is governed by the renormalization group equation,

$$\beta(\alpha_s) = Q^2 \frac{\partial \alpha_s}{\partial Q^2}$$

¹In contrast to QED where the coupling α becomes strong at high energies.

if $\beta < 0$, the theory is asymptotically free. The asymptotic freedom of QCD also explains why we can apply a perturbative approach to explore the structure of matter at short distances (= high energies). This prediction of QCD was first discovered in the early 1970s by H. David Politzer [2] and by David J. Gross and Frank Wilczek [3]. For this work they were awarded the 2004 Nobel Prize in Physics.

• **Colour confinement:** the strong force of quarks becomes stronger when the distance increases, which implies low energy. So it would take an infinite amount of energy to move apart two quarks; they are always confined inside hadrons. Confinement is widely believed to be true (although analytically unproven yet; a lot of QCD theorems are based on assumptions) in the sense that no free quark and gluon degrees of freedom have been observed at colliders².

The recent progress in the understanding of strong interactions has been due to the comparison between precise higher order perturbative QCD calculations and accurate experimental data. The perturbative calculation of jet cross sections is based on the QCD improved parton model picture [4–6], which has been made rigorous since the discovery of asymptotic freedom. In this model a hard scattering process between two hadrons can be thought of as an interaction between the quarks and gluons, which are the constituents of the incoming hadrons. Much of the techniques of perturbative QCD derive from the well known methods of QED apart from the fact that QCD is a non-abelian gauge theory; however, there are still big differences between the two theories.

Importantly because the quanta of QCD, quarks and gluons, the analogues of electrons and photons in QED, are always bound into hadrons and not observed as free particles at colliders. At low energies, confinement effects dominate and non-perturbative approaches become more important. The most widely used method is lattice QCD. At very high energies, however, one still cannot avoid confinement effects due to the fact that the asymptotic incoming and outgoing partons consist of hadrons. For certain quantities, factorization theorems [7–9] allow the two scales to be appropriately separated, and the low energy pieces can be treated by parametrizations, model calculations or factoring them into the parton distribution functions (PDF). The remaining quantities involve only high momentum transfers (and therefore short distances and short times) and is insensitive to long distances behaviour of QCD. Thus, these quantities are calculable in perturbation theory because of asymptotic freedom. The factorization theorem states that the short distance behaviour of parton scattering (the hard part) does not interfere with the long distance process that turns partons (quarks and gluons) into hadrons, hadronization. This factorization property can be proved to be valid to all orders in perturbation theory.

²Note, that all hadrons are colourless (or colour singlets) and only colour singlet states can be observed as free particles; *i.e.* we never observe a free quark/gluon since quark/gluon carries colour charge.

Another important ingredient of perturbative QCD is infrared safety, which is the guiding principle of higher order perturbative calculations. In general, we consider any quantity that is infrared finite. That is to say, infrared safe quantities do not depend on the long distance behaviour of QCD (except for the long distance effects that are factored into the parton distribution functions), so that it makes sense to calculate any quantity in perturbative QCD. The proof of infrared safety comes from the KLN theorem [10, 11], which is a fundamental quantum mechanical result and which provides the theoretical argument to the problem of collinear and infrared singularities due to massless charged particles. It states that fully inclusive measurements, which sum over all degenerate initial and final states, are free from infrared divergences.

In this thesis we will only deal with the physics of hard processes at Next-to-Leading Order precision, in which case the infrared divergences have to be treated carefully. The properties of any given hadron species will be irrelevant because of factorization theorems. This is why jet production is computed as simple parton scattering. The probability that partons will produce hadrons is unity.

1.2 Colour $SU(3)_C$ and quark confinement

First, let us review the addition of angular momenta in Quantum Mechanics. Addition of two spin-1/2 particles $j_A = j_B = 1/2$ has total spin J = 0 or J = 1. Symbolically, we have³

$$\mathbf{2}\otimes\mathbf{2} = \mathbf{1}\oplus\mathbf{3} \tag{1.1}$$

Now, combining a third spin-1/2 particle, we have

$$(\mathbf{2} \otimes \mathbf{2}) \otimes \mathbf{2} = (\mathbf{1} \oplus \mathbf{3}) \otimes \mathbf{2} = (\mathbf{1} \otimes \mathbf{2}) \oplus (\mathbf{3} \otimes \mathbf{2}) = \mathbf{2} \oplus \mathbf{2} \oplus \mathbf{4}$$
(1.2)

At the end, we have a quartet of spin 3/2 and two doublets of spin 1/2. The quarks in the spin-3/2 baryons are in a symmetrical state of spin, space and $SU(3)_f$ flavour degrees of freedom, e.g. we consider the pion-nucleon resonance⁴ Δ^{++} with spin-3/2

$$\left|\Delta^{++}, J_3 = 3/2\right\rangle = \left|u\uparrow, u\uparrow, u\uparrow\right\rangle \tag{1.3}$$

Here, J_3 is the third component of the total angular momentum for Δ^{++} and the arrow represents the spin aligned up. However, this state is not allowed because the wave function has to be totally antisymmetric under interchange of any of the two quarks due to the Fermi-Dirac statistics. To

³Here, we use the dimension (*i.e.* the size of multiplet 2J + 1) to label the irreducible representation.

 $^{{}^{4}\}Delta^{++}$ is made of three *u*-quarks.

reconcile the baryon spectrum and the Fermi-Dirac statistics, one can introduce an additional quantum number, called colour [121, 122]. Hence, we can construct colour singlet states

$$|M\rangle \sim |q_a \bar{q}_a
angle \quad \text{or} \quad |M\rangle = \frac{1}{\sqrt{3}} \delta^{ab} |q_a \bar{q}_b
angle \quad \text{and} \quad |B\rangle \sim \epsilon^{abc} |q_a q_b q_c
angle$$

for mesons⁵ and baryons⁶, respectively, where ϵ^{abc} is the totally antisymmetric tensor and a, b, c (= 1, 2, 3) represent three colours of quarks. It is then easy to construct the totally antisymmetric wave function for Δ^{++}

$$\left| \Delta^{++}, J_3 = 3/2 \right\rangle = \epsilon_{abc} \left| u^a \uparrow, u^b \uparrow, u^c \uparrow \right\rangle$$
(1.4)

The state Eq. (1.4) is then a singlet 1 representation of $SU(3)_C$. Next, we consider the confinement effect in QCD. Quark confinement is directly related to the fact that quarks (gluons) are coloured quanta and hence cannot be observed in nature. All physical hadrons are colourless. In order to construct colour singlet states we have to pick out a singlet representation in the decomposition of the product of two/three quarks into irreducible representation. For meson state, we have⁷

$$\mathbf{3} \otimes \bar{\mathbf{3}} = \mathbf{1} \oplus \mathbf{8} \tag{1.5}$$

For baryon state, we have

$$\mathbf{3} \otimes \mathbf{3} \otimes \mathbf{3} = (\,\bar{\mathbf{3}} \oplus \mathbf{6}\,) \otimes \mathbf{3} = (\,\bar{\mathbf{3}} \otimes \mathbf{3}\,) \oplus (\,\mathbf{6} \otimes \mathbf{3}\,) = \mathbf{1} \oplus \mathbf{8} \oplus \mathbf{8} \oplus \mathbf{10} \tag{1.6}$$

Diquark and four-quark states belong to colour nonsinglets:

$$|qq>: \mathbf{3}\otimes\mathbf{3} = \mathbf{\bar{3}}\oplus\mathbf{6}$$

$$|qqqq>: \mathbf{3}\otimes\mathbf{3}\otimes\mathbf{3}\otimes\mathbf{3} = \mathbf{3}\oplus\mathbf{3}\oplus\mathbf{3}\oplus\mathbf{\bar{6}}\oplus\mathbf{\bar{6}}\oplus\mathbf{15}\oplus\mathbf{15}\oplus\mathbf{15}\oplus\mathbf{15}\oplus\mathbf{15}'$$

$$|\bar{q}\bar{q}\bar{q}>: \mathbf{\bar{3}}\otimes\mathbf{\bar{3}} = \mathbf{3}\oplus\mathbf{\bar{6}}$$

$$|\bar{q}\bar{q}\bar{q}\bar{q}\bar{q}>: \mathbf{\bar{3}}\otimes\mathbf{\bar{3}}\otimes\mathbf{\bar{3}}\otimes\mathbf{\bar{3}} = \mathbf{\bar{3}}\oplus\mathbf{\bar{3}}\oplus\mathbf{\bar{3}}\oplus\mathbf{\bar{6}}\oplus\mathbf{6}\oplus\mathbf{\overline{15}}\oplus\mathbf{\overline{15}}\oplus\mathbf{\overline{15}}\oplus\mathbf{\overline{15}}\oplus\mathbf{\overline{15}}'$$

$$(1.7)$$

Only $|q\bar{q}\rangle$ and $|qqq\rangle$ states belong to colour singlets. The conjecture that only colour singlet states can be observed is the same as that of the quark confinement.

1.3 QCD Lagrangian

Strong interactions between quarks and gluons are described by non-abelian local gauge theory and $SU(3)_C$ is the gauge group. Each quark field (flavour) forms a triplet in the fundamental

⁵Mesons are quark-antiquark bound states.

⁶Baryons are bound states of three quarks (antiquarks).

⁷Quark field belongs to the fundamental **3** representation of $SU(3)_C$ and antiquark is then assigned to the complex-conjugate $\bar{\mathbf{3}}$ representation. The **6** is symmetric and $\bar{\mathbf{3}}$ is antisymmetric under interchange of the two quarks.

representation of $SU(3)_C$

$$q_a = \begin{pmatrix} q_{\text{red}} \\ q_{\text{blue}} \\ q_{\text{green}} \end{pmatrix} , \qquad (a = 1, 2, 3)$$
(1.8)

and eight gluon fields G^A_{μ} form an octet in the adjoint representation (defined to have the same dimensions as the gauge group). The index A runs over the eight colour degrees of freedom of the gluon field ($A = 1, \dots, 8$). The QCD Lagrangian density is given by

$$\mathcal{L}_{\text{QCD}} = \mathcal{L}_{\text{classical}} + \mathcal{L}_{\text{gauge-fixing}} + \mathcal{L}_{\text{ghost}}$$
(1.9)

where the classical Lagrangian density is

$$\mathcal{L}_{\text{classical}} = -\frac{1}{4} G^A_{\mu\nu} G^{\mu\nu}_A + \sum_{\text{flavours}} \bar{q}_a \left(i \not\!\!D - m \right)_{ab} q_b \tag{1.10}$$

which describes the interaction of spin- $\frac{1}{2}$ quarks of mass m and massless spin-1 gluons. Din Eq. (1.10) is a symbolic notation for $\gamma_{\mu}D^{\mu}$ and the spinor indices of γ_{μ} and q_a have been suppressed. The sum in Eq. (1.10) runs over the n_f different flavours of quarks (= u, d, c, s, t, b). We follow the standard notation with metric given by $g^{\mu\nu} = \text{diag}(1, -1, -1, -1)$ and set $\hbar = c = 1$. The gamma matrices satisfy the Clifford algebra

$$\{\gamma^{\mu}, \gamma^{\nu}\} = 2 g^{\mu\nu} \tag{1.11}$$

 $G^{A}_{\mu\nu}$ is the field strength tensor, which can be derived from the gluon field $G^{A}_{\mu\nu}$

$$G^{A}_{\mu\nu} = \partial_{\mu}G^{A}_{\nu} - \partial_{\nu}G^{A}_{\mu} - gf^{ABC}G^{B}_{\mu}G^{C}_{\nu} , \qquad (A, B, C = 1, \cdots, 8)$$
(1.12)

where g is the strength of the strong coupling constant⁸ between coloured quanta (quarks and gluons). It is the third term on the right-hand-side of Eq. (1.12) that gives rise to cubic and quartic gauge boson (gluon) self-interactions⁹. Note that the mass terms $m^2 G^A_{\mu} G^{\mu}_A$ for the gauge bosons (gluons) are not gauge invariant! Gauge bosons of unbroken non-abelian gauge theory are massless. Gauge invariance combined with renormalizability (absence of higher powers of fields and covariant derivatives in Lagrangian) determines gauge boson/fermion couplings and gauge boson self-interactions. In order to preserve the renormalizability of QCD, each term in the Lagrangian has to have mass dimension four. It follows, that the dimensions of the fields q_a and G^A_{μ} are 3/2 and 1, respectively. f^{ABC} are the structure constants of the $SU(3)_C$ colour group. D is the covariant derivative, which acts on triplet and octet fields according to

$$(D_{\mu})_{ab} = \partial_{\mu} \delta_{ab} + ig \left(t^{C} G_{\mu}^{C} \right)_{ab}$$

$$(D_{\mu})_{AB} = \partial_{\mu} \delta_{AB} + ig \left(T^{C} G_{\mu}^{C} \right)_{AB}$$

$$(1.13)$$

⁸Note that the notations g and g_s can be exchanged with each other without making any difference in the later discussions.

⁹This term leads to the property of asymptotic freedom at the end.

where t and T are matrices in the fundamental and adjoint representations of $SU(3)_C$ colour group respectively.

$$\begin{bmatrix} t^{A}, t^{B} \end{bmatrix} = i f^{ABC} t^{C}$$

$$\begin{bmatrix} T^{A}, T^{B} \end{bmatrix} = i f^{ABC} T^{C}$$

$$\begin{pmatrix} T^{A} \end{pmatrix}_{BC} = -i f^{ABC}$$
(1.14)

The generators t^A can be represented by the eight Gell-Mann matrices. These matrices are hermitian and traceless,

$$t^A = \frac{1}{2}\lambda^A \tag{1.15}$$

with

$$\lambda^{1} = \begin{pmatrix} 0 & 1 & 0 \\ 1 & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}, \quad \lambda^{2} = \begin{pmatrix} 0 & -i & 0 \\ i & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}, \quad \lambda^{3} = \begin{pmatrix} 1 & 0 & 0 \\ 0 & -1 & 0 \\ 0 & 0 & 0 \end{pmatrix},$$
$$\lambda^{4} = \begin{pmatrix} 0 & 0 & 1 \\ 0 & 0 & 0 \\ 1 & 0 & 0 \end{pmatrix}, \quad \lambda^{5} = \begin{pmatrix} 0 & 0 & -i \\ 0 & 0 & 0 \\ i & 0 & 0 \end{pmatrix}, \quad \lambda^{6} = \begin{pmatrix} 0 & 0 & 0 \\ 0 & 0 & 1 \\ 0 & 1 & 0 \end{pmatrix},$$
$$\lambda^{7} = \begin{pmatrix} 0 & 0 & 0 \\ 0 & 0 & -i \\ 0 & i & 0 \end{pmatrix}, \quad \lambda^{8} = \frac{1}{\sqrt{3}} \begin{pmatrix} 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & -2 \end{pmatrix}$$
(1.16)

The colour charges of the SU(N) matrices can be chosen to be (see Fig. 1.1)

$$\operatorname{Tr} t^{A} t^{B} = T_{R} \delta^{AB}, \quad T_{R} = \frac{1}{2}, \quad \sum_{A} t^{A}_{ab} t^{A}_{bc} = C_{F} \delta_{ac}, \quad C_{F} = \frac{N^{2} - 1}{2N}$$
$$\operatorname{Tr} T^{C} T^{D} = \sum_{A,B} f^{ABC} f^{ABD} = C_{A} \delta^{CD}, \quad C_{A} = N$$
(1.17)

The colour charge is directly related to the Casimir operator $C_i = T^2 = T^A T^A$ where, $(T^A)_{BC} \equiv -i f^{ABC}$ if *i* is a gluon and $T^A_{ab} \equiv t^A_{ab} (-t^A_{ba})$ if *i* is a quark (antiquark). The Casimir operator commutes with all group generators, *i.e.*

$$\left[T^B, T^A T^A\right] = i f^{BAC} T^C T^A + i f^{BAC} T^A T^C = i \underbrace{f^{BAC}}_{\text{antisymmetric}} \underbrace{\left\{T^C, T^A\right\}}_{\text{symmetric}} = 0 \qquad (1.18)$$

Hence, T^2 is an invariant of the algebra. For the specific case of $SU(3)_C$ we have

$$C_F = \frac{4}{3}, \qquad C_A = 3$$
 (1.19)

For the anticommutator of the t matrices in the fundamental representation, we have

$$\{t^{A}, t^{B}\} = \frac{1}{N} \delta^{AB} I + d^{ABC} t^{C} , \quad \sum_{A,B} d^{ABC} d^{ABD} = \frac{N^{2} - 4}{N} \delta^{CD} , \quad d^{AAC} \equiv 0 \quad (1.20)$$



Figure 1.1: The definitions of the colour charges in the SU(N) gauge group. Repeated indices are summed over the $N^2 - 1(N)$ values of A, B, C(a, b, c) of the adjoint (fundamental) representation. Here, the curl line means gluon field and solid line means quark field. We can also define $C_i = C_F = (N^2 - 1)/(2N)$ if *i* is a quark or antiquark and $C_i = C_A = N$ if *i* is a gluon.

1.4 Local gauge invariance

Eq. (1.10) is invariant under local gauge transformations. This means the parameters $\theta^A(x)$ which characterize the transformation depend on the spacetime coordinates and we can redefine the quark fields independently at every spacetime point, without changing the physics.

$$q_a(x) \to q'_a(x) = \Omega(x)_{ab} q_b(x), \quad \Omega(x) = \exp\{i t^A \theta^A(x)\}, \quad \Omega^{\dagger} \Omega = \Omega \Omega^{\dagger} = 1$$
(1.21)

If Ω is unitary, the t^A are hermitian matrices, called group generators. They generate infinitesimal transformations around the unit matrix element of the Lie group. For SU(N) matrix (unitary and det $\Omega = 1$), there are $N^2 - 1$ traceless, hermitian generators $t^A = 1/2 \lambda^A$.

$$\Omega(x) = 1 + i t^A \theta^A(x) + \mathcal{O}(\theta^2)$$
(1.22)

The covariant derivative transforms as

$$D_{\mu}q(x) \rightarrow D'_{\mu}q'(x) \equiv \Omega(x)D_{\mu}q(x)$$
(1.23)

Here we have omitted the colour labels of the quark fields. We can use Eq. (1.23) to derive the transformation property of the gluon field $G_{\mu}(x)$

$$D'_{\mu}q'(x) = (\partial_{\mu} + i g t \cdot G'_{\mu}) \Omega(x)q(x)$$

= $(\partial_{\mu} \Omega(x)) q(x) + \Omega(x)\partial_{\mu}q(x) + i g t \cdot G'_{\mu} \Omega(x)q(x)$ (1.24)

where

$$t \cdot G_{\mu} = \sum_{A} t^A G^A_{\mu} \tag{1.25}$$

Thus we find

$$t \cdot G'_{\mu} = \Omega(x) t \cdot G_{\mu} \Omega^{-1}(x) + \frac{i}{g} \left(\partial_{\mu} \Omega(x)\right) \Omega^{-1}(x)$$
(1.26)

which in terms of the infinitesimal parameters $\theta(x)$ can be rewritten as

$$G_{\mu}^{\prime A} = G_{\mu}^{A} - f^{ABC} \theta^{B} G_{\mu}^{C} - \frac{1}{g} \partial_{\mu} \theta^{A} + \mathcal{O}(\theta^{2})$$
(1.27)

The third term is similar to the abelian case. The second term is specific to the non-abelian gauge theory. Introducing the generators T^A in the adjoint representation of the gauge group, one may write the infinitesimal transformation as

$$\delta G^A_\mu = -\frac{1}{g} (D_\mu)_{AB} \,\theta^B, \qquad G'^A_\mu = G^A_\mu + \delta G^A_\mu$$
(1.28)

It is straightforward to show that the transformation property of the field strength tensor $G_{\mu\nu}$ is¹⁰

$$t \cdot G_{\mu\nu}(x) \to t \cdot G'_{\mu\nu}(x) = \Omega(x) t \cdot G_{\mu\nu}(x) \Omega^{-1}(x) ,$$
 (1.29)

which may be derived using the relation

$$[D_{\mu}, D_{\nu}] = i g t \cdot G_{\mu\nu} . \qquad (1.30)$$

1.5 Renormalization

The running coupling constant and renormalization group equation (RGE)

Suppose A is a dimensionless quantity which depends on a single energy scale Q. By assumption the scale Q is much bigger than another mass scale: $Q^2 \gg m^2$. In the limit $m \to 0$, A is independent of Q by dimensional analysis.

$$A = A(Q^2/m^2, \alpha_s) \longrightarrow A(\alpha_s) \text{ as } m \to 0$$
 (1.31)

After quantization, the theory must be renormalized due to the presence of ultraviolet (UV) divergences. Hence an arbitrary mass scale μ has to be introduced.

$$A \longrightarrow A(Q^2/\mu^2, \alpha_s)$$
 after quantization (1.32)

The scale μ is arbitrary, and physical results cannot depend on it. Mathematically, the μ independence of A may be expressed by

$$\mu^2 \frac{d}{d\mu^2} A(Q^2/\mu^2, \alpha_s) = \mu^2 \left(\frac{\partial}{\partial\mu^2} + \frac{\partial\alpha_s}{\partial\mu^2}\frac{\partial}{\partial\alpha_s}\right) A = 0$$
(1.33)

This is a renormalization group equation (RGE). In order to solve RGE, one defines

$$t = \ln Q^2 / \mu^2, \quad \beta(\alpha_s) = \mu^2 \frac{\partial \alpha_s}{\partial \mu^2}$$
 (1.34)

Using

$$\frac{\partial}{\partial \mu^2} = \frac{\partial t}{\partial \mu^2} \frac{\partial}{\partial t}$$
(1.35)

we have

$$\left(-\frac{\partial}{\partial t} + \beta(\alpha_s)\frac{\partial}{\partial\alpha_s}\right)A = 0$$
(1.36)

¹⁰In contrast to QED, the field strength tensor is not gauge invariant in QCD, because of the gluon selfinteractions. However, the trace $\text{Tr} G_{\mu\nu} G^{\mu\nu}$ is gauge invariant. Also, the gluon fields are coloured quanta, in contrast to QED, where the photons are electrically neutral.

The strong running coupling $\alpha_s(Q^2)$ is then introduced

$$t = \int_{\alpha_s}^{\alpha_s(Q^2)} \frac{dx}{\beta(x)}, \quad \alpha_s(\mu^2) \equiv \alpha_s$$
(1.37)

We can then take derivatives with respect to t and α_s (the two independent variables) on both sides of Eq. (1.37). By taking d/dt we obtain

$$1 = \frac{1}{\beta(\alpha_s(Q^2))} \frac{\partial \alpha_s(Q^2)}{\partial t}, \qquad \frac{\partial \alpha_s}{\partial t} = 0$$
(1.38)

By taking $d/d\alpha_s$ we obtain

$$0 = \frac{1}{\beta(\alpha_s(Q^2))} \frac{\partial \alpha_s(Q^2)}{\partial \alpha_s} - \frac{1}{\beta(\alpha_s)}$$
(1.39)

These two relations show explicit the dependence of the running coupling on t and α_s :

$$\frac{\partial \alpha_s(Q^2)}{\partial t} = \beta(\alpha_s(Q^2)), \quad \frac{\partial \alpha_s(Q^2)}{\partial \alpha_s} = \frac{\beta(\alpha_s(Q^2))}{\beta(\alpha_s)}$$
(1.40)

from which it follows that

$$A(Q^2/\mu^2, \alpha_s) = A(1, \alpha_s(Q^2))$$
(1.41)

is a solution of Eq. (1.36). Thus, the scale dependence of A is known if that of $\alpha_s(Q^2)$ is known.

The β function

Instead of discussing different UV renormalization schemes, let us compute a simple renormalization scale dependent parameter: the running strong coupling α_s . In QCD, the differential equation for the strong coupling α_s is

$$\beta(\alpha_s) = Q^2 \frac{\partial \alpha_s}{\partial Q^2} \tag{1.42}$$

Thus α_s is a function of the energy scale at which it is evaluated and runs according to the β function, which at the one-loop level is given by

$$\beta(\alpha_s) = -\frac{1}{2\pi}\beta_0 \alpha_s^2 + \cdots, \qquad \beta_0 = \frac{11}{6}C_A - \frac{2}{3}n_f T_R$$
(1.43)

where (\cdots) represents the terms beyond the one-loop level. In the following we will explain in detail how Eq. (1.43) is computed. Let us consider bottom pair production at the LHC: $q\bar{q} \rightarrow b\bar{b}$. The Feynman diagram is an *s*-channel off-shell gluon. The physical parameters we can



Figure 1.2: The one-loop contributions to the renormalization counterterms for the gluon and fermion self-energies and the gluon-fermion vertex. The curl line means gluon field, solid line means quark field and dash line means ghost field.

renormalize in this process are the strong coupling α_s and the bottom quark mass. Wave function renormalization constants are not physical. Here we assume that all quarks are massless. To compute the β function, one has to calculate three types of virtual diagrams (see Fig. 1.2): the internal gluon self-energies with a renormalization constant Z_A , the external quark self-energies with a renormalization constant Z_{ψ} and the gluon-fermion vertex $Z_{A\overline{\psi}\psi}$. The strong coupling renormalization constant Z_g is related to $Z_{A\overline{\psi}\psi}$, Z_A and Z_{ψ} by

$$Z_{A\overline{\psi}\psi} = Z_g Z_A^{1/2} Z_\psi \tag{1.44}$$

where¹¹

$$Z_{A} = 1 + \frac{\alpha_{s}}{4\pi} \left(\frac{5}{3} C_{A} - \frac{4}{3} n_{f} T_{R} \right) \Gamma(\epsilon) \mu^{-2\epsilon}$$

$$Z_{\psi} = 1 - \frac{\alpha_{s}}{4\pi} C_{F} \Gamma(\epsilon) \mu^{-2\epsilon}$$

$$Z_{A\overline{\psi}\psi} = 1 + \frac{\alpha_{s}}{4\pi} (C_{F} + C_{A}) \Gamma(\epsilon) \mu^{-2\epsilon}$$
(1.45)

For the gluon self-energies (see the first block of Fig. 1.2) the fermion loop contribution gives

$$i \left(q^2 g^{\mu\nu} - q^{\mu} q^{\nu}\right) \delta^{AB} \left(-\frac{\alpha_s}{4\pi} \frac{4}{3} n_f T_R \mu^{-2\epsilon} \frac{1}{\epsilon} + \cdots\right)$$
(1.46)

while the rest of the diagrams give

$$i \left(q^2 g^{\mu\nu} - q^{\mu} q^{\nu}\right) \delta^{AB} \left(\frac{\alpha_s}{4\pi} \frac{5}{3} C_A \mu^{-2\epsilon} \frac{1}{\epsilon} + \cdots\right)$$
(1.47)

For the fermion self-energies (see the second block of Fig. 1.2), we have

$$i\frac{\alpha_s}{4\pi}\not\!\!\!/ C_F\,\mu^{-2\epsilon}\frac{1}{\epsilon}+\cdots$$
(1.48)

and for the gluon-fermion vertex (see the last block of Fig. 1.2), we have

$$i\frac{\alpha_s}{4\pi}g_s t^A \gamma^\mu \left(C_F + C_A\right) \mu^{-2\epsilon} \frac{1}{\epsilon} + \cdots$$
(1.49)

We define the function $\tilde{\beta}(g_s)$ via the relation $\beta(\alpha_s) = g_s \tilde{\beta}(g_s)/(4\pi)$, where α_s and g_s are related by $g_s^2 = 4\pi \alpha_s$. Hence, the result for $\tilde{\beta}(g_s)$ is¹²

$$\tilde{\beta}(g_s) = (-2) \frac{g_s^3}{(4\pi)^2} \left[(C_F + C_A) - C_F + \frac{1}{2} \left(\frac{5}{3} C_A - \frac{4}{3} n_f T_R \right) \right]$$
(1.50)

¹²Here we have changed the notation for the strong coupling constant by g_s , which we denoted with g in the last section.

¹¹Here we calculate in $d = 4 - 2\epsilon$ dimensions.

or equivalently

$$\beta(\alpha_s) = -\frac{\alpha_s^2}{2\pi}\beta_0 \tag{1.51}$$

The coupling constant α_s has an expression, which relates two different scales: Q^2 at which α_s is calculated and the renormalization scale μ^2 . At leading order in the perturbative expansion, we can solve Eq. (1.42) with Eq. (1.43) to obtain

$$\alpha_s(Q^2) = \frac{\alpha_s(\mu^2)}{1 + \alpha_s(\mu^2) b_0 \ln(Q^2/\mu^2)}, \qquad b_0 = \frac{\beta_0}{2\pi}$$
(1.52)

In QCD with $n_f \leq 16$ (or $\beta < 0$) the strong coupling α_s becomes weaker as the energy scale increases. This is known as asymptotic freedom. This behaviour of asymptotic freedom is in contrast to QED where the coupling becomes strong at high energies. In QED the β function is

$$\beta_{\text{QED}}(\alpha) = \frac{1}{3\pi} \alpha^2 + \cdots$$
 (1.53)

1.6 Parton branching at Next-to-Leading Order

In this section we will discuss in detail infrared and collinear singularity¹³ that we used throughout the main part of this thesis. Fig. 1.3 shows the kinematics and notation for the splitting of parton a into b and c in the final state, *e.g.* a virtual quark splits into a real quark plus a real gluon.

For the propagator we have¹⁴

propagator =
$$\frac{1}{(p+k)^2 - m^2} = \frac{1}{2p \cdot k} = \frac{1}{2E_b E_c (1 - A \cos \vartheta)}$$
 (1.54)

We have to take the square of the amplitude and integrate over the final state phase space, all together, we get $d^3\vec{k}/E_b \cdot 1/E_b^2 \sim E_b dE_b \cdot 1/E_b^2 \sim dE_b/E_b$. When E_b goes to zero this corresponds to a soft singularity. For $m \to 0$ we have $A \to 1$ and $(1 - A \cos \vartheta)$ vanishes at $\cos \vartheta = 1$. This corresponds to a collinear mass singularity. However, infrared (soft plus

$$-i e Q_f \gamma_\mu \rightarrow -i g_s t^A_{ab} \gamma_\mu$$

¹⁴Here we choose $p = (E_c, \vec{p}_c)$, $k = (E_b, \vec{p}_b)$, hence $|\vec{p}_c| = E_c \sqrt{1 - m^2/E_c^2} \equiv E_c A$ and $|\vec{p}_b| = E_b$. Note also that the notation p_b and k can be exchanged with each other without making any difference; *i.e.* we will use both notations interchangeably without being noticed. Similarly for p_c and p.

¹³We will explain this in more detail in Section 5.5 and Section 5.6, where we discussed soft and collinear photon radiations. The generalization to soft and collinear gluon radiations in QCD is straightforward, one can simply replace photon field with gluon field and take the colour factors and QCD coupling constant g_s into account, *i.e.* we make the following substitution for vertex:



Figure 1.3: A final-state parton branching.

collinear) singularities cancel, if we add virtual diagrams (see *e.g.* Fig. 1.2). This is a consequence of the KLN theorem [10, 11].

In QCD there is a generic property, that the real emission amplitude ((m+1)-parton phase space) can be factorized into a Born-level amplitude (*m*-parton phase space) in the soft and collinear limits based on the factorization theorems

$$\mathcal{M}_{m+1} \approx v_{\ell} \cdot \mathcal{M}_m \tag{1.55}$$

where v_{ℓ} means the singular factor. One can also refer to Eq. (5.116) in appendix for soft-gluon approximation and Eq. (5.133) in appendix for collinear gluon emission¹⁵, *e.g.* for a collinear final-state qqg parton branching (see Fig. 1.3) we have

$$\mathcal{M}_{m+1} \sim \bar{u}(p) \left[-i g_s t^A_{ab} \gamma_\mu \right] \varepsilon^{*\mu}(k) \frac{i \not p_a}{p_a^2} \mathcal{A}(p_a) \sim \frac{1}{t} \bar{u}(p) \left[g_s t^A_{ab} \gamma_\mu \right] \varepsilon^{*\mu}(k) \sum_{\text{spins}} u(p_a) \underbrace{\bar{u}(p_a) \mathcal{A}(p_a)}_{=\mathcal{M}_m} \\ \sim \sum_{\text{spins}} \frac{1}{t} \bar{u}(p) \left[g_s t^A_{ab} \gamma_\mu \right] \varepsilon^{*\mu}(k) u(p_a) \mathcal{M}_m$$
(1.56)

where $\bar{u}(p)$ is the spinor of the fermion and $\mathcal{A}(p_a)$ the remaining part of the amplitude. Hence¹⁶

$$|v_{\ell}|^2 \sim \frac{1}{t} P_{qq}(z)$$
 with $P_{qq}(z) = C_F \frac{1+z^2}{1-z}$ (1.57)

¹⁵In Section 5.5 and Section 5.6 we discussed photon radiations; for the gluon radiations we simply make the following substitution for vertex:

$$-\,i\,e\,Q_f\,\gamma_\mu\,
ightarrow\,-\,i\,g_s\,t^A_{ab}\,\gamma_\mu$$

¹⁶Here $P_{qq}(z)$ is the spin-averaged splitting function [6]. In the collinear limit, the radiation of additional partons is described by universal splitting functions, independent of the hard interaction. The Eq. (1.57) shows that the $m \to m+1$ amplitude factorizes into a soft/collinear part 1/t from the propagator, the splitting function $P_{qq}(z)$ and a hard $m \to m$ amplitude without IR singularities.

Here we have assumed that¹⁷

$$p_a^2 = (p_b + p_c)^2 \equiv t \gg p_b^2, \, p_c^2$$
(1.58)

and the energy fraction is defined by

$$z = E_b/E_a = 1 - E_c/E_a \tag{1.59}$$

Hence for small angles we obtain

$$t \equiv p_a^2 = 2 E_b E_c (1 - \cos \vartheta) = z (1 - z) E_a^2 \vartheta^2$$
 (1.60)

The limit where p_b and p_c become collinear can be precisely defined using Sudakov decomposition

$$p_{b} = z p_{a} + p_{T} + \zeta_{b} n$$

$$p_{c} = (1 - z) p_{a} - p_{T} + \zeta_{c} n$$

$$2 p_{b} \cdot p_{c} = -\frac{p_{T}^{2}}{z (1 - z)}, \quad p_{T} \to 0$$

$$p_{b}^{2} = 0 \Rightarrow \zeta_{b} = -\frac{p_{T}^{2}}{2 z n \cdot p_{a}}$$

$$p_{c}^{2} = 0 \Rightarrow \zeta_{c} = -\frac{p_{T}^{2}}{2 (1 - z) n \cdot p_{a}}$$
(1.61)

where $p_T \cdot p_a = p_T \cdot n = n^2 = 0$. In Eq. (1.61), n^{μ} is an auxiliary lightlike vector, that is necessary to specify the transverse component p_T ($p_T^2 < 0$). We can split the angle ϑ for massless partons according to Fig. 5.2 of [109]

$$\vartheta = \vartheta_b + \vartheta_c \quad \text{and} \quad \frac{\vartheta_b}{\vartheta_c} = \frac{p_T}{E_b} \left(\frac{p_T}{E_c}\right)^{-1} = \frac{1-z}{z} \quad \Leftrightarrow \quad \vartheta = \frac{\vartheta_b}{1-z} = \frac{\vartheta_c}{z} \quad (1.62)$$

In order to calculate the cross section for the various splitting processes, we need to know the factorization of the phase space from (m+1)-parton phase space into m-parton phase space. We consider the multi-particle phase space decomposition (see also appendix). For the branching of parton a into b + c, we can decompose the collinear phase space according to¹⁸

$$dPS_{m+1} = dPS_m (2\pi)^4 \delta^4 (p_a - p_b - p_c) \frac{d^3 \vec{p}_b}{(2\pi)^3 2 E_b} \frac{d^3 \vec{p}_c}{(2\pi)^3 2 E_c} \frac{dp_a^2}{2\pi}$$

$$= dPS_m \frac{1}{(2\pi)^3} \frac{1}{4 E_b E_c} \delta^4 (p_a - p_b - p_c) d^3 \vec{p}_b d^3 \vec{p}_c dt$$

$$= dPS_m \frac{1}{2 (2\pi)^3} \int E_b dE_b \vartheta_b d\vartheta_b d\varphi dt \frac{dz}{1-z} \delta(t - E_b E_c \vartheta^2) \delta(z - E_b/E_a)$$

$$= dPS_m \frac{1}{4 (2\pi)^3} dt dz d\varphi$$
(1.63)

¹⁷Here we choose $p_a = (E_a, \vec{p}_a)$. ¹⁸ $d^3\vec{p}_c = d^3\vec{p}_a$ at fixed \vec{p}_b , $d^3\vec{p}_b = E_b^2 dE_b \vartheta_b d\vartheta_b d\varphi$ and $\sin \vartheta_b \sim \vartheta_b$.

where φ is the azimuthal angle. Adding the matrix elements to this factorization of the collinear phase space we can derive the cross section for one collinear emission

$$d\sigma_{m+1} \sim |\mathcal{M}_{m+1}|^2 dPS_{m+1}$$

$$= |\mathcal{M}_{m+1}|^2 dPS_m \frac{1}{4(2\pi)^3} dt dz d\varphi$$

$$= |\mathcal{M}_{m+1}|^2 dPS_m \frac{1}{4(2\pi)^2} dt dz \quad \text{spherically symmetric}$$

$$= \frac{2 g_s^2}{t} P_{ab}(z) |\mathcal{M}_m|^2 dPS_m \frac{dt dz}{16\pi^2} \quad \text{assuming} \quad \left||\mathcal{M}_{m+1}|^2 = \frac{2 g_s^2}{t} P_{ab}(z) |\mathcal{M}_m|^2\right|$$
(1.64)

Here we have neglected the initial-state flux factor \mathcal{F} . Using $d\sigma_m \sim |\mathcal{M}_m|^2 dPS_m$ we can write the most general form of Eq. (1.64)

$$d\sigma_{m+1} = d\sigma_m \frac{dt}{t} dz \frac{\alpha_s}{2\pi} P_{ab}(z)$$
(1.65)

The Eq. (1.65) means that we can calculate the (m+1)-particle cross section from the *m*-particle cross section convoluted with the universal splitting functions $P_{ab}(z)$. In Chapter 2 we will discuss various branching processes, *e.g.* a quark splitting into a quark and a gluon (qqg), a gluon splitting into two quarks $(gq\bar{q})$, a gluon splitting into two gluons (ggg) and derive alternative splitting functions in the final (initial) states using a slightly different kinematics and momentum mapping. Our new subtraction scheme is based on these improved splitting functions.

Chapter 2

Nagy-Soper dipoles

2.1 Motivation

The main topic of this thesis is the calculation of QCD cross sections in high energy hadron colliders or lepton colliders at Next-to-Leading Order (NLO) accuracy. For the LHC we will be faced with complex hadronic scatterings with many particles in the final state and we need to understand the standard model (SM) predictions precisely in order to dig out any signal from physics beyond the SM (BSM). Therefore, processes have in general to be calculated at least to NLO precision. Another reason why we bother with higher order calculations is that the naïve parton model picture corresponds to the Leading-Order (LO) approximation; and the LO calculations only predict the rough order of magnitude of a given cross section due to poor convergence of perturbative expansion; there is still a strong dependence on the unphysical input scales (renormalization and factorization scales). NLO QCD calculations can help us to reduce dependence on the renormalization scale for observables including $\alpha_s(\mu_F^2)$, which at the end leads to stable predictions for the cross sections. In higher order calculations, we have to consider real emission corrections and virtual contributions. Furthermore we will often be facing two different sources of singularities: ultraviolet (UV) divergences and infrared (IR) divergences; the UV singularities, which are only present in the virtual diagrams, can be removed by the standard renormalization procedure and infrared singularities (soft and collinear), which instead can appear both in the real and in the virtual contributions, also cancel when we sum over real and virtual contributions. That is a consequence of the KLN theorem. In general, when we compute cross sections with initial state hadrons; there are still left-over collinear singularities, which need to be factorized into the universal and process independent parton distribution functions (PDFs). As a result, cross sections are finite at the parton level order by order in perturbation theory. Recent progress for results at NLO have been available for all $2 \rightarrow 2$ and $2 \rightarrow 3$, and for some $2 \rightarrow 4$ processes at hadron colliders.

There are, generally speaking, two types of algorithms widely used for dealing with the infrared divergences in NLO QCD calculations: the <u>phase space slicing</u> and the <u>subtraction</u> methods. Now suppose that we are interested in calculating the integral

$$I = \lim_{\epsilon \to 0} \left(\int_0^1 \frac{dx}{x} \, x^{-\epsilon} \, M(x) + \frac{1}{\epsilon} \, M_0 \right) \tag{2.1}$$

where M(x) is a complicated function which is nonsingular at x = 0; and it depends on the hard scattering processes. The first term on the right-hand side can be thought of the contribution from real graphs and the second term plays the role of the contribution from virtual diagrams. Here xis analogous to the energy of a gluon, or to the angle between two partons. There is a singularity at x = 0. Using dimensional regularization by lowering the dimension to $d = 4 - 2\epsilon$, this singularity is regularized by a factor $x^{-\epsilon}$. The integral is divergent as $\epsilon \to 0$, but the divergence is cancelled by the second term $+ (1/\epsilon) M_0$, which is a result of the KLN theorem. In this case KLN theorem also ensures that $\lim_{x\to 0} M(x) = M_0$. In practice, the function M(x) could be very complicated for an increasing number of external particles in the final state. The question is how to calculate the value of I numerically (and efficiently) if the function M(x) is too complicated such that an analytic computation cannot be allowed.

Slicing: the widely used method is called the phase space slicing method. Introducing an arbitrary cutoff δ (we choose δ ≪ 1 and δ ≫ |ε|), one can split the integration region into two pieces: 0 < x < δ and δ < x < 1. For the region 0 < x < δ, we can use the simple approximation that M(x) → M(0). This gives

$$I = \int_{\delta}^{1} \frac{dx}{x^{1+\epsilon}} M(x) + \int_{0}^{\delta} \frac{dx}{x^{1+\epsilon}} M(x) + \frac{1}{\epsilon} M_{0} \simeq \int_{\delta}^{1} \frac{dx}{x} M(x) + \frac{1}{\epsilon} \left(1 - \delta^{-\epsilon}\right) M_{0}$$
$$\simeq \int_{\delta}^{1} \frac{dx}{x} M(x) + \ln(\delta) M_{0}$$
(2.2)

Now the first integration can be integrated numerically in the respective Monte Carlo program. As long as δ is small, the result will be independent of δ . The details of this method are explained by Baer, Ohnemus and Owens [12] in the context of a calculation of photoproduction of jets. This method has also been applied to NLO calculations of three-jet cross sections in e^+e^- annihilation [13, 14].

• Subtraction: the idea was first used for QCD calculations of jet structure in e^+e^- annihilation by R. K. Ellis, Ross and Terrano [15], and later by Z. Kunszt and P. Nason [16]. The basic idea is that one can write

$$I = \int_{0}^{1} \frac{dx}{x^{1+\epsilon}} M(x) - \int_{0}^{1} \frac{dx}{x^{1+\epsilon}} M_{0} + \int_{0}^{1} \frac{dx}{x^{1+\epsilon}} M_{0} + \frac{1}{\epsilon} M_{0}$$

= $\int_{0}^{1} dx \frac{M(x) - M_{0}}{x^{1+\epsilon}} + \left(-\frac{1}{\epsilon} + \frac{1}{\epsilon}\right) M_{0} \simeq \int_{0}^{1} \frac{dx}{x} \left[M(x) - M_{0}\right]$ (2.3)

The integration can now be performed numerically by Monte Carlo integration. In summary, both the phase space slicing and the subtraction algorithms provide the foundation for setting up a Monte Carlo program, which can be used to implement arbitrary higher order QCD calculations in a given process. As for the subtraction algorithm, a general NLO formalism has been applied to calculate three-jet cross sections in e^+e^- annihilation and cross sections up to two-jet production in the final state at hadron colliders [17–19]; the algorithm of [17] has been modified to deal with three-jet cross sections at next-to-leading order [20]. This formalism is also applicable to n-jet production in e^+e^- annihilation and in hadron collisions. The treatment of massive partons has also been considered in the case of heavy quark correlations in hadron collisions at next-to-leading order [20,21].

In recent years, an important calculational tool for the implementation of NLO QCD corrections in Monte Carlo style programs are dipole subtraction schemes [22–25]. The key point for the dipole subtraction method is that the QCD squared real-emission matrix element can be factorized into Born matrix element in the soft and collinear limits based on the factorization properties of QCD matrix elements [26, 27]. Dipole subtraction schemes introduce local counterterms, which mimic the behaviour of the real-emission matrix element in the singular limits. After standard UV-renormalization, the soft and collinear singularities then cancel when the integrated subtraction terms are added to the virtual cross section. Hence, the results to the NLO cross section are finite and the further phase-space integrations can be performed numerically by Monte Carlo techniques.

The various schemes [22,23,25] differ in the phase-space momentum mapping, which relates LO and NLO kinematics. In the standard scheme of Catani and Seymour [22], the universal local counterterms need to be re-calculated for each emitter/spectator pair¹. Therefore, this scheme suffers from a large number of momentum mappings needed to evaluate the subtraction terms. Basically, the number of momentum mappings scales like N^3 for a LO $2 \rightarrow N$ process. This scaling leads to a rapidly rising number of momentum mappings for a large number of external particles in the final state. Following an approach suggested by Zoltan Nagy and Dave Soper [28–30], we employ a subtraction scheme with a slightly altered momentum mapping, such that the number of kinematic transformations is greatly reduced. Basically, the number of mappings scales like N^2 for a LO $2 \rightarrow N$ process, thereby reducing the number of matrix element computations by a factor of N. In addition, the dipole subtraction terms in this alternative scheme are based on splitting functions which have been proposed in the context of an improved parton shower formulation including quantum interference effects. Hence, the new scheme facilitates numerical implementations of higher order corrections in Monte Carlo Event Generators and also allows for easy matching with a parton shower using the same splitting functions.

¹The number of momentum mappings = number of emitters.

We begin in Section 2.2 by giving a brief overview of the general subtraction procedure. In Section 2.3 we discuss the general framework setup and the momentum mapping between m- and (m + 1)-particle phase space. In Section 2.4 we will give the explicit expressions of splitting functions for each process in both the initial and final states, as well as the eikonal splitting functions and soft splitting functions. In Section 2.5 we will show the complete integrated splitting functions including collinear and soft integrals. In Chapter 3 we will show our first applications to NLO processes at hadron and lepton colliders. Finally we will summarize in Chapter 4.

2.2 General structure of the NLO cross section and subtraction procedure

2.2.1 The general subtraction procedure

In this Section we explain the general subtraction procedure for calculating NLO cross sections at lepton and hadron colliders. Suppose that we want to calculate the jet quantity σ at NLO accuracy

$$\sigma = \sigma^{LO} + \sigma^{NLO} \tag{2.4}$$

Suppose also that there are m partons in the final state at LO, then we have

$$\sigma^{LO} = \int_m d\sigma^B \tag{2.5}$$

Here, $d\sigma^B$ is the Born-level cross section, which can be symbolically written as

$$d\sigma^B = dPS_m \left| \mathcal{M}_m \right|^2 F_J^{(m)} \tag{2.6}$$

where dPS_m denotes the phase space of m particles in four dimensions, \mathcal{M}_m is the matrix element and $F_J^{(m)}$ is a function of cuts defining the jet observables, which we will discuss in Section 2.2.3. By definition, the LO cross section is finite so that Eq. (2.5) can be integrated (analytically or numerically) in four dimensions.

At NLO, we have to consider both the real and virtual contributions. There are m + 1 partons in the final state for the real emission and m partons in the final state for the virtual one-loop correction. So we can write

$$\sigma^{NLO} = \int d\sigma^{NLO} = \int_{m+1} d\sigma^R + \int_m d\sigma^V$$
(2.7)

The first integral on the right-hand side of Eq. (2.7) is the contribution from real diagrams, which contains IR divergences, and the second integral on the right-hand side of Eq. (2.7) is the contribution from virtual diagrams, which contains both UV and IR divergences. A traditional way

for dealing with IR singularities is by introducing an infinitesimal regulator, e.g. by lowering the dimension to $d = 4 - 2\epsilon$, the so-called dimensional regularization scheme, in which the Feynman diagrams are computed in d dimensions and the singularities in the integral can be extracted as double (soft and collinear) poles $1/\epsilon^2$ and single (soft, collinear or UV) poles $1/\epsilon$. Here we suppose that one has already performed the renormalization procedure in $d\sigma^V$ so that all its UV singularities have been removed. This way, the analytic cancellation of the respective divergent parts for fully inclusive measurements is straightforward. However, numerical implementations of parts containing infinitesimal regulators for multi-particle processes proved to be challenging. In subtraction schemes, the difficulty is circumvented by introducing universal local counterterms (or dipole terms), which mimic the behaviour of the squared real emission matrix elements in the singular regions; adding back the respective one particle integrated counterparts to the virtual contributions results in finite integrands for both real contribution (m + 1-particle phase space) and virtual correction (m-particle phase space). Symbolically, we write

$$d\sigma^{NLO} = \left[d\sigma^R - d\sigma^A\right] + \left[d\sigma^A + d\sigma^V\right]$$
(2.8)

where $d\sigma^A$ is regarded as a local counterterm (or dipole), which mimics the singular behaviour of $d\sigma^R$. In the next step we introduce the phase space integration. One can safely perform the limit $\epsilon \to 0$ in the first term on the right-hand side of Eq. (2.8) by definition. For the second term on the right-hand side of Eq. (2.8), one can carry out analytically the integrated dipole term $d\sigma^A$ over the emitted one parton phase space, leading to the poles in ϵ that are necessary to cancel the soft singularities in virtual one-loop cross section. Thus we can perform the limit $\epsilon \to 0$ after all the divergences are cancelled. The final structure of the calculation is given by

$$\sigma^{NLO} = \underbrace{\int_{m+1} \left[d\sigma^R - d\sigma^A \right]}_{\text{finite}} + \underbrace{\int_{m+1} d\sigma^A + \int_m d\sigma^V}_{\text{finite}}$$
$$= \int_{m+1} \left[d\sigma^R_{\epsilon=0} - d\sigma^A_{\epsilon=0} \right] + \int_m \left[\int_1^{\Lambda} d\sigma^A + d\sigma^V \right]_{\epsilon=0}$$
(2.9)

Both integrands are now finite, meaning that we can perform all integrations numerically in the respective Monte Carlo program. The explicit expressions of the cross section σ for m and m+1 particle contributions to the total NLO cross section are given by

$$\int_{m} \left[d\sigma^{B} + d\sigma^{V} + \int_{1} d\sigma^{A} \right] = \int dP S_{m} \left[\left| \mathcal{M}_{m} \right|^{2} + \left| \mathcal{M}_{m} \right|^{2}_{\text{one-loop}} + \sum_{\ell} \mathfrak{B}_{\ell} \left| \mathcal{M}_{m} \right|^{2} \right]$$
$$\int_{m+1} \left[d\sigma^{R} - d\sigma^{A} \right] = \int dP S_{m+1} \left[\left| \mathcal{M}_{m+1} \right|^{2} - \sum_{\ell} \mathcal{D}_{\ell} \left| \mathcal{M}_{m} \right|^{2} \right]$$
(2.10)

Here, $\epsilon \to 0$ in the m + 1 particle phase space is always understood. Convolution with jet functions then ensures that the σ is infrared and collinear safe and that the Born-level contribution



Figure 2.1: Dipole factorization procedure [22]. When the partons *i* and *j* become soft and/or collinear, the (m + 1)-parton matrix element factorizes into the *m*-parton matrix element times a singular factor $V_{ij,k}$ which in our notation is denoted with $v_{\ell}(\{\hat{p}\}_{m+1})$.

is well defined. In Eq. (2.10), the sum goes over all local counterterms needed to match the complete singularity structure of the real emission contribution. For each singular limit, *i.e.* when two partons become collinear or when one parton becomes soft, the real emission matrix element factorizes into the Born-level matrix element according to² (Fig. 2.1)

$$\mathcal{M}_{m+1}(\{\hat{p}\}_{m+1}) \longrightarrow \sum_{\ell} v_{\ell}(\{\hat{p}\}_{m+1}) \otimes \mathcal{M}_m(\{p\}_m) \quad \text{and} \quad \mathcal{D}_{\ell} \propto v_{\ell}^2$$
(2.11)

where \mathcal{D}_{ℓ} denote the dipoles containing the respective singularity structure. Here \mathcal{D}_{ℓ} is just a symbolic notation. The explicit expressions for each splitting process will be given in Section 2.4. The symbol \otimes represents properly defined phase-space, spin and colour convolutions. \hat{p} and p represent momenta in (m+1)- and m-parton phase spaces, respectively. As $|\mathcal{M}_{m+1}|^2$ and $|\mathcal{M}_m|^2$ live in different phase spaces, a mapping of the respective momenta from (m+1)- to m-particle phase space needs to be introduced, which is defined by a mapping function F_{map} according to $p = F_{\text{map}}(\hat{p})$. \mathcal{D}_{ℓ} and \mathfrak{B}_{ℓ} are related by $\mathfrak{B}_{\ell} = \int d\zeta_p \mathcal{D}_{\ell}$, where $d\zeta_p$ is an unresolved one parton integration measure. In summary, any subtraction scheme needs to fulfill the following requirements:

- dipole subtraction terms \mathcal{D}_{ℓ} must match the behavior of the real emission matrix element in each soft and collinear region.
- the mapping function F_{map} guarantees total energy momentum conservation as well as the on-shell condition for all external particles both before and after the mapping.

²In Fig. 2.1 we follow Catani-Seymour's notation to explain the dipole factorization procedure; it is worth mentioning that in this thesis we used ℓ for the mother parton instead of ij, which Catani and Seymour used.



Figure 2.2: Hadron collision.

• subtraction terms have to lead to correct IR poles when carrying out the analytical integration over the one parton phase space in *d* dimensions that are necessary to cancel the soft singularities in virtual one-loop matrix element.

2.2.2 Generalization to hadron collisions

Now we consider the cross sections in hadron collisions (Fig. 2.2). In the case of processes with two initial-state hadrons A and B carrying momenta p_A and p_B , respectively, the calculation of the QCD cross sections must be convoluted with parton distribution functions (PDFs):

$$\sigma(p_A, p_B) = \sum_{a, b} \int_0^1 d\eta_a \, f_{a/A}(\eta_a, \mu_F^2) \int_0^1 d\eta_b \, f_{b/B}(\eta_b, \mu_F^2) \left[\sigma_{ab}^{LO}(p_a, p_b) + \sigma_{ab}^{NLO}(p_a, p_b, \mu_F^2) \right] \\ + \mathcal{O}\left(\frac{\Lambda_{\text{QCD}}^n}{Q^n}\right)$$
(2.12)

where $p_a = \eta_a p_A$ and $p_b = \eta_b p_B$ are parton momenta, while η_a and η_b are the momentum fractions of the partons. The parton distribution functions $f_{a/A}(\eta_a, \mu_F^2)$ gives the probability of finding parton *a* inside hadron *A* with momentum fraction η_a when the hadron is probed at the scale μ_F . In general, the hard scattering cross sections σ^{LO} and σ^{NLO} depend on $\alpha_s(\mu_F^2)$ and the ratio Q^2/μ_F^2 . The parameter μ_F is the factorization scale at which long distance physics (PDFs) and short distance process (hard scattering cross sections) can be separately treated. The scale μ_F is arbitrary that is introduced in order to renormalize the UV divergences after quantization, and physical results cannot depend on it. The corresponding parton level cross sections are:

$$\sigma_{ab}^{LO}(p_a, p_b) = \int_m d\sigma_{ab}^B(p_a, p_b)$$

$$\sigma_{ab}^{NLO}(p_a, p_b, \mu_F^2) = \int_{m+1} d\sigma_{ab}^R(p_a, p_b) + \int_m d\sigma_{ab}^V(p_a, p_b) + \int_m d\sigma_{ab}^C(p_a, p_b, \mu_F^2)$$
(2.13)

In general, the initial-state collinear singularities in hadron collisions do not cancel. However, they are universal and process-independent in all orders in perturbation theory. Therefore they can be cancelled by universal collinear counterterms $d\sigma_{ab}^{C}$, which are generated by the renormalization of the PDFs of the incoming particles. The explicit form is given by³:

$$\int_{m} d\sigma_{ab}^{C}(p_{a}, p_{b}, \mu_{F}^{2}) = \frac{\alpha_{s}}{2\pi} \frac{1}{\Gamma(1-\epsilon)} \sum_{c} \int_{0}^{1} dx \int_{m} d\sigma_{cb}^{B}(xp_{a}, p_{b}) \frac{1}{\epsilon} \left(\frac{4\pi\mu^{2}}{\mu_{F}^{2}}\right)^{\epsilon} P^{ac}(x) + \frac{\alpha_{s}}{2\pi} \frac{1}{\Gamma(1-\epsilon)} \sum_{c} \int_{0}^{1} dx \int_{m} d\sigma_{ac}^{B}(p_{a}, xp_{b}) \frac{1}{\epsilon} \left(\frac{4\pi\mu^{2}}{\mu_{F}^{2}}\right)^{\epsilon} P^{bc}(x) \quad (2.14)$$

Here, the $P^{ab}(x)$ are the Altarelli-Parisi kernels in four dimensions. The collinear counterterm depends on the factorization scheme. Here we have chosen the most commonly used $\overline{\text{MS}}$ scheme. As in the case of UV renormalization, the full hadronic cross section is unaltered under a change of the factorization scheme, provided that the PDFs are also suitably changed.

For processes with incoming hadrons, the subtraction procedure is applied to $\sigma_{ab}^{NLO}(p_a, p_b, \mu_F^2)$ as previously described and we can write it as follows

$$\sigma_{ab}^{NLO}(p_a, p_b, \mu_F^2) = \int_{m+1} \left[d\sigma_{ab}^R(p_a, p_b) - d\sigma_{ab}^A(p_a, p_b) \right] \\ + \int_m \left[\int d\sigma_{ab}^V(p_a, p_b) + \int_1 d\sigma_{ab}^A(p_a, p_b) + d\sigma_{ab}^C(p_a, p_b, \mu_F^2) \right]_{\epsilon=0}$$
(2.15)

where $\int_{1} d\sigma_{ab}^{A} + d\sigma_{ab}^{C}$ can be written as

$$\int_{m} \left[\int_{1} d\sigma_{ab}^{A}(p_{a}, p_{b}) + d\sigma_{ab}^{C}(p_{a}, p_{b}, \mu_{F}^{2}) \right]$$

$$= \int_{m} d\sigma_{ab}^{B}(p_{a}, p_{b}) \otimes I(\epsilon) + \int_{0}^{1} dx \int_{m} d\sigma_{ab}^{B}(xp_{a}, p_{b}) \otimes \left[K^{a}(xp_{a}) + P(x, \mu_{F}^{2}) \right]$$

$$+ \int_{0}^{1} dx \int_{m} d\sigma_{ab}^{B}(p_{a}, xp_{b}) \otimes \left[K^{b}(xp_{b}) + P(x, \mu_{F}^{2}) \right]$$
(2.16)

This equation defines the insertion operators $I(\epsilon)$, K(x), $P(x, \mu_F^2)$ at the cross section level, where we follow the standard notation introduced in [22]. The symbol \otimes denotes all possible spin, colour and flavour correlations. Eq. (2.16) can be divided into two parts: the first part

³Here the splitting functions P_{ab} and P^{ab} can be exchanged with each other without making any difference, *i.e.* $P_{ab} = P^{ab}$ in the later discussions.

is the universal insertion operator $I(\epsilon)$, which contains all the poles in ϵ that are necessary to cancel the soft singularities in the virtual cross section. The universal insertion operator has the LO kinematics. The second part consists of the finite pieces that are left over after absorbing the initial-state collinear singularities into a redefinition of the parton distribution functions at NLO. It involves an additional one dimensional integration over the momentum fraction x of an incoming parton with the LO cross sections and the x-dependent structure functions.

2.2.3 Observable-dependent formulation of the subtraction method

The jet observables which we are interested in should be well defined experimentally or theoretically in such a way, that the cross sections are infrared and collinear safe. In general, the jet function may contain θ functions (which define cuts and corresponding cross sections) and δ functions (which define differential cross sections). To be more specific, we consider the following expressions for Born-level and the corresponding NLO cross sections

$$\sigma^{LO} = \int dP S_m(p_1, \cdots, p_m) \left| \mathcal{M}_m(p_1, \cdots, p_m) \right|^2 F_J^{(m)}(p_1, \cdots, p_m)$$

$$\sigma^{NLO} = \int dP S_{m+1}(p_1, \cdots, p_{m+1}) \left| \mathcal{M}_{m+1}(p_1, \cdots, p_{m+1}) \right|^2 F_J^{(m+1)}(p_1, \cdots, p_{m+1})$$

$$+ \int dP S_m(p_1, \cdots, p_m) \left| \mathcal{M}_m(p_1, \cdots, p_m) \right|^2_{\text{one-loop}} F_J^{(m)}(p_1, \cdots, p_m)$$
(2.17)

where $|\mathcal{M}_m|^2$, $|\mathcal{M}_{m+1}|^2$ and $|\mathcal{M}_m|^2_{one-loop}$ are the squared LO matrix element, the squared NLO real emission matrix element and the squared NLO virtual matrix element, respectively. There is a formal requirement on the jet function $F_J^{(m)}$. For cases where two partons become collinear or where one parton becomes soft, the function $F_J^{(m+1)}$ reduces to $F_J^{(m)}$, *i.e.* in the soft and collinear limits, the jet function fulfils the following properties

$$F_{J}^{(m+1)}(p_{1}, \cdots, p_{j} = \lambda q, \cdots, p_{m+1}) \to F_{J}^{(m)}(p_{1}, \cdots, p_{m+1}) \quad \text{if} \quad \lambda \to 0$$

$$F_{J}^{(m+1)}(p_{1}, .., p_{i}, .., p_{j}, .., p_{m+1}) \to F_{J}^{(m)}(p_{1}, .., p, .., p_{m+1}) \quad \text{if} \quad p_{i} \to z p, \ p_{j} \to (1-z) p$$

$$F_{J}^{(m)}(p_{1}, \cdots, p_{m}) \to 0 \quad \text{if} \quad p_{i} \cdot p_{j} \to 0$$
(2.18)

The first two conditions of Eq. (2.18) define the essential property of the jet function that the jet observable has to be infrared and collinear safe for any number m of partons in the final state, *i.e.* to any order in QCD perturbation theory. The last condition of Eq. (2.18) guarantees that the Born-level cross section is well defined. To summarise, we require that

$$F_J^{(m+1)} \to F_J^{(m)}$$

in the singular limits.

2.3 Scheme setup and momentum mapping

In the scheme presented here, the NLO dipole subtraction terms are derived from the splitting functions introduced in [28]. The subtraction scheme is based on a physical picture that partons can split or join. For parton splitting, one of the m partons splits, producing m + 1 partons in the final state. For parton joining, two partons can join, reducing the m + 1 partons back into m partons. Parton splitting is needed to describe a parton shower, while parton joining is relevant in NLO QCD calculations. A momentum mapping function is needed to relate different phase spaces, and the mappings from m + 1 to m partons needed correspond to the inverse transformation of the respective shower m to m + 1 partons mappings.

In the following, the partons are labelled by an index that takes values a or b for the initial state partons and $1, 2, \dots, m$ for the final state partons. We will denote an m parton state by momenta $\{p\}_m$. The partons will be labelled by $\{a, b, 1, 2, \dots, m\}$ with momenta

$$\{p_a, p_b, p_1, p_2, \cdots, p_m\}$$

After the splitting, we have an m + 1 parton state with momenta $\{\hat{p}\}_{m+1}$. The momenta are labelled by

$$\{\hat{p}_a, \hat{p}_b, \hat{p}_1, \hat{p}_2, \cdots, \hat{p}_{m+1}\}$$

Now suppose that partons ℓ and j are produced by the splitting of a mother parton. Here the mother parton is in an m-parton state $\{p\}_m$, while partons ℓ and j are in an (m + 1)-parton state $\{\hat{p}\}_{m+1}$. There are two situations. For a final state splitting, ℓ and j are in the $\{\hat{p}_1, \hat{p}_2, \dots, \hat{p}_{m+1}\}$ configuration. The mother parton emerges from the hard interaction and splits into partons ℓ and j. In this case, the momentum of the mother parton is labelled by $p_{\ell} \in \{p_1, p_2, \dots, p_m\}$ while the momenta of partons ℓ and j are labelled by \hat{p}_{ℓ} and \hat{p}_j , respectively. For an initial state splitting, j is in the $\{\hat{p}_1, \hat{p}_2, \dots, \hat{p}_{m+1}\}$ configuration and ℓ is in the $\{\hat{p}_a, \hat{p}_b\}$ configuration. Parton ℓ splits into parton j and an initial state parton that enters the hard interaction. Hence the momentum of mother parton is labelled by p_a (or p_b) while the momenta of partons ℓ and j are labelled by p_a (or p_b) and \hat{p}_j , respectively. Our description for an initial state splitting follows backwards evolution, in which the initial state parton that enters the hard interaction is the mother parton. In summary, the notation is that mother parton ℓ splits into partons with labels ℓ and j for both initial and final state splittings, while the other parton state, \hat{p}_{ℓ} is the emitter, \hat{p}_j the emitted parton, and \hat{p}_k the spectator⁴.

The momenta $\{\hat{p}\}_{m+1}$ after splitting are determined by the momenta $\{p\}_m$ and a momentum splitting variable ζ_p , which defines the momenta of the daughter partons. Here we consider an

⁴In contrast to [22], in our case a spectator only needs to be specified if \hat{p}_i denotes a gluon.

example: when we join two partons ℓ and j, the momentum of mother parton, p_{ℓ} , is approximately $p_{\ell} \approx \hat{p}_{\ell} + \hat{p}_j$ for a final state splitting, and $p_{\ell} \approx \hat{p}_{\ell} - \hat{p}_j$ for an initial state splitting. As stated previously, hatted and unhatted momenta correspond to m + 1 and m-parton phase spaces, respectively. The momenta of the other partons, which we can denote by p_n , are approximately unchanged, meaning that $p_n \approx \hat{p}_n$ for $n \notin \{\ell, j\}$. However, these relations cannot be exact because the momenta after parton joining should be on shell. In order for the mother parton momentum p_{ℓ} to be on-shell, we have to take some momenta from the spectators, so $p_n \neq \hat{p}_n$ and $p_{\ell} \neq \hat{p}_{\ell} \pm \hat{p}_j$. So the definition of a mapping $\{p\}_m \leftrightarrow \{\hat{p}\}_{m+1}$ should guarantee that all external partons are on-shell, as well as total energy momentum conservation.

There are many ways to define the momentum mapping. The most widely used scheme is that of Catani and Seymour [22]. This scheme may be called a local mapping. They define a emit-ter/spectator pair, the momentum fraction goes to one additional parton only. Hence, the momenta of the remaining partons are left unaltered. In this scheme, the momentum mapping follows the rule that two partons are mapped into three partons according to

$$(p_{\ell}, p_k) \leftrightarrow (\hat{p}_{\ell}, \hat{p}_j, \hat{p}_k)$$

for each emitter/spectator pair. Here the color connection between a spectator and an emitter has to be considered. The antenna factorization [31, 32] also uses a local mapping. In the new scheme, we apply a global mapping, in which the mapping takes all the partons into account at once when going from (m + 1)- to m-particle phase space, instead of separately summing over all possible emitter/spectator pairs. We will restrict our expressions to subtractions on the parton level and to the massless case; details on convolution with PDFs are given in [28].

In the following subsections, we will first describe the final state splitting, then continue with the more complicated initial state splitting case. For the final state splitting we first show how $\{\hat{p}\}_{m+1}$ is obtained from $\{p\}_m$ and $\{\zeta_p\}$, then we reverse the transformation from $\{\hat{p}\}_{m+1}$ to $\{p\}_m$ and $\{\zeta_p\}$. In the same way we will derive the initial state splitting.

2.3.1 Splitting a final state parton

Parton splitting

We will neglect parton masses in the kinematics. Suppose the mother parton ℓ with momentum p_{ℓ} emerges from the hard interaction and splits into daughter partons ℓ and j with momenta \hat{p}_{ℓ} and \hat{p}_j , respectively. The on-shell condition ensures that $p_{\ell}^2 = \hat{p}_{\ell}^2 = \hat{p}_j^2 = 0$. We always have $(\hat{p}_{\ell} + \hat{p}_j)^2 \ge 0$. For a final state splitting, we can always leave the momenta of the initial state partons unchanged:

$$p_a = \hat{p}_a, \quad p_b = \hat{p}_b \tag{2.19}$$

Let Q be the total momentum of the final state partons

$$Q \equiv \sum_{n=1}^{m} p_n = p_a + p_b$$
 (2.20)

Here the momenta of the incoming partons remain the same, hence $\hat{Q} = \hat{p}_a + \hat{p}_b$ is the same as Q. We define

$$a_{\ell} = \frac{Q^2}{2\,p_{\ell} \cdot Q} \tag{2.21}$$

Note that $a_{\ell} \ge 1$. The transformation also has to keep all of the momenta on-shell. In order to define the momentum mapping for the final state splitting, we first parametrize the total momenta of the two daugther partons \hat{p}_{ℓ} and \hat{p}_{j} according to

$$P_{\ell} = \hat{p}_{\ell} + \hat{p}_{j} = \lambda \, p_{\ell} + \frac{1 - \lambda + y}{2 \, a_{\ell}} \, Q \tag{2.22}$$

There are two parameters in this definition: y and λ . y is the measure for the virtuality of the splitting and λ is a function of y that we will determine later. In order for the mother parton momentum p_{ℓ} to be on-shell as well as to preserve momentum conservation, we have to take some momentum fractions from the final state spectators. In contrast to the Catani-Seymour scheme, in which only a single spectator parton in the final state donates the needed momentum, we can choose that all of the final state spectator partons, except partons ℓ and j, donate a momentum fraction. So the needed momentum from all of the final state spectators can be obtained by a Lorentz transformation, which relates the momenta after and before the splitting,

$$\hat{p}_n^{\mu} = \Lambda(K, K)^{\mu}{}_{\nu} p_n^{\nu}, \quad n \notin \{\ell, j = m+1\}$$
(2.23)

Here, K is the total momentum of the final state spectators before the splitting

$$K = Q - p_{\ell} \tag{2.24}$$

and \hat{K} is the total momentum of the final state spectators after the splitting

$$\hat{K} = Q - P_{\ell} \tag{2.25}$$

Since each final state spectator is changed by a Lorentz transformation, we have

$$\hat{K}^{\mu} = \Lambda^{\mu}{}_{\nu}K^{\nu} \tag{2.26}$$

The Lorentz transformation is given by

$$\Lambda(\hat{K},K)^{\mu}{}_{\nu} = g^{\mu}{}_{\nu} - \frac{2\,(\hat{K}+K)^{\mu}\,(\hat{K}+K)_{\nu}}{(\hat{K}+K)^{2}} + \frac{2\,\hat{K}^{\mu}\,K_{\nu}}{K^{2}}$$
(2.27)

which automatically implies

$$\hat{K}^2 = K^2 \tag{2.28}$$

In a particular case in which there are only two massless partons in the final state, this corresponds to $a_{\ell} = 1$ and $K^2 = 0$. In this case, an alternative representation of the Lorentz transformation has to be introduced so that the boost in Eq. (2.27) still remains well defined when $K^2 = 0$.

$$\Lambda(\hat{K},K)^{\mu}{}_{\nu} = g^{\mu}{}_{\nu} + \left(\frac{K \cdot n}{\hat{K} \cdot n} - 1\right) n^{\mu} \bar{n}_{\nu} + \left(\frac{\hat{K} \cdot n}{K \cdot n} - 1\right) \bar{n}^{\mu} n_{\nu}$$
(2.29)

where n and \bar{n} are lightlike vectors in the Q- p_{ℓ} plane with $n \cdot \bar{n} = 1$ and $(p_{\ell} \cdot n/p_{\ell} \cdot \bar{n}) < (Q \cdot n/Q \cdot \bar{n})$.

The parameters λ and y can be determined from Eq. (2.28).

$$\lambda = \sqrt{(1+y)^2 - 4 a_{\ell} y}, \qquad y = \frac{\hat{p}_{\ell} \cdot \hat{p}_j}{p_{\ell} \cdot Q}$$
(2.30)

There is a maximum value of y corresponding to $\lambda = 0$

$$y_{\max} = \left(\sqrt{a_{\ell}} - \sqrt{a_{\ell} - 1}\right)^2 = 2a_{\ell} - 1 - 2\sqrt{a_{\ell}(a_{\ell} - 1)}$$
(2.31)

Another important relation, connecting hatted and unhatted quantities, is given by

$$p_{\ell} \cdot Q = (\hat{p}_{\ell} + \hat{p}_j) \cdot Q - \hat{p}_{\ell} \cdot \hat{p}_j$$
(2.32)

which can also be rewritten as

$$2 P_{\ell} \cdot Q = (1+y) 2 p_{\ell} \cdot Q \tag{2.33}$$

As stated previously, the momenta $\{\hat{p}\}_{m+1}$ after splitting are determined by the momenta $\{p\}_m$ and a momentum splitting variable ζ_p . It will be convenient to define the daughter parton momenta by

$$\zeta_p \equiv (\hat{p}_\ell, \hat{p}_j) \tag{2.34}$$

Hence, the momentum mapping from the m to the m + 1 particle phase space is given by a transformation⁵

$$\{\hat{p}, \hat{f}\}_{m+1} = R_{\ell}(\{\{p, f\}_m, \{\zeta_p, \zeta_f\}\})$$
(2.35)

where f denotes the flavour of each parton: $f \in \{g, u, \bar{u}, d, \bar{d}, \dots\}$. Here, the splitting variable ζ_f is given by the flavours of the daughter partons, so we have

$$\zeta_f = (\hat{f}_\ell, \hat{f}_j) \tag{2.36}$$

⁵More precisely, after the splitting $\{\hat{p}, \hat{f}\}_{m+1}$ is determined from $\{p, f\}_m$ and splitting variable $\{\zeta_p, \zeta_f\}$, where ζ_p denotes the daughter momenta and ζ_f denotes the daughter flavours.

The flavours of the spectator partons remain unchanged

$$\hat{f}_n = f_n, \quad n \notin \{\ell, j = m + 1\}$$
(2.37)

while the flavour of the mother parton f_{ℓ} obeys

$$\hat{f}_{\ell} + \hat{f}_j = f_{\ell} \tag{2.38}$$

e.g. if the mother parton ℓ is a quark/antiquark, then we have $\zeta_f = (q/\bar{q}, g)$. If the mother parton ℓ is a gluon, then ζ_f can be a pair of gluons $\zeta_f = (g, g)$, which corresponds to $g \to g g$ splitting, or any choice of quark/antiquark flavours $\zeta_f = (q, \bar{q})$, which corresponds to $g \to q \bar{q}$ splitting⁶.

Parton joining

There is an inverse transformation of Eq. (2.35), which maps the (m + 1)-parton momenta to the *m*-parton momenta. So we start with $\{\hat{p}\}_{m+1}$ and determine $\{p\}_m$ and $\{\zeta_p\}$. The splitting variable for the momenta is given by the momenta of the daughter partons $\zeta_p \equiv (\hat{p}_{\ell}, \hat{p}_j)$. Let Qbe the total momentum of the final state partons:

$$Q = \sum_{n=1}^{m+1} \hat{p}_n = \hat{p}_a + \hat{p}_b$$
(2.39)

One can determine the lightlike momentum p_{ℓ} by rearranging Eq. (2.22)

$$p_{\ell} = \frac{1}{\lambda} \left(\hat{p}_{\ell} + \hat{p}_j \right) - \frac{1 - \lambda + y}{2 \lambda a_{\ell}} Q$$
(2.40)

Here, the parameter a_{ℓ} still depends on the mother parton momentum p_{ℓ} , which can be mapped into her daughters \hat{p}_{ℓ} and \hat{p}_{j} through Eq. (2.32). Now we need the inverse Lorentz transformation to Eq. (2.23). All non-emitting final state spectators are mapped using the following Lorentz transformation. From $K = Q - p_{\ell}$ and $\hat{K} = Q - P_{\ell}$, we have

$$p_n^{\mu} = \Lambda(K, \bar{K})^{\mu}{}_{\nu} \, \hat{p}_n^{\nu}, \quad n \notin \{\ell, j = m+1\}$$
(2.41)

where $\Lambda(K, \hat{K})^{\mu}{}_{\nu}$ can be obtained using Eq. (2.27) with \hat{K} and K interchanged. Now the inverse transformation from the m + 1 to the m particle phase space is given by

$$\{\{p, f\}_m, \{\zeta_p, \zeta_f\}\} = Q_\ell(\{\hat{p}, \hat{f}\}_{m+1})$$
(2.42)

⁶For final state $q \to qg$ or $\bar{q} \to \bar{q}g$ splittings, we use j for the label of the gluon. For a final state $g \to q\bar{q}$ splitting, we use j for the label of the \bar{q} .

The transformation of the flavours is similar to the case of parton splitting. The splitting variable ζ_f is given by the flavours of the daughter partons $\zeta_f = (\hat{f}_\ell, \hat{f}_j)$. The flavour of the mother parton f_ℓ is given by

$$f_{\ell} = \hat{f}_{\ell} + \hat{f}_j \tag{2.43}$$

with the rule of adding flavours, q + g = q and $q + \bar{q} = g$. The flavours of the spectators remain unchanged

$$f_n = \hat{f}_n, \quad n \notin \{\ell, j = m + 1\}$$
 (2.44)

In summary, it is the transformation R_{ℓ} that is needed to describe a parton shower, while Q_{ℓ} is needed to describe the NLO QCD calculations.

The integration measure for final state splitting

In order to calculate the various splitting processes and extract the correct singularities in ϵ , we need to know the factorization of the phase space from m + 1 to m partons

$$\int d\{\hat{p}, \hat{f}\}_{m+1} g(\{\hat{p}, \hat{f}\}_{m+1}) = \int d\{p, f\}_m d\zeta_p g(\{\hat{p}, \hat{f}\}_{m+1})$$
(2.45)

where $g({\hat{p}, \hat{f}}_{m+1})$ is an arbitrary function. The definition of the unresolved one parton integration measure is [28]

$$d\zeta_{p} = dy \,\theta(y_{\min} < y < y_{\max}) \,\lambda^{d-3} \,\frac{p_{\ell} \cdot Q}{\pi} \,\frac{d^{d} \hat{p}_{\ell}}{(2 \,\pi)^{d}} \,2 \,\pi \,\delta^{+}(\hat{p}_{\ell}^{2}) \,\frac{d^{d} \hat{p}_{j}}{(2 \,\pi)^{d}} \,2 \,\pi \,\delta^{+}(\hat{p}_{j}^{2}) \\ \times (2 \,\pi)^{d} \,\delta^{d} \left(\hat{p}_{\ell} + \hat{p}_{j} - \lambda \,p_{\ell} - \frac{1 - \lambda + y}{2 \,a_{\ell}} Q\right)$$
(2.46)

Here, $y_{\min} = 0$ for massless partons and y_{\max} is given by Eq. (2.31). The final expressions in terms of the integration variables (see Section 2.4.3) can be found in Section 5.2.1.

2.3.2 Splitting an initial state parton

Parton splitting

For an initial state splitting, we follow the backwards evolution description that an initial state daughter parton ℓ splits into an initial state mother parton that enters the hard part of the process and a final state daughter parton j when going forward in time, *i.e.* $\hat{p}_a \rightarrow p_a + \hat{p}_j$. For simplicity we have chosen the convention that

$$\ell = a \tag{2.47}$$
For the case $\ell = b$, we just simply interchange $a \leftrightarrow b$. Here the final state daughter parton j is in $\{\hat{p}_1, \hat{p}_2, \dots, \hat{p}_{m+1}\}$ configuration. In this subsection, we will describe how $\{\hat{p}\}_{m+1}$ is determined from $\{p\}_m$ and a splitting variable $\{\zeta_p\}$. To describe the initial state partons momenta, we start by assuming the incoming hadron momenta p_A and p_B to be massless, so

$$p_A^2 = p_B^2 = 0$$
 and $2 p_A \cdot p_B = s$ (2.48)

where s is the center-of-mass energy of incoming hadrons. We will restrict our expressions to massless partons in the kinematics, so $p_a^2 = p_b^2 = \hat{p}_a^2 = \hat{p}_j^2 = 0$, where

$$p_{a} = \eta_{a} p_{A}$$

$$p_{b} = \eta_{b} p_{B}$$

$$\hat{p}_{a} = \hat{\eta}_{a} p_{A}$$
(2.49)

Here, η_a and η_b are momentum fractions of the incoming partons before the splitting and $\hat{\eta}_a$ is the momentum fraction after the splitting. We choose the momentum fraction of parton b to remain unaltered

$$\hat{\eta}_b = \eta_b \tag{2.50}$$

The momentum fraction $\hat{\eta}_a$ after the splitting will be determined by the final-state daughter parton \hat{p}_j . As in the case of a final-state splitting, the relation $\hat{p}_a \approx p_a + \hat{p}_j$ cannot hold, as the momenta after parton splitting should be on-shell. In order to allow all partons to be on-shell, while conserving momentum, we choose to take the needed momenta from the final state spectator partons. This can be achieved by a Lorentz transformation as in the case of a final state splitting

$$\hat{p}_{n}^{\mu} = \Lambda(K, K)^{\mu}{}_{\nu} p_{n}^{\nu}, \quad n \in \{1, \cdots, m\} \text{ and } n \neq j$$
 (2.51)

Here, K is the total momentum of the final state spectators before the splitting

$$K = p_a + p_b \tag{2.52}$$

and \hat{K} is the total momentum of the final state spectators after the splitting

$$\hat{K} = \hat{p}_a + p_b - \hat{p}_j = \hat{Q} - \hat{p}_j, \quad \hat{Q} = \hat{p}_a + p_b$$
 (2.53)

Since each final state spectator is changed by a Lorentz transformation, we have

$$\hat{K}^{\mu} = \Lambda^{\mu}{}_{\nu} K^{\nu} \tag{2.54}$$

with the Lorentz transformation given by Eq. (2.27). In order that K is related to \hat{K} by a Lorentz transformation, we need $K^2 = \hat{K}^2$. Hence, we get

$$K^2 = \alpha \eta_a = \tilde{K}^2 \tag{2.55}$$

where

$$\alpha = \eta_b s \tag{2.56}$$

From this it follows that

$$\eta_a \eta_b s = (\hat{p}_a + p_b - \hat{p}_j)^2 = \hat{\eta}_a \eta_b s - 2 [\hat{\eta}_a p_A \cdot \hat{p}_j + \eta_b p_B \cdot \hat{p}_j]$$
(2.57)

which defines

$$\hat{\eta}_{a} = \frac{\eta_{a} \eta_{b} s + 2 \eta_{b} p_{B} \cdot \hat{p}_{j}}{\eta_{b} s - 2 p_{A} \cdot \hat{p}_{j}} \le 1$$
(2.58)

A consequence of Eq. (2.58) is

$$\eta_a < \hat{\eta}_a < 1 \tag{2.59}$$

Again, we can introduce the splitting variable ζ_p that defines the momenta of the daughter partons:

$$\zeta_p \equiv (\hat{p}_a, \hat{p}_j) \tag{2.60}$$

so that $\{p\}_m$ together with ζ_p determines $\{\hat{p}\}_{m+1}$. There is a transformation R_a relating m to the m+1 particle phase space,

$$\{\hat{p}, \hat{f}\}_{m+1} = R_a(\{\{p, f\}_m, \{\zeta_p, \zeta_f\}\})$$
(2.61)

Here, the splitting variable ζ_f is given by the flavours of the daughter partons, so we have

$$\zeta_f = (\hat{f}_a, \hat{f}_j) \tag{2.62}$$

The flavours of the spectator partons remain unchanged

$$\hat{f}_n = f_n, \quad n \notin \{a, j = m+1\}$$
 (2.63)

while the flavours of the daughter partons \hat{f}_a and \hat{f}_j obey

$$\hat{f}_a + \hat{f}_j = f_a \tag{2.64}$$

Parton joining

Now we consider the inverse transformation needed in the subtraction, meaning that we combine an initial-state parton with a final-state parton into a mother parton that enters the hard part. So we start with $\{\hat{p}\}_{m+1}$ and determine $\{p\}_m$ and $\{\zeta_p\}$.

The splitting variable for the momenta is given by the momenta of the daughter partons

$$\zeta_p \equiv (\hat{p}_a, \hat{p}_j) \tag{2.65}$$

One can determine η_a by solving $K^2 = \hat{K}^2$, so we can express η_a in terms of $\hat{\eta}_a$. This gives

$$\eta_a = \frac{\hat{K}^2}{\alpha} = \frac{\hat{\eta}_a \eta_b s - 2 \left[\hat{\eta}_a p_A \cdot \hat{p}_j + \eta_b p_B \cdot \hat{p}_j\right]}{\eta_b s}$$
(2.66)

Once we have η_a , we can construct $K = p_a + p_b$. Further we need the inverse Lorentz transformation of Eq. (2.51). From K and \hat{K} , we have

$$p_n^{\mu} = \Lambda(K, \hat{K})^{\mu}{}_{\nu} \hat{p}_n^{\nu}, \quad n \in \{1, \cdots, m\} \text{ and } n \neq j$$
 (2.67)

where $\Lambda(K, \hat{K})^{\mu}{}_{\nu}$ can be obtained using Eq. (2.27) with \hat{K} and K interchanged. The mapping for the initial state parton a is

$$p_a = \left(1 - \frac{\hat{p}_j \cdot \hat{Q}}{\hat{p}_a \cdot p_b}\right) \hat{p}_a \tag{2.68}$$

Now the inverse transformation from the (m + 1)- to the *m*-particle phase space is given by

$$\{\{p, f\}_m, \{\zeta_p, \zeta_f\}\} = Q_a\left(\{\hat{p}, \hat{f}\}_{m+1}\right)$$
(2.69)

The transformation of the flavours is similar to the case of parton splitting. The splitting variable ζ_f is given by the flavours of the daughter partons $\zeta_f = (\hat{f}_a, \hat{f}_j)$ in the sense of backwards evolution. The flavour of the mother parton f_a is given by

$$f_a = f_a + f_j \tag{2.70}$$

The flavours of the spectators remain unchanged

$$f_n = \hat{f}_n, \quad n \notin \{a, j = m+1\}$$
 (2.71)

The integration measure for initial state splitting

The phase space factorization from m + 1 to m partons takes a similar form as in the final state splitting, *i.e.* we have again

$$\int d\{\hat{p}, \hat{f}\}_{m+1} g(\{\hat{p}, \hat{f}\}_{m+1}) = \int d\{p, f\}_m d\zeta_p g(\{\hat{p}, \hat{f}\}_{m+1})$$
(2.72)

where $g({\hat{p}, \hat{f}}_{m+1})$ is an arbitrary function. The definition of the unresolved one parton integration measure is [28]

$$d\zeta_p = \frac{d^d \hat{p}_j}{(2\pi)^d} 2\pi \,\delta^+(\hat{p}_j^2) \frac{\alpha}{\hat{\alpha}}$$
(2.73)

Here, $\hat{\alpha} = \eta_b s - 2 p_A \cdot \hat{p}_j$. The factor $\alpha/\hat{\alpha}$ is just the derivative $d\hat{\eta}_a/d\eta_a$ calculated from the relation $\hat{K}^2 = K^2$. The final expressions in terms of the integration variables (see Section 2.4.4) can be found in Section 5.2.2.

2.4 Splitting functions

In the Nagy-Soper dipole subtraction scheme, the dipoles are based on splitting functions [28,29] that will be used to generate the shower. The new subtraction scheme also allows for easy matching with a parton shower using the same splitting functions in the spin-averaged approximation. In addition, the use of the dipoles as splitting functions in the shower, when combined with NLO calculations, simplifies the treatment of double counting.

Consider for a moment the QCD scattering amplitude without spin, colour or flavours. In the singular limit the transition amplitude factorizes according to

$$\mathcal{M}_{m+1}(\{\hat{p}\}_{m+1}) > \approx v_{\ell}(\{\hat{p}\}_{m+1}) \cdot \mid \mathcal{M}_{m}(\{p\}_{m}) >$$
(2.74)

where $v_{\ell}(\{\hat{p}\}_{m+1})$ is the splitting amplitude proportional to $1/\hat{p}_{\ell} \cdot \hat{p}_j$. The momentum mapping between $\{\hat{p}\}_{m+1}$ and $\{p\}_m$ is described in Section 2.3. In reality, we have soft as well as collinear singularities in QCD, we also have to consider spin, colour and parton flavours. We will describe splitting amplitudes in more detail including all these factors in the next subsection.

It has been known, that scattering amplitudes can be factorized out in a general way by using the factorization properties of QCD amplitudes in the soft and collinear limits [31–35]. At the next-to-next-to-leading order and beyond, the factorization of QCD scattering amplitudes can be found in [36].

2.4.1 Definition of the splitting amplitudes

The splitting functions described in [28,29] are based on the spin dependent splitting amplitudes. The QCD scattering amplitude for m + 1 partons is a vector in colour \otimes spin space.

$$|\mathcal{M}(\{\hat{p},\hat{f}\}_{m+1})>$$
 (2.75)

when two partons ℓ and j are almost collinear, this amplitude becomes

$$|\mathcal{M}(\{\hat{p}, \hat{f}\}_{m+1}) > \sim |\mathcal{M}_{\ell}(\{\hat{p}, \hat{f}\}_{m+1}) >$$
(2.76)

In the limit, that parton j becomes soft, then all of the $|\mathcal{M}_{\ell}(\{\hat{p}, \hat{f}\}_{m+1}) >$ amplitudes contribute, and we have $(|\mathcal{M}_{\ell}\rangle)$ is to be defined in Eq. (2.78).

$$|\mathcal{M}(\{\hat{p}, \hat{f}\}_{m+1}) > \sim \sum_{\ell} |\mathcal{M}_{\ell}(\{\hat{p}, \hat{f}\}_{m+1}) >$$
(2.77)

After splitting parton ℓ , the amplitude $| \mathcal{M}_{\ell}(\{\hat{p}, \hat{f}\}_{m+1}) >$ can be factorized into a splitting operator times the *m*-parton matrix element according to

$$|\mathcal{M}_{\ell}(\{\hat{p},\hat{f}\}_{m+1})\rangle = t_{\ell}^{\dagger}(f_{\ell} \to \hat{f}_{\ell} + \hat{f}_{j}) V_{\ell}^{\dagger}(\{\hat{p},\hat{f}\}_{m+1}) |\mathcal{M}(\{p,f\}_{m})\rangle$$
(2.78)

Here, the Born amplitude for producing *m* partons is evaluated at momenta and flavours $\{p, f\}_m$ determined from $\{\hat{p}, \hat{f}\}_{m+1}$ according to the transformation $Q_{\ell}(\{\hat{p}, \hat{f}\}_{m+1})$ in Eq. (2.42) or Eq. (2.69). $V_{\ell}^{\dagger}(\{\hat{p}, \hat{f}\}_{m+1})$ is an operator acting on the spin part of the colour \otimes spin space, while $t_{\ell}^{\dagger}(f_{\ell} \rightarrow \hat{f}_{\ell} + \hat{f}_{j})$ is an operator acting on the colour part of the colour \otimes spin space. The spin-dependent splitting operator can be described in the spin space $|\{s\}_m >$,

$$<\{\hat{s}\}_{m+1} \mid V_{\ell}^{\dagger}(\{\hat{p},\hat{f}\}_{m+1}) \mid \{s\}_{m} >$$
(2.79)

we can take Eq. (2.79) to be diagonal

$$<\{\hat{s}\}_{m+1} \mid V_{\ell}^{\dagger}(\{\hat{p},\hat{f}\}_{m+1}) \mid \{s\}_{m} > = \left(\prod_{n \notin \{\ell,j=m+1\}} \delta_{\hat{s}_{n},s_{n}}\right) v_{\ell}(\{\hat{p},\hat{f}\}_{m+1},\hat{s}_{j},\hat{s}_{\ell},s_{\ell})$$
(2.80)

Here, the splitting amplitudes $v_{\ell}(\{\hat{p}, \hat{f}\}_{m+1}, \hat{s}_j, \hat{s}_\ell, s_\ell)$ can be derived from the QCD vertices, spinors and polarization vectors for on-shell partons. We can use the mother momentum p_{ℓ} to describe the splitting amplitudes v_{ℓ} , and the relation between p_{ℓ} and $\{\hat{p}, \hat{f}\}_{m+1}$ is given by the transformation $Q_{\ell}(\{\hat{p}, \hat{f}\}_{m+1})$.

In the following, we will illustrate splitting amplitudes by giving some examples. First we consider the case of a $g \rightarrow g g$ splitting. The ggg QCD vertex is given by

$$v^{\alpha\beta\gamma}(p_{a}, p_{b}, p_{c}) = g^{\alpha\beta}(p_{a} - p_{b})^{\gamma} + g^{\beta\gamma}(p_{b} - p_{c})^{\alpha} + g^{\gamma\alpha}(p_{c} - p_{a})^{\beta}$$
(2.81)

In order to define the propagator for the gluon field properly, we have to make a choice of gauge. The choice

$$\mathcal{L}_{\text{gauge-fixing}} = -\frac{1}{2\lambda} \left(n^{\mu} G^{A}_{\mu} \right)^{2}$$
(2.82)

is called the axial gauge, in which we introduce an additional vector n. Here the parameter λ will break the gauge invariance of the theory, however, the physical results will be independent of λ at the end. The advantage of the axial gauge is that ghost fields are not required. Now we consider a special case in which $\lambda = 0$ and $n^2 = 0$. This is called the light-cone gauge. We define

$$D^{\mu\nu}(P,n) = -g^{\mu\nu} + \frac{P^{\mu} n^{\nu} + n^{\mu} P^{\nu}}{P \cdot n}$$
(2.83)

In the limit $P^2 \rightarrow 0$ we have

$$n^{\mu} D_{\mu\nu}(P,n) = 0, \qquad P^{\mu} D_{\mu\nu}(P,n) = 0$$
 (2.84)

In the axial gauge $n \cdot G^A = 0$, only two physical polarization states propagate, which are orthogonal to n and P. Hence we can use the propagator for an off-shell gluon

$$\frac{D^{\mu\nu}(\hat{p}_{\ell} + \hat{p}_j, n_{\ell})}{(\hat{p}_{\ell} + \hat{p}_j)^2}$$
(2.85)

where the lightlike vector n_{ℓ} is chosen to lie in the plane of p_{ℓ} and Q (see [28, 29]),

$$n_{\ell} = Q - a_{\ell} p_{\ell} = \frac{1 + y + \lambda}{2\lambda} Q - \frac{a_{\ell}}{\lambda} (\hat{p}_{\ell} + \hat{p}_{j})$$
(2.86)

for the final state splitting. More generally, the vector n_{ℓ} has the following form for both initial and final state splittings,

$$n_{\ell} = \begin{cases} p_{B} & \text{for } \ell = a \\ p_{A} & \text{for } \ell = b \\ Q - a_{\ell} p_{\ell} & \text{for } \ell \in \{1, \cdots, m\} \end{cases}$$
(2.87)

In the case of a final-state $g \rightarrow g g$ splitting (Fig. 2.3), we derive

$$v_{\ell}(\{\hat{p},\hat{f}\}_{m+1},\hat{s}_{j},\hat{s}_{\ell},s_{\ell}) = \frac{\sqrt{4\pi\alpha_{s}}}{2\,\hat{p}_{j}\cdot\hat{p}_{\ell}}\,\varepsilon_{\alpha}(\hat{p}_{j},\hat{s}_{j},\hat{Q})^{*}\,\varepsilon_{\beta}(\hat{p}_{\ell},\hat{s}_{\ell},\hat{Q})^{*}\varepsilon^{\nu}(p_{\ell},s_{\ell},\hat{Q}) \\ \times v^{\alpha\beta\gamma}(\hat{p}_{j},\hat{p}_{\ell},-\hat{p}_{j}-\hat{p}_{\ell})\,D_{\gamma\nu}(\hat{p}_{\ell}+\hat{p}_{j},n_{\ell})$$
(2.88)

In the case of an initial-state $g \rightarrow g g$ splitting (Fig. 2.4), we have

$$v_{\ell}(\{\hat{p},\hat{f}\}_{m+1},\hat{s}_{j},\hat{s}_{\ell},s_{\ell}) = -\frac{\sqrt{4\pi\alpha_{s}}}{2\,\hat{p}_{j}\cdot\hat{p}_{\ell}}\varepsilon_{\alpha}(\hat{p}_{j},\hat{s}_{j},\hat{Q})^{*}\varepsilon_{\beta}(\hat{p}_{\ell},\hat{s}_{\ell},\hat{Q})\varepsilon^{\nu}(p_{\ell},s_{\ell},\hat{Q})^{*} \times v^{\alpha\beta\gamma}(\hat{p}_{j},-\hat{p}_{\ell},\hat{p}_{\ell}-\hat{p}_{j}) D_{\gamma\nu}(\hat{p}_{\ell}-\hat{p}_{j},n_{\ell})$$
(2.89)

Here we have the exact QCD vertex and the off-shell gluon propagator in the axial gauge $n \cdot G^A = 0$ for both initial- and final-state ggg splittings. In order for the mother gluon to be on-shell, we can make an approximation so that the mother gluon is projected onto the physical degrees of freedom as it emerges from the hard matrix element. This projection is contained in the on-shell polarization vector $\varepsilon(p_\ell, s_\ell, \hat{Q})$.

For a final-state $q \rightarrow q g$ splitting (Fig. 2.5), we derive

$$v_{\ell}(\{\hat{p},\hat{f}\}_{m+1},\hat{s}_{j},\hat{s}_{\ell},s_{\ell}) = \sqrt{4\pi\alpha_{s}} \varepsilon_{\mu}(\hat{p}_{j},\hat{s}_{j},\hat{Q})^{*} \frac{\overline{U}(\hat{p}_{\ell},\hat{s}_{\ell}) \gamma^{\mu} (\not\!\!\!p_{\ell}+\not\!\!\!p_{j}) \not\!\!\!p_{\ell} U(p_{\ell},s_{\ell})}{(\hat{p}_{\ell}+\hat{p}_{j})^{2} 2p_{\ell} \cdot n_{\ell}}$$
(2.90)

For an initial-state $q \rightarrow q g$ splitting (Fig. 2.6), we have

$$v_{\ell}(\{\hat{p},\hat{f}\}_{m+1},\hat{s}_{j},\hat{s}_{\ell},s_{\ell}) = -\sqrt{4\pi\alpha_{s}} \varepsilon_{\mu}(\hat{p}_{j},\hat{s}_{j},\hat{Q})^{*} \frac{\overline{U}(p_{\ell},s_{\ell}) \not h_{\ell} \left(\not p_{\ell}-\not p_{j}\right) \gamma^{\mu} U(\hat{p}_{\ell},\hat{s}_{\ell})}{(\hat{p}_{\ell}-\hat{p}_{j})^{2} 2p_{\ell} \cdot n_{\ell}}$$
(2.91)

Here, \overline{U} and U denote Dirac spinors for the quark fields satisfying $(\not q - m) U(\vec{q}, s) = 0$ and $\overline{U}(\vec{q}, s) (\not q - m) = 0$. The Dirac spinors for the antiquark fields obey $(\not q + m) V(\vec{q}, s) = 0$ and $\overline{V}(\vec{q}, s) (\not q + m) = 0$. They are normalized to

$$\overline{U}(\vec{p},s) \gamma^{\mu} U(\vec{p},s) = 2 p^{\mu},$$

$$\overline{V}(\vec{p},s) \gamma^{\mu} V(\vec{p},s) = 2 p^{\mu}$$
(2.92)

The projection operators are

$$\not q + m = \sum_{s} U(\vec{q}, s) \overline{U}(\vec{q}, s), \qquad \not q - m = \sum_{s} V(\vec{q}, s) \overline{V}(\vec{q}, s)$$
(2.93)

The polarization vector ε_{μ} for the gluon field is defined in a timelike axial gauge so that

$$\hat{p}_j \cdot \varepsilon = \hat{Q} \cdot \varepsilon = 0, \qquad \hat{Q} = Q$$
 (2.94)

For other flavour choices, the results can be found in [28]. For the splitting functions, we have to square the splitting amplitudes. In the new formalism, there are two sorts of splitting functions: direct splitting functions (Fig. 2.7) and interference splitting functions (Fig. 2.8). The direct splitting functions correspond to the scattering amplitude $| \mathcal{M}_{\ell}(\{\hat{p}, \hat{f}\}_{m+1}) >$ for a parton ℓ to split times its complex conjugate scattering amplitude $< \mathcal{M}_{\ell}(\{\hat{p}, \hat{f}\}_{m+1}) |$ for that same parton ℓ to split, while the interference splitting functions correspond to the interference between the scattering amplitude $| \mathcal{M}_{\ell}(\{\hat{p}, \hat{f}\}_{m+1}) |$ for that same parton ℓ to split, while the interference splitting functions correspond to the interference between the scattering amplitude $| \mathcal{M}_{\ell}(\{\hat{p}, \hat{f}\}_{m+1}) >$ for a parton ℓ to split into partons ℓ and j and the complex-conjugate scattering amplitude $< \mathcal{M}_k(\{\hat{p}, \hat{f}\}_{m+1}) |$ for another parton k to split into partons with labels k and j. These functions generate leading singularities when parton j is a soft gluon.

The direct splitting function is the product of a splitting amplitude v_{ℓ} and its complex-conjugate splitting amplitude v_{ℓ}^* , so we have

$$v_{\ell}(\{\hat{p},f\}_{m+1},\hat{s}_{j},\hat{s}_{\ell},s_{\ell}) v_{\ell}(\{\hat{p},f\}_{m+1},\hat{s}_{j},\hat{s}_{\ell},s_{\ell})^{*}$$
(2.95)

which, after summing over the daughter parton spins and averaging over the mother parton spins, leads to the spin averaged splitting functions

$$\overline{W}_{\ell\ell} = \frac{1}{2} \sum_{\hat{s}_{\ell}, \hat{s}_j, s_{\ell}} \left| v_{\ell}(\{\hat{p}, \hat{f}\}_{m+1}, \hat{s}_j, \hat{s}_{\ell}, s_{\ell}) \right|^2$$
(2.96)

2.4.2 Eikonal factor

When parton j is a gluon, there is a common result in the limit $\hat{p}_j \to 0$, in which the splitting amplitude v_ℓ can be replaced by the eikonal approximation,

$$v_{\ell}^{\text{eikonal}}(\{\hat{p}, \hat{f}\}_{m+1}, \hat{s}_j, \hat{s}_{\ell}, s_{\ell}) = \sqrt{4\pi\alpha_s} \,\delta_{\hat{s}_{\ell}, s_{\ell}} \,\frac{\varepsilon(\hat{p}_j, \hat{s}_j, Q)^* \cdot \hat{p}_{\ell}}{\hat{p}_j \cdot \hat{p}_{\ell}}$$
(2.97)

The eikonal approximation of the spin-averaged splitting functions $\overline{W}_{\ell\ell}$ is then

$$\overline{W}_{\ell\ell}^{\text{eikonal}} = 4 \pi \alpha_s \frac{\hat{p}_\ell \cdot D(\hat{p}_j, Q) \cdot \hat{p}_\ell}{(\hat{p}_j \cdot \hat{p}_\ell)^2}$$
(2.98)



Figure 2.3: A final-state ggg splitting.



Figure 2.4: An initial-state ggg splitting.



Figure 2.5: A final-state qqg splitting.



Figure 2.6: An initial-state qqg splitting.



Figure 2.7: Collinear diagram: parton j is emitted from parton ℓ in the scattering amplitude $| \mathcal{M}_{\ell}(\{\hat{p}, \hat{f}\}_{m+1}) >$ and parton j is emitted from that same parton ℓ in the complex-conjugate scattering amplitude $< \mathcal{M}_{\ell}(\{\hat{p}, \hat{f}\}_{m+1}) |$.



Figure 2.8: Soft diagram: parton j is emitted from parton ℓ in the scattering amplitude and parton j is emitted from parton k in the complex-conjugate scattering amplitude.

Here, $D_{\mu\nu}$ is the transverse projection tensor for an off-shell gluon, which is given by

$$D_{\mu\nu}(\hat{p}_j, \hat{Q}) = -g_{\mu\nu} + \frac{\hat{p}_j^{\mu} \, \hat{Q}^{\nu} + \hat{Q}^{\mu} \, \hat{p}_j^{\nu}}{\hat{p}_j \cdot \hat{Q}} - \frac{\hat{Q}^2 \, \hat{p}_j^{\mu} \, \hat{p}_j^{\nu}}{(\hat{p}_j \cdot \hat{Q})^2}$$
(2.99)

It will be convenient if we define a dimensionless function F:

$$F = \frac{\hat{p}_{\ell} \cdot \hat{p}_j}{4 \pi \alpha_s} \overline{W}_{\ell\ell}$$
(2.100)

then as the gluon j becomes soft, $\hat{p}_j \to 0$, we can define the function F_{eik} :

$$F_{\text{eik}} = \frac{\hat{p}_{\ell} \cdot \hat{p}_j}{4 \pi \alpha_s} \overline{W}_{\ell\ell}^{\text{eikonal}} = \frac{\hat{p}_{\ell} \cdot D(\hat{p}_j, \hat{Q}) \cdot \hat{p}_{\ell}}{\hat{p}_{\ell} \cdot \hat{p}_j} = \frac{2 \, \hat{p}_{\ell} \cdot Q}{\hat{p}_j \cdot Q} - \frac{Q^2 \, \hat{p}_{\ell} \cdot \hat{p}_j}{(\hat{p}_j \cdot Q)^2} \tag{2.101}$$

We can also introduce the notation v_{eik}^2 or \bar{v}_{eik}^2 , which we will use throughout in the following discussions, and which is defined through the spin-averaged splitting function $\overline{W}_{\ell\ell}^{\text{eikonal}}$

$$\overline{W}_{\ell\ell}^{\text{eikonal}} = \overline{v}_{\text{eik}}^2 = \begin{cases} \frac{1}{2} v_{\text{eik}}^2 & \text{for quarks} \\ \\ \frac{1}{2(1-\epsilon)} v_{\text{eik}}^2 & \text{for gluons} \end{cases}$$
(2.102)

The prefactor 1/2 is the quark spin (or helicity) degrees of freedom, while the factor $1/2/(1-\epsilon)$ is the gluon spin degrees of freedom. More precisely, if we consider both the colour and spin average, we can introduce a notation $1/\omega(a)$, which is defined by

$$\omega(a) = \begin{cases} 2N & \text{if } a = q \\ 2(1-\epsilon)D_A & \text{if } a = g \end{cases}$$
(2.103)

 $\omega(a)$ is the number of colour and spin degrees of freedom for the flavour a. $D_A = N^2 - 1$ is the dimension of the adjoint representation of the SU(N) colour group, while N is the dimension of the fundamental representation of the colour group, e.g. for $SU(3)_C$ we have $D_A = 8$ colour degrees of freedom of the gluon field in the adjoint representation and N = 3 colour degrees of freedom of the quark field in the fundamental representation of $SU(3)_C$. In principle we also have to consider δ^{ab} or δ^{AB} (which equals 3 or 8) in the quark or gluon propagators, where the indices a, b = 1, 2, 3 and $A, B = 1, \dots, 8$. In practice all these factors are included in the matrix element.

In Eq. (2.102), v_{eik}^2 denotes the spin-unaveraged splitting function. So we have

$$v_{\text{eik}}^{2} = 4\pi \alpha_{s} \times \begin{cases} \left[2 / \left(\hat{p}_{\ell} \cdot \hat{p}_{j} \right)^{2} \right] \hat{p}_{\ell} \cdot D(\hat{p}_{j}, \hat{Q}) \cdot \hat{p}_{\ell} & \text{for quarks} \\ \\ \left[2 \left(1 - \epsilon \right) / \left(\hat{p}_{\ell} \cdot \hat{p}_{j} \right)^{2} \right] \hat{p}_{\ell} \cdot D(\hat{p}_{j}, \hat{Q}) \cdot \hat{p}_{\ell} & \text{for gluons} \end{cases}$$
(2.104)

For the interference (soft) term, we obtain

$$\Delta W = \overline{W}_{\ell\ell}^{\text{eikonal}} - \overline{W}_{\ell k} = \begin{cases} \frac{1}{2} \left(v_{\text{eik}}^2 - v_{\text{soft}}^2 \right) & \text{for quarks} \\ \frac{1}{2\left(1-\epsilon\right)} v_{\text{eik}}^2 - \frac{1}{2} v_{\text{soft}}^2 & \text{for gluons} \end{cases}$$
(2.105)

where the soft term is given by $\overline{W}_{\ell k} = 4 \pi \alpha_s 2 A_{\ell k} \hat{p}_{\ell} \cdot D(\hat{p}_j, \hat{Q}) \cdot \hat{p}_k / (\hat{p}_j \cdot \hat{p}_{\ell} \hat{p}_j \cdot \hat{p}_k)$ and $A_{\ell k}$ is to be defined precisely in Eq. (2.162) (see also Section 2.4.5).

2.4.3 Collinear splitting functions: final state splittings

In this and the next subsections, we will give the explicit expressions of splitting function for each process. In the following, we will remove the common factor $4\pi\alpha_s$, which we will add it back in the end. We also ignore the colour factors for a moment, which we will include when we consider the integrated splitting functions. For the splitting functions in (m + 1)-parton phase spaces we always work in four dimensions, meaning that we can safely put d = 4 (or $\epsilon = 0$) in the following discussions. We will also include the eikonal factor for the collinear splitting functions if the emitted parton j is a gluon. The eikonal splitting function will turn out to be important when we incorporate the soft gluon interference diagrams. We will explain this in more detail in Section 2.4.5.

Kinematics: integration variables

We can introduce the two integration variables

$$x = \frac{\hat{p}_j \cdot Q}{P_\ell \cdot Q}, \qquad z = \frac{\hat{p}_j \cdot n_\ell}{P_\ell \cdot n_\ell}$$
(2.106)

Hence we could express most dot products in terms of $x, y, z, \lambda, a_{\ell}$. The parameters a_{ℓ}, λ and y have been given in Section 2.3. Note, however, that these are not all independent variables; a_{ℓ} depends on the kinematics before splitting/after recombination respectively, and $\lambda = \lambda(a_{\ell}, y)$. The variable x is given by

$$x = \frac{\lambda}{1+y} z + \frac{2 a_{\ell} y}{(1+y) (1+y+\lambda)}$$
(2.107)

So we are left with a_{ℓ} , y, z as free variables; we will use y and z in the integration. If we want to eliminate the x dependence and go back to y and z as integration variables and keep a_{ℓ} as a fixed parameter coming from the m-particle phase space, we can introduce

$$x_0 = \frac{1 - \lambda + y}{1 + \lambda + y} = x_0(a_\ell, y)$$
(2.108)

Using

$$\gamma = \frac{1}{2} \left(1 + y + \lambda \right) \tag{2.109}$$

we obtain a couple of useful relations, e.g.

$$x_{0} = \frac{y a_{\ell}}{\gamma^{2}}, \quad 1 - x_{0} = \frac{\lambda}{\gamma}, \quad 1 + x_{0} = \frac{1 + y}{\gamma}$$
$$x (1 + y) = \gamma [x_{0} + z (1 - x_{0})], \quad (1 - x) (1 + y) = \gamma [1 - z (1 - x_{0})] \quad (2.110)$$

When integrating the interference terms, we need to make use of an additional integration variable parametrizing the angle between emitter and spectator. To parametrize this in a Lorentz invariant way, we introduce the variables

$$v = \frac{\hat{p}_j \cdot \hat{p}_k}{\hat{p}_k \cdot P_\ell}, \qquad \bar{z} = \frac{\hat{p}_k \cdot n_\ell \, \hat{p}_\ell \cdot \hat{p}_j}{\hat{p}_k \cdot P_\ell \, n_\ell \cdot P_\ell} = \frac{y}{\gamma} \frac{\hat{p}_k \cdot n_\ell}{\hat{p}_k \cdot P_\ell} = \frac{y}{\gamma} \tilde{a}$$
(2.111)

and

$$\hat{p}_k \cdot Q = \frac{\hat{p}_k \cdot P_\ell}{\gamma} \left(\lambda \,\tilde{a} + a_\ell\right) \tag{2.112}$$

with

$$\tilde{a} = \tilde{a}(y) = \frac{\hat{p}_k \cdot n_\ell}{\hat{p}_k \cdot P_\ell}$$
(2.113)

It is convenient to introduce the angles θ , θ_k and φ in the integration measure such that in the center of mass system we have⁷

$$\hat{p}_{j} = A_{j} \begin{pmatrix} 1\\ \sin\theta\cos\varphi\\ \sin\theta\sin\varphi\\ \cos\theta \end{pmatrix}, \qquad \hat{p}_{k} = A_{k} \begin{pmatrix} 1\\ \sin\theta_{k}\\ 0\\ \cos\theta_{k} \end{pmatrix}, \qquad P_{\ell} = P \begin{pmatrix} 1\\ 0\\ 0\\ 0\\ 0 \end{pmatrix}, \quad A_{j} = \frac{P}{2}$$
(2.114)

In the integration measure, we will need an additional change of variable,

$$v' = \frac{1}{2} (1 - \cos \varphi) = \frac{v - v_{\min}}{v_{\max} - v_{\min}}$$
(2.115)

where

$$v = \frac{1}{2} (1 - \sin \theta \sin \theta_k \cos \varphi - \cos \theta \cos \theta_k)$$

$$v_{\text{max}} = \frac{1}{2} (1 + \sin \theta \sin \theta_k - \cos \theta \cos \theta_k)$$

$$v_{\text{min}} = \frac{1}{2} (1 - \sin \theta \sin \theta_k - \cos \theta \cos \theta_k)$$
(2.116)

⁷In the reference frame where P_{ℓ} is at rest (center of mass frame), we have $\hat{p}_j = (p_j^0, \vec{p}_j) = (p_j^0, p_j^0 \hat{u})$ and $\hat{p}_{\ell} = (p_{\ell}^0, -\vec{p}_j) = (p_{\ell}^0, -p_j^0 \hat{u}) = (p_j^0, -p_j^0 \hat{u})$. Hence, the 4-vectors \hat{p}_j and \hat{p}_{ℓ} have the same energy content in the massless case: $A_j = p_j^0 = p_{\ell}^0 = P/2$. Here, the unit vector is $\hat{u} = \sin \theta \cos \varphi \, \hat{x} + \sin \theta \sin \varphi \, \hat{y} + \cos \theta \, \hat{z}$.

We can also write \bar{z} and z in terms of the angles θ_k and θ ,

$$\bar{z} = \frac{1}{2}(1 - \cos\theta_k), \qquad z = \frac{1}{2}(1 - \cos\theta)$$
 (2.117)

Finally we give a list of useful relations for dot products of vectors, which we have used throughout the dipole calculations,

$$P_{\ell} \cdot Q = (1+y) p_{\ell} \cdot Q$$

$$\hat{p}_{j} \cdot Q = x (1+y) p_{\ell} \cdot Q$$

$$\hat{p}_{\ell} \cdot Q = (1-x) (1+y) p_{\ell} \cdot Q$$

$$P_{\ell} \cdot p_{\ell} = \frac{1+y-\lambda}{2a_{\ell}} p_{\ell} \cdot Q$$

$$P_{\ell} \cdot n_{\ell} = \frac{1+y+\lambda}{2} p_{\ell} \cdot Q$$

$$p_{\ell} \cdot \hat{p}_{j} = \frac{1}{\lambda} \left(y - \frac{1-\lambda+y}{2a_{\ell}} x (1+y) \right) p_{\ell} \cdot Q$$

$$p_{\ell} \cdot \hat{p}_{\ell} = \frac{1}{\lambda} \left(y - \frac{1-\lambda+y}{2a_{\ell}} (1-x) (1+y) \right) p_{\ell} \cdot Q$$

$$(2.118)$$

The subprocess qqg splitting

First we consider the collinear qqg splitting function in the final state. The $\bar{q}\bar{q}g$ final state splitting function is the same for massless quarks. The dipole including the eikonal splitting function is given by

$$v_{qqg}^2 - v_{\text{eik}}^2 = \frac{2}{y(p_\ell \cdot Q)} \left\{ \frac{(\lambda - 1 + y)^2 + 4y}{4\lambda} F_{\text{eik}} + \frac{d - 2}{4} z (1 + y + \lambda) \right\}$$
(2.119)

which, in terms of the momenta, can be rewritten as

$$v_{qqg}^{2} - v_{eik}^{2} = \frac{2}{\hat{p}_{\ell} \cdot \hat{p}_{j}} \left\{ \left[\left(\frac{P_{\ell} \cdot n_{\ell}}{p_{\ell} \cdot Q} - 1 \right) + \frac{a_{\ell} \, \hat{p}_{\ell} \cdot \hat{p}_{j}}{\sqrt{(p_{\ell} \cdot Q)^{2} + (\hat{p}_{\ell} \cdot \hat{p}_{j})^{2} + 2 \, p_{\ell} \cdot Q \, \hat{p}_{\ell} \cdot \hat{p}_{j} \, (1 - 2 \, a_{\ell})} \right] \\ \times \left(\frac{2 \, \hat{p}_{\ell} \cdot Q}{\hat{p}_{j} \cdot Q} - \frac{Q^{2} \, \hat{p}_{\ell} \cdot \hat{p}_{j}}{(\hat{p}_{j} \cdot Q)^{2}} \right) + 2 \, \left(\frac{d - 2}{4} \right) \, \left(\frac{\hat{p}_{j} \cdot Q - a_{\ell} \, p_{\ell} \cdot \hat{p}_{j}}{p_{\ell} \cdot Q} \right) \right\}$$
(2.120)

where

$$F_{\text{eik}} = 2 \frac{1-x}{x} - \frac{2 a_{\ell} y}{x^2 (1+y)^2}$$
(2.121)

The dipole $v_{qqg}^2 - v_{eik}^2$ denotes the spin-unaveraged splitting function. After averaging over the incoming parton spins gives an additional factor 1/2. The relation, connecting hatted and unhatted quantities, is given in Eq. (2.32). In Eq. (2.119), only the collinear singularity is left

over after subtracting the eikonal factor. Note that $\hat{Q} = Q$ for the final state splittings and $P_{\ell} \cdot n_{\ell} = (\hat{p}_{\ell} + \hat{p}_j) \cdot Q - a_{\ell} (\hat{p}_{\ell} + \hat{p}_j) \cdot p_{\ell}$ is reduced to $p_{\ell} \cdot Q$ if $a_{\ell} = 1$. It corresponds to the case in which there are only two massless partons in the final state. If we want to eliminate x and switch to x_0 and z as variables⁸, we obtain

$$F_{\text{eik}} = 2\left(-1 + \frac{1+x_0}{x_0 + z\left(1-x_0\right)} - \frac{x_0}{\left(x_0 + z\left(1-x_0\right)\right)^2}\right)$$
(2.122)

Note also that as $y \to 0$ (which means x = z), F turns into the standard Altarelli-Parisi splitting function,

$$F_{AP}(z) = \frac{1 + (1 - z)^2}{z}$$
(2.123)

The subprocess $gq\bar{q}$ splitting

Only the collinear singularity is present for the $gq\bar{q}$ splitting, so we do not subtract the eikonal splitting function in this case. Hence the dipole for the massless quarks is given by

$$v_{gq\bar{q}}^{2} = \frac{2}{y \, p_{\ell} \cdot Q} \, \left(1 - \epsilon - 2 \, z \, (1 - z)\right) \tag{2.124}$$

which, in terms of the momenta, can be rewritten as

$$v_{gq\bar{q}}^{2} = \frac{1}{\hat{p}_{\ell} \cdot \hat{p}_{j}} (d-4) + \frac{2}{\left(\hat{p}_{\ell} \cdot \hat{p}_{j}\right)^{2}} \left\{ \frac{\hat{p}_{\ell} \cdot Q \, \hat{p}_{j} \cdot p_{\ell} + \hat{p}_{j} \cdot Q \, \hat{p}_{\ell} \cdot p_{\ell}}{p_{\ell} \cdot Q} - \frac{Q^{2} \, \hat{p}_{\ell} \cdot p_{\ell} \, \hat{p}_{j} \cdot p_{\ell}}{\left(p_{\ell} \cdot Q\right)^{2}} \right\}$$
(2.125)

Averaging over the incoming parton spins gives an additional factor $1/2/(1 - \epsilon)$. For massless quarks, the $g\bar{q}q$ splitting function is the same.

The subprocess ggg splitting

For the ggg splitting, we have to do something slightly different because there are two identical gluons in the final state. We have to make sure that only the daughter parton j generates a singularity. The unaveraged splitting function is given by

$$v_{ggg}^{2} = \frac{1}{2(\hat{p}_{\ell} \cdot \hat{p}_{j})^{2}} \left\{ (d-2) \left[\hat{p}_{\ell} \cdot D_{j} \cdot \hat{p}_{\ell} + \hat{p}_{j} \cdot D_{\ell} \cdot \hat{p}_{j} \right] - k_{\perp}^{2} \operatorname{Tr} \left[D_{\ell} \cdot D_{j} \right] \right\}$$
(2.126)

⁸Here x_0 is a function of a_ℓ and y, *i.e.* $x_0 = x_0(a_\ell, y)$.

where $D_j = D(\hat{p}_j, \hat{Q}), D_\ell = D(\hat{p}_\ell, \hat{Q})$ and

$$\hat{p}_{\ell} \cdot D_{j} \cdot \hat{p}_{\ell} = \frac{2 y p_{\ell} \cdot Q}{x_{0} + z (1 - x_{0})} \left[1 - z (1 - x_{0}) - \frac{x_{0}}{x_{0} + z (1 - x_{0})} \right]$$

$$\hat{p}_{j} \cdot D_{\ell} \cdot \hat{p}_{j} = \frac{2 y p_{\ell} \cdot Q}{1 - z (1 - x_{0})} \left[x_{0} + z (1 - x_{0}) - \frac{x_{0}}{1 - z (1 - x_{0})} \right]$$

$$k_{\perp}^{2} = -2 y z (1 - z) p_{\ell} \cdot Q$$

$$\text{Tr} \left[D_{\ell} \cdot D_{j} \right] = d - 2 - 2 \Delta + \Delta^{2}$$

$$\Delta = \frac{Q^{2} \hat{p}_{\ell} \cdot \hat{p}_{j}}{\hat{p}_{\ell} \cdot Q \hat{p}_{j} \cdot Q} = \frac{2 x_{0}}{(x_{0} + z (1 - x_{0})) (1 - z (1 - x_{0}))}$$
(2.127)

Here, the additional statistical factor 1/2 in Eq. (2.126) appears only for a final state ggg splitting because the two gluons are identical. Instead of using Eq. (2.126) as a dipole subtraction term, however, we will use a slightly modified splitting function in order to well separate the singularities in the ggg splitting final state. Then we add a term $v'_{ggg}^2 = v_2^2 - v_3^2$ to Eq. (2.126), where v_2^2 and v_3^2 are defined in [29]. The additional term v'_{ggg}^2 with the roles of the two daughter gluons ℓ and j interchanged does not change the result. This way, there is a singularity when daughter gluon j becomes soft, but not when daughter gluon ℓ becomes soft. The additional term v'_{ggg}^2 is given by

$$v'_{ggg}^2 = v_2^2 - v_3^2 = \frac{d-2}{2\,(\hat{p}_\ell \cdot \hat{p}_j)^2} \,[\hat{p}_\ell \cdot D_j \cdot \hat{p}_\ell - \hat{p}_j \cdot D_\ell \cdot \hat{p}_j]$$
(2.128)

In the end, we obtain

$$\tilde{v}_{ggg}^{2} = v_{ggg}^{2} + v_{ggg}^{\prime 2} = \frac{1}{2(\hat{p}_{\ell} \cdot \hat{p}_{j})^{2}} \left\{ 2(d-2)\hat{p}_{\ell} \cdot D_{j} \cdot \hat{p}_{\ell} - k_{\perp}^{2} \operatorname{Tr} [D_{\ell} \cdot D_{j}] \right\} = \frac{2(1-\epsilon)}{(\hat{p}_{\ell} \cdot \hat{p}_{j})^{2}} \hat{p}_{\ell} \cdot D_{j} \cdot \hat{p}_{\ell} - \frac{k_{\perp}^{2}}{2(\hat{p}_{\ell} \cdot \hat{p}_{j})^{2}} \operatorname{Tr} [D_{\ell} \cdot D_{j}]$$
(2.129)

Hence, the dipole, including the eikonal splitting function, is given by

$$\tilde{v}_{ggg}^2 - v_{\text{eik}}^2 = -\frac{k_{\perp}^2}{2\,(\hat{p}_{\ell} \cdot \hat{p}_j)^2} \,\text{Tr}\left[D_{\ell} \cdot D_j\right] = \frac{z\,(1-z)}{y\,p_{\ell} \cdot \hat{Q}} \left[d - 2 - 2\,\Delta + \Delta^2\right] \tag{2.130}$$

which, in terms of the momenta, can be rewritten as

$$\tilde{v}_{ggg}^{2} - v_{\text{eik}}^{2} = \frac{\left(\hat{p}_{j} \cdot Q - a_{\ell} \, \hat{p}_{j} \cdot p_{\ell}\right) \left(\hat{p}_{\ell} \cdot Q - a_{\ell} \, \hat{p}_{\ell} \cdot p_{\ell}\right)}{\left(P_{\ell} \cdot n_{\ell}\right)^{2} \, \hat{p}_{\ell} \cdot \hat{p}_{j}} \left\{ 2 - 2 \frac{\hat{p}_{j} \cdot \hat{p}_{\ell} \, Q^{2}}{\hat{p}_{\ell} \cdot Q \, \hat{p}_{j} \cdot Q} + \left(\frac{\hat{p}_{j} \cdot \hat{p}_{\ell} \, Q^{2}}{\hat{p}_{\ell} \cdot Q \, \hat{p}_{j} \cdot Q}\right)^{2} \right\}$$
(2.131)

Averaging over the incoming parton spins gives an additional factor $1/2/(1-\epsilon)$.



Figure 2.9: Initial-state splitting processes.

2.4.4 Collinear splitting functions: initial state splittings

In this subsection, we will again ignore the common factor $4\pi\alpha_s$ and the colour factors for a moment. In (m+1)-parton phase spaces we always work in four dimensions. The eikonal factor will be considered if the emitted parton j is a gluon. For the initial state splittings, there is an additional scattering process in which the daughter quark comes in and splits a mother gluon that enters the hard interaction (see Fig. 2.9).

Kinematics: integration variables

For initial state splitting, we use the following integration variables:

$$x = \frac{\hat{p}_{a} \cdot p_{b} - \hat{p}_{a} \cdot \hat{p}_{j} - p_{b} \cdot \hat{p}_{j}}{\hat{p}_{a} \cdot p_{b}} = 1 - \frac{2 p_{A} \cdot \hat{p}_{j}}{\eta_{b} s} - \frac{2 p_{B} \cdot \hat{p}_{j}}{\hat{\eta}_{a} s} = \frac{\eta_{a}}{\hat{\eta}_{a}}$$

$$y = \frac{\hat{p}_{a} \cdot \hat{p}_{j}}{\hat{p}_{a} \cdot p_{b}} = \frac{\hat{\eta}_{a} p_{A} \cdot \hat{p}_{j}}{\frac{1}{2} \hat{\eta}_{a} \eta_{b} s} = \frac{2 p_{A} \cdot \hat{p}_{j}}{\eta_{b} s}$$
(2.132)

Turning the equations above around, we then obtain

$$p_A \cdot \hat{p}_j = \frac{y \eta_b s}{2}, \quad p_B \cdot \hat{p}_j = \frac{\eta_a s}{2 x} (1 - x - y), \quad \hat{p}_j \cdot \hat{Q} = \frac{\eta_a \eta_b s}{2 x} (1 - x)$$
(2.133)

and

$$\hat{\eta}_a = \frac{\eta_a}{x}, \qquad 0 < \hat{\eta}_a < 1 \qquad \Rightarrow \quad x > \eta_a$$

$$(2.134)$$

Here, $\hat{\eta}_a$ and η_a are momentum fractions after and before the splittings. As in the final state interference integrals, we also need to parametrize the additional angle which appears in interference terms. We define

$$v = \frac{\hat{\eta}_a \eta_b s \, \hat{p}_j \cdot \hat{p}_k}{2 \, \hat{p}_k \cdot \hat{Q} \, \hat{p}_j \cdot \hat{Q}} = \frac{1}{1 - x} \frac{\hat{p}_j \cdot \hat{p}_k}{\hat{p}_k \cdot \hat{Q}}$$
(2.135)

and

$$\bar{z} = \frac{\hat{p}_a \cdot \hat{p}_k}{\hat{p}_k \cdot \hat{Q}}, \qquad y' = \frac{y}{1-x}$$
(2.136)

As in the final state interference case, it is convenient to introduce the angles θ , θ_k and φ in the integration measure such that in the center of mass system we have

$$\hat{p}_{a} = \frac{\sqrt{\hat{s}}}{2} \begin{pmatrix} 1\\0\\0\\1 \end{pmatrix}, \qquad \hat{p}_{j} = A_{j} \begin{pmatrix} 1\\\sin\theta\cos\varphi\\\sin\theta\sin\varphi\\\cos\theta \end{pmatrix}, \qquad \hat{p}_{k} = A_{k} \begin{pmatrix} 1\\\sin\theta_{k}\\0\\\cos\theta_{k} \end{pmatrix}$$
(2.137)

In the integration measure, we will need an additional change of variable,

$$v' = \frac{1}{2} \left(1 - \cos \varphi \right) = \frac{v - v_{\min}}{v_{\max} - v_{\min}}$$
(2.138)

where

$$v = \frac{1}{2} (1 - \sin \theta \sin \theta_k \cos \varphi - \cos \theta \cos \theta_k)$$

$$v_{\text{max}} = \frac{1}{2} (1 + \sin \theta \sin \theta_k - \cos \theta \cos \theta_k)$$

$$v_{\text{min}} = \frac{1}{2} (1 - \sin \theta \sin \theta_k - \cos \theta \cos \theta_k)$$
(2.139)

We can also write \bar{z} and y' in terms of the angles θ_k and θ ,

$$\bar{z} = \frac{1}{2}(1 - \cos\theta_k), \qquad y' = \frac{1}{2}(1 - \cos\theta)$$
 (2.140)

The subprocess qq(x)g splitting

We assume that the gluon j is emitted from the parton a. For massless quarks, the unaveraged qq(x)g splitting function is given by

$$v_{qqg}^{2} = (d-2) \frac{2 p_{B} \cdot \hat{p}_{j}}{\eta_{a} \hat{\eta}_{a} s (p_{A} \cdot \hat{p}_{j})} + \frac{2 \eta_{b}^{2} p_{B} \cdot \hat{p}_{j} s}{(\hat{p}_{j} \cdot \hat{Q})^{2} p_{A} \cdot \hat{p}_{j}}$$
(2.141)

where $s = 2 p_A \cdot p_B$. Here, $\hat{\eta}_a$ and η_a are momentum fractions after and before the splittings. With our kinematics, the momentum fraction of parton *b* remains unchanged, *i.e.* $\hat{\eta}_b = \eta_b$. We denote the total momentum of the final state partons before the splitting by $Q = p_a + p_b$ and after the splitting by $\hat{Q} = \hat{p}_a + p_b$. In the splitting function, we choose the lightlike vector to be $n_a = p_B$ and it will be convenient to define $P_a = \hat{p}_a - \hat{p}_j$. Using the variables *x* and *y* (defined in previous subsection), Eq. (2.141) can be rewritten as

$$v_{qqg}^2 = \frac{2(1-x-y)}{\eta_a \eta_b y s} \left(d - 2 + \frac{4x}{(1-x)^2} \right)$$
(2.142)

We also define the eikonal approximation for soft gluon emission as discussed in Section 2.4.2,

$$F_{\text{eik}} = \frac{\hat{p}_a \cdot D(\hat{p}_j, \hat{Q}) \cdot \hat{p}_a}{\hat{p}_a \cdot \hat{p}_j}$$
(2.143)

or equivalently we can write the unaveraged eikonal splitting function as

$$v_{\text{eik}}^{2} = \frac{2\eta_{b}^{2} p_{B} \cdot \hat{p}_{j} s}{(\hat{p}_{j} \cdot \hat{Q})^{2} p_{A} \cdot \hat{p}_{j}} = \frac{8 x (1 - x - y)}{(1 - x)^{2} \eta_{a} \eta_{b} y s}$$
(2.144)

Using the variables x and y, the unaveraged splitting function including the eikonal factor is given by

$$v_{qqg}^{2} - v_{eik}^{2} = \frac{(d-2) 2 p_{B} \cdot \hat{p}_{j}}{\eta_{a} \hat{\eta}_{a} s (p_{A} \cdot \hat{p}_{j})} = \frac{(d-2) 2 p_{b} \cdot \hat{p}_{j}}{x \hat{Q}^{2} (\hat{p}_{a} \cdot \hat{p}_{j})} = \frac{(d-2) 2 (1-x-y)}{\eta_{a} \eta_{b} y s}$$
(2.145)

Averaging over the incoming parton spins gives an additional factor 1/2. For massless quarks, the $\bar{q}\bar{q}(x)g$ splitting function is the same.

The subprocess qqg(x) splitting

For the initial qqg(x) splitting process, in which the daughter quark with momentum \hat{p}_a comes in and splits a mother gluon with momentum p_a which enters the hard interaction, only the collinear

singularity is present. Hence, we do not subtract the eikonal splitting function in this case. The unaveraged qqg(x) splitting function for massless quarks is given by

$$v_{qqg}^2 = \frac{2(d-2)x}{\eta_a \eta_b y s} + \frac{8x(1-x-y)}{\eta_a \eta_b s y (x+y)^2}$$
(2.146)

which, in terms of the momenta, can be rewritten as

$$v_{qqg}^{2} = \frac{d-2}{\hat{\eta}_{a} p_{A} \cdot \hat{p}_{j}} + \frac{2}{\eta_{b} p_{B} \cdot P_{a}} \left\{ -1 + \frac{1}{\hat{\eta}_{a} p_{A} \cdot \hat{p}_{j}} \left(\hat{p}_{j} \cdot \hat{Q} + \eta_{b} p_{B} \cdot \hat{p}_{j} + \frac{2 \eta_{b} (p_{B} \cdot \hat{p}_{j})^{2}}{p_{B} \cdot P_{a}} \right) \right\}$$
$$= \frac{d-2}{\hat{p}_{a} \cdot \hat{p}_{j}} + \frac{2}{p_{b} \cdot P_{a}} \left\{ -1 + \frac{1}{\hat{p}_{a} \cdot \hat{p}_{j}} \left(\hat{p}_{j} \cdot \hat{Q} + p_{b} \cdot \hat{p}_{j} + \frac{2 (p_{b} \cdot \hat{p}_{j})^{2}}{p_{b} \cdot P_{a}} \right) \right\}$$
(2.147)

Averaging over the incoming parton spins gives an additional factor 1/2. For massless quarks, the $\bar{q}\bar{q}g(x)$ splitting function is the same.

The subprocess $gq\bar{q}$ splitting

Only the collinear singularity is present for the $gq\bar{q}$ splitting, so we do not subtract the eikonal splitting function in this case. The unaveraged splitting function for massless quarks is given by

$$v_{gq\bar{q}}^{2} = \frac{2(d-2)}{\eta_{a}\eta_{b}ys} + \frac{8(x+y)}{\eta_{a}\eta_{b}ys}(x+y-1)$$
(2.148)

which, in terms of the momenta, can be rewritten as

$$v_{gq\bar{q}}^{2} = \frac{d-2}{\eta_{a} p_{A} \cdot \hat{p}_{j}} + \frac{4}{\eta_{a} \eta_{b} s} \left\{ 1 - \frac{1}{\hat{\eta}_{a} p_{A} \cdot \hat{p}_{j}} \left(\hat{p}_{j} \cdot \hat{Q} + \eta_{b} p_{B} \cdot \hat{p}_{j} \right) + \frac{4 p_{B} \cdot \hat{p}_{j}}{\hat{\eta}_{a} s} \left(-1 + \frac{\hat{p}_{j} \cdot \hat{Q}}{\hat{\eta}_{a} p_{A} \cdot \hat{p}_{j}} \right) \right\}$$

$$= \frac{d-2}{x \hat{p}_{a} \cdot \hat{p}_{j}} + \frac{4}{x \hat{Q}^{2}} \left\{ 1 - \frac{1}{\hat{p}_{a} \cdot \hat{p}_{j}} \left(\hat{p}_{j} \cdot \hat{Q} + p_{b} \cdot \hat{p}_{j} \right) + \frac{4 p_{b} \cdot \hat{p}_{j}}{\hat{Q}^{2}} \left(-1 + \frac{\hat{p}_{j} \cdot \hat{Q}}{\hat{p}_{a} \cdot \hat{p}_{j}} \right) \right\}$$

$$(2.149)$$

Averaging over the incoming parton spins gives an additional factor $1/2/(1-\epsilon)$. For massless quarks, the $g\bar{q}q$ splitting function is the same.

The subprocess ggg splitting

The unaveraged ggg splitting function is given by

$$v_{ggg}^{2} = \frac{1}{(\hat{p}_{a} \cdot \hat{p}_{j})^{2}} \left\{ 2\left(1 - \epsilon\right) \frac{\hat{\eta}_{a}^{2} \eta_{b}^{2} s p_{A} \cdot \hat{p}_{j} p_{B} \cdot \hat{p}_{j}}{(\hat{p}_{j} \cdot \hat{Q})^{2}} + \frac{4 p_{A} \cdot \hat{p}_{j} p_{B} \cdot \hat{p}_{j}}{s} \left[2\left(1 - \epsilon\right) \left(1 + \frac{\hat{\eta}_{a}^{2} s^{2}}{4 \left(p_{B} \cdot P_{a}\right)^{2}}\right) - \frac{s \hat{\eta}_{a}^{2} p_{A} \cdot \hat{p}_{j}}{p_{B} \cdot P_{a} \hat{p}_{j} \cdot \hat{Q}} \right] \right\}$$
(2.150)

which, in terms of the variables x and y, can be rewritten as

$$v_{ggg}^{2} = \frac{4x(1-x-y)}{y\eta_{a}\eta_{b}s}(d-2)\left[1 + \frac{1}{(1-x)^{2}} + \frac{1}{(x+y)^{2}}\right] - \frac{8x(1-x-y)}{\eta_{a}\eta_{b}s(1-x)(x+y)}$$
(2.151)

If we include the eikonal factor, we obtain

$$v_{ggg}^2 - v_{eik}^2 = \frac{8x(1-x-y)}{\eta_a \eta_b y s} \left[(1-\epsilon) \left(1 + \frac{1}{(x+y)^2} \right) - \frac{\epsilon}{(1-x)^2} \right] - \frac{8x(1-x-y)}{\eta_a \eta_b s (1-x) (x+y)}$$
(2.152)

Averaging over the incoming parton spins gives an additional factor $1/2/(1 - \epsilon)$. Using the variables \hat{p}_a , \hat{p}_j , p_b and \hat{Q} , we then have

$$v_{ggg}^{2} - v_{eik}^{2} = \frac{4 p_{A} \cdot \hat{p}_{j} p_{B} \cdot \hat{p}_{j}}{s (\hat{p}_{a} \cdot \hat{p}_{j})^{2}} \left[2 \left(1 + \frac{\hat{\eta}_{a}^{2} s^{2}}{4 (p_{B} \cdot P_{a})^{2}} \right) - \frac{s \hat{\eta}_{a}^{2} p_{A} \cdot \hat{p}_{j}}{p_{B} \cdot P_{a} \hat{p}_{j} \cdot \hat{Q}} \right]$$
$$= \frac{4 \hat{p}_{a} \cdot \hat{p}_{j} p_{b} \cdot \hat{p}_{j}}{\hat{Q}^{2} (\hat{p}_{a} \cdot \hat{p}_{j})^{2}} \left[2 \left(1 + \frac{\hat{Q}^{4}}{4 (p_{b} \cdot P_{a})^{2}} \right) - \frac{\hat{Q}^{2} \hat{p}_{a} \cdot \hat{p}_{j}}{p_{b} \cdot P_{a} \hat{p}_{j} \cdot \hat{Q}} \right]$$
(2.153)

This is the dipole subtraction term for (m + 1)-parton phase spaces, so we can safely put d = 4.

2.4.5 Soft splitting functions

We have discussed the spin-averaged splitting functions $\overline{W}_{\ell\ell}$ in which the parton j is emitted from the emitter ℓ in the scattering amplitude and parton j is emitted from that same emitter ℓ in the complex-conjugate scattering amplitude. In higher-order QCD calculations, double poles in splitting functions only arise if the emitted parton j is a gluon. In this case, interference diagrams between different emitters have to be taken into account. This means that the emitted parton jcan be emitted from emitter ℓ in the amplitude and parton j can also be emitted from different emitter k in the complex-conjugate amplitude (Fig. 2.8). The interference splitting function is then

$$v_{\ell}(\{\hat{p}, \hat{f}\}_{m+1}, \hat{s}_j, \hat{s}_{\ell}, s_{\ell}) v_k(\{\hat{p}, \hat{f}\}_{m+1}, \hat{s}_j, \hat{s}_k, s_k)^* \delta_{\hat{s}_{\ell}, s_{\ell}} \delta_{\hat{s}_k, s_k}$$
(2.154)

The splitting function Eq. (2.154) contains a singularity when the emitted gluon j is soft. However when gluon j is collinear with parton ℓ or parton k, it does not contribute a leading singularity. In the special case that \hat{p}_j is soft, or possibly soft and collinear with \hat{p}_{ℓ} , a simpler splitting amplitude can be used. When \hat{p}_j is soft, we have

$$|\mathcal{M}(\{\hat{p}, \hat{f}\}_{m+1}) > \sim \sum_{\ell} |\mathcal{M}_{\ell}^{\text{soft}}(\{\hat{p}, \hat{f}\}_{m+1}) >$$
(2.155)

where

$$|\mathcal{M}_{\ell}^{\text{soft}}(\{\hat{p},\hat{f}\}_{m+1})\rangle = t_{\ell}^{\dagger}(f_{\ell} \to \hat{f}_{\ell} + \hat{f}_{j}) V_{\ell}^{\dagger \text{ soft}}(\{\hat{p},\hat{f}\}_{m+1}) |\mathcal{M}(\{p,f\}_{m})\rangle$$
(2.156)

The spin dependent splitting operator can be described in the spin space $|\{s\}_m >$,

$$<\{\hat{s}\}_{m+1} \mid V_{\ell}^{\dagger \operatorname{soft}}(\{\hat{p}, \hat{f}\}_{m+1}) \mid \{s\}_{m} > = \left(\prod_{n \notin \{\ell, j=m+1\}} \delta_{\hat{s}_{n}, s_{n}}\right) v_{\ell}^{\operatorname{soft}}(\{\hat{p}, \hat{f}\}_{m+1}, \hat{s}_{j}, \hat{s}_{\ell}, s_{\ell})$$

$$(2.157)$$

If parton j is a quark or antiquark, $v_{\ell}^{\text{soft}} = 0$. When parton j is a gluon, we can use a simple eikonal approximation to the splitting amplitude,

$$v_{\ell}^{\text{soft}}(\{\hat{p}, \hat{f}\}_{m+1}, \hat{s}_j, \hat{s}_{\ell}, s_{\ell}) = \sqrt{4\pi\alpha_s} \,\delta_{\hat{s}_{\ell}, s_{\ell}} \,\frac{\varepsilon(\hat{p}_j, \hat{s}_j, Q)^* \cdot \hat{p}_{\ell}}{\hat{p}_j \cdot \hat{p}_{\ell}}$$
(2.158)

Having used the eikonal approximation, the interference splitting function becomes

$$\overline{W}_{\ell k} \sim v_{\ell}^{\text{soft}}(\{\hat{p}, \hat{f}\}_{m+1}, \hat{s}_j, \hat{s}_\ell, s_\ell) v_k^{\text{soft}}(\{\hat{p}, \hat{f}\}_{m+1}, \hat{s}_j, \hat{s}_k, s_k)^* \,\delta_{\hat{s}_\ell, s_\ell} \,\delta_{\hat{s}_k, s_k} \tag{2.159}$$

In the Nagy-Soper scheme, we split the collinear and soft parts of the respective spin-averaged splitting functions according to

$$\overline{W}_{\ell\ell} - \overline{W}_{\ell k} = \left(\overline{W}_{\ell\ell} - \overline{W}_{\ell\ell}^{\text{eikonal}}\right) + \left(\overline{W}_{\ell\ell}^{\text{eikonal}} - \overline{W}_{\ell k}\right)$$
(2.160)

where the spin-averaged soft splitting function $\overline{W}_{\ell k}$ is given by

$$\overline{W}_{\ell k} = 4 \pi \alpha_s 2 A_{\ell k} \frac{\hat{p}_{\ell} \cdot D(\hat{p}_j, Q) \cdot \hat{p}_k}{\hat{p}_j \cdot \hat{p}_\ell \, \hat{p}_j \cdot \hat{p}_k}$$
(2.161)

Here, $A_{\ell k}$ is the partitioning weight function, which in principle can depend on the momenta $\{\hat{p}\}_{m+1}$. It specifies how the two interference diagrams in Fig. 2.8 are partitioned into separate terms. In [28], the default value is taken to be $A_{\ell k} = A_{k\ell} = 1/2$. We can also define the partitioning function as function of momenta $A_{\ell k}(\{\hat{p}\}_{m+1})$ and $A_{k\ell}(\{\hat{p}\}_{m+1})$, so another choice would be

$$A_{\ell k}(\{\hat{p}\}_{m+1}) = \frac{B_{\ell k}(\{\hat{p}\}_{m+1})}{B_{\ell k}(\{\hat{p}\}_{m+1}) + B_{k\ell}(\{\hat{p}\}_{m+1})}$$
(2.162)

where

$$B_{\ell k}(\{\hat{p}\}_{m+1}) = \frac{\hat{p}_j \cdot \hat{p}_k}{\hat{p}_j \cdot \hat{p}_\ell} \hat{p}_\ell \cdot D(\hat{p}_j, \hat{Q}) \cdot \hat{p}_\ell$$
(2.163)

Here, $D(\hat{p}_j, \hat{Q})$ is defined in Eq. (2.99). The partitioning functions are non-negative and obey $A_{\ell k}(\{\hat{p}\}_{m+1}) + A_{k\ell}(\{\hat{p}\}_{m+1}) = 1.$

One of the advantages from dividing the splitting functions into two parts is that the two terms $\overline{W}_{\ell\ell} - \overline{W}_{\ell\ell}^{\text{eikonal}}$ and $\overline{W}_{\ell\ell}^{\text{eikonal}} - \overline{W}_{\ell k}$ are positive, and thus we can use these splitting functions as dipoles to construct a parton shower Monte Carlo program without needing separate partitioning

weight function. Another result from splitting Eq. (2.160) into two pieces is that the first term of Eq. (2.160) only contains a collinear singularity, while the second term of Eq. (2.160) has both the soft singularity and the soft \otimes collinear singularity.

The second part of Eq. (2.160) can be then expressed in terms of dipole partitioning functions $A'_{\ell k}$, which is given in [30]:

$$\overline{W}_{\ell\ell}^{\text{eikonal}} - \overline{W}_{\ell k} = 4 \pi \alpha_s A_{\ell k}' \frac{-\hat{P}_{\ell k}^2}{(\hat{p}_j \cdot \hat{p}_\ell \, \hat{p}_j \cdot \hat{p}_k)^2}$$
(2.164)

where $\hat{P}_{\ell k} = \hat{p}_j \cdot \hat{p}_\ell \hat{p}_k - \hat{p}_j \cdot \hat{p}_k \hat{p}_\ell$. Several choices for $A'_{\ell k}$ have been proposed in [30]. All results given here have been obtained using Eq. (7.12) therein, which is given by

$$A'_{\ell k}(\{\hat{p}\}_{m+1}) = \frac{\hat{p}_j \cdot \hat{p}_k \, \hat{p}_\ell \cdot \hat{Q}}{\hat{p}_j \cdot \hat{p}_k \, \hat{p}_\ell \cdot \hat{Q} + \hat{p}_j \cdot \hat{p}_\ell \, \hat{p}_k \cdot \hat{Q}}$$
(2.165)

The partitioning weight function $A'_{\ell k}$ also obeys the relation $A'_{\ell k}(\{\hat{p}\}_{m+1}) + A'_{k\ell}(\{\hat{p}\}_{m+1}) = 1$. The general form of the interference spin-averaged splitting function is then given by

$$\Delta W = \overline{W}_{\ell\ell}^{\text{eikonal}} - \overline{W}_{\ell k} = \frac{2\,\hat{p}_{\ell}\cdot\hat{p}_{k}\,\hat{p}_{\ell}\cdot\hat{Q}}{\hat{p}_{\ell}\cdot\hat{p}_{j}\,\left(\hat{p}_{j}\cdot\hat{p}_{k}\,\hat{p}_{\ell}\cdot\hat{Q}+\hat{p}_{\ell}\cdot\hat{p}_{j}\,\hat{p}_{k}\cdot\hat{Q}\right)}$$
(2.166)

Here, we have removed the common factor $4 \pi \alpha_s$, which we will add back in the end. The only singularity in Eq. (2.166) arises from the factor $\hat{p}_{\ell} \cdot \hat{p}_j$ in the denominator. The interference term is constructed in such a way that it vanishes for $\hat{p}_j \cdot \hat{p}_k \to 0$. We also assume that the variables considered are such that they are finite for $p_{\ell} \cdot p_k \to 0$, *i.e.* singularities arising in this limit should be taken care of by the definition of the jet function. The interference term only needs to be considered if the emitted parton j is a gluon. If parton j is a quark or antiquark, this term vanishes.

Eq. (2.166) corresponds to interference between final states. It is worth mentioning that if we replace \hat{p}_{ℓ} by \hat{p}_a in Eq. (2.166), then the interference spin-averaged splitting function corresponds to interference between initial and final states,

$$\Delta W_{ak} = \frac{2p_A \cdot \hat{p}_k \eta_b s}{p_A \cdot \hat{p}_j \left(\hat{p}_j \cdot \hat{p}_k \eta_b s + 2p_A \cdot \hat{p}_j \hat{p}_k \cdot \hat{Q} \right)} \\ = \frac{2 \hat{p}_a \cdot \hat{p}_k \hat{Q}^2}{\hat{p}_a \cdot \hat{p}_j \left(\hat{p}_j \cdot \hat{p}_k \hat{Q}^2 + 2 \hat{p}_a \cdot \hat{p}_j \hat{p}_k \cdot \hat{Q} \right)} \\ = \frac{4 \bar{z} x}{\eta_a \eta_b s y' (1 - x)^2 (v + y')}$$
(2.167)

where the variables \bar{z} , y', x and v are defined in Section 2.4.4. The Eq. (2.167) can be further reduced by the following replacement

$$\hat{p}_k \rightarrow p_b \quad \text{or} \quad \bar{z} = \frac{\hat{p}_a \cdot \hat{p}_k}{\hat{p}_k \cdot \hat{Q}} \rightarrow 1$$
 (2.168)

in which case the soft splitting function Eq. (2.166) corresponds to interference between initial states:

$$\Delta W_{ab} = \frac{\eta_b s}{p_A \cdot \hat{p}_j \, \hat{p}_j \cdot \hat{Q}} = \frac{\hat{Q}^2}{\hat{p}_a \cdot \hat{p}_j \, \hat{p}_j \cdot \hat{Q}} = \frac{4 \, x}{(1 - x) \, \eta_a \, \eta_b \, y \, s}$$
(2.169)

Again, if we replace \hat{p}_k by \hat{p}_a in Eq. (2.166), then the interference splitting function corresponds to interference between final and initial states (see *e.g.* Eq. (2.204) or Eq. (2.205)).

2.5 Integrated splitting functions

In this section we will list all the integrated splitting functions, which are needed for the *m*-parton phase spaces. The integrated splitting functions contain all the singularities in ϵ that are necessary to cancel the poles in one-loop virtual matrix element. In order to achieve the cancellation of singularities we have to define good parametrizations in the integration measures which we will discuss in Section 5.2. We will consider the collinear integrals in both the initial state and final state splittings as well as the interference terms.

For the initial and final state collinear integrals we have been using the colour algebra relations defined by

$$\mathbf{T}_{\ell}^2 \equiv C_i \tag{2.170}$$

where $C_i = C_A(C_F)$ in the ggg(qqg) splitting. For the splitting process $gq\bar{q}$ we use the colour charge T_R . For the soft splitting functions the appropriate colour charges are given by

$$\left(-\sum_{k\neq\ell} \mathbf{T}_{\ell} \cdot \mathbf{T}_{k}\right) \equiv C_{i} \tag{2.171}$$

We will discuss this issue in more detail in Section 5.3. Note that parton ℓ is the emitter and parton k is the spectator.

2.5.1 Collinear integrals: final state splittings

The subprocess qqg

After subtracting the eikonal factor, which will be combined with the interference term, the collinear part of the integrated splitting function is given by

$$\frac{4\pi\alpha_s}{2} C_F \mu^{2\epsilon} \int d\zeta_p \left[v_{qqg}^2 - v_{eik}^2 \right] = \frac{\alpha_s}{4\pi} C_F \left(\frac{4\pi\mu^2}{2p_\ell \cdot Q} \right)^\epsilon \frac{1}{\Gamma(1-\epsilon)} \times \left\{ -\frac{1}{\epsilon} - 2 \left[2a_\ell + 1 + 2(a_\ell - 1)^2 \ln(a_\ell - 1) + 2a_\ell (2 - a_\ell) \ln a_\ell - 4(a_\ell - 1) y_{max} + y_{max}^2 + 2\ln y_{max} \right] + \frac{1}{2} \left[2(1 - 2a_\ell) y_{max} + \frac{1}{2} y_{max}^2 + \ln y_{max} - \frac{7}{2} + a_\ell + (a_\ell^2 - 1) \ln(a_\ell - 1) - a_\ell^2 \ln a_\ell \right] + 4 I_{fin}(a_\ell) \right\}$$
(2.172)

where

$$I_{\rm fin}(a_\ell) = -\int_0^{y_{\rm max}} dy \, \frac{(\lambda - 1 + y)^2 + 4y}{4y} \, \frac{(1 + x_0) \ln x_0}{1 - x_0}$$
(2.173)

For $a_{\ell} = 1$, this simplifies to

$$\frac{4\pi\alpha_s}{2}C_F\,\mu^{2\epsilon}\int d\zeta_p\left[v_{qqg}^2 - v_{\text{eik}}^2\right] = \frac{\alpha_s}{4\,\pi}C_F\,\left(\frac{4\pi\mu^2}{Q^2}\right)^\epsilon\frac{1}{\Gamma(1-\epsilon)}\left(-\frac{1}{\epsilon} - 14 + \frac{4\pi^2}{3} + \mathcal{O}(\epsilon)\right) \tag{2.174}$$

which has been used in dijet production at NLO.

The subprocess $gqar{q}$

Including all prefactors, we obtain

$$\frac{4\pi\alpha_s}{2(1-\epsilon)} T_R \mu^{2\epsilon} \int d\zeta_p \, v_{gq\bar{q}}^2 = \frac{\alpha_s}{\pi} T_R \left(\frac{4\pi\,\mu^2}{2\,p_\ell \cdot Q}\right)^\epsilon \, \frac{1}{\Gamma(1-\epsilon)} \left[-\frac{1}{3\,\epsilon} - \frac{8}{9} + \frac{1}{3}\left[(a_\ell - 1)\,\ln(a_\ell - 1) - a_\ell\,\ln\,a_\ell\right]\right] \quad (2.175)$$

For $a_{\ell} = 1$, we exactly reproduce the result in [22]. Note that the first two terms in Eq. (2.175) are exactly the same as in [22]; differences in the finite terms stem from the difference in the (m + 1)- to m-parton momentum matching.

The subprocess ggg

Including all prefactors, we obtain

$$\frac{4\pi\,\alpha_s}{2\,(1-\epsilon)}\,C_A\,\mu^{2\epsilon}\,\int\,d\zeta_p\,\left[\tilde{v}_{ggg}^2-v_{\rm eik}^2\right]\,=\,\frac{\alpha_s}{2\,\pi}\,C_A\,\left(\frac{4\,\pi\,\mu^2}{2\,p_\ell\cdot Q}\right)^\epsilon\,\frac{1}{\Gamma(1-\epsilon)}\\\times\left[-\frac{1}{6\,\epsilon}\,-\,\frac{4}{9}\,+\,\frac{1}{6}\,\left[\,(a_\ell-1)\,\ln(a_\ell-1)\,-\,a_\ell\ln\,a_\ell\,\right]\,+\,I_{\rm fin}(a_\ell)\right]\tag{2.176}$$

where

$$I_{\text{fin}}(a_{\ell}) = -2 a_{\ell} \int_{0}^{y_{\text{max}}} \frac{dy}{\lambda (1+y)^{2}} \left\{ \left[(1+y)^{2} + 2 a_{\ell} y \right] + 4 a_{\ell} \frac{y \ln x_{0}}{\lambda (1+y)} \left[(1+y)^{2} - a_{\ell} y \right] \right\}$$

$$\stackrel{a_{\ell} \neq 1}{=} a_{\ell} \left\{ 1 - \sqrt{a_{\ell}} \ln \left(\frac{\sqrt{a_{\ell}} + 1}{\sqrt{a_{\ell}} - 1} \right) - \ln \left(\frac{a_{\ell}}{a_{\ell} - 1} \right) + 8 a_{\ell} \int_{0}^{y_{\text{max}}} dy \frac{y \ln x_{0}}{\lambda^{2} (1+y)^{3}} \left[a_{\ell} y - (1+y)^{2} \right] \right\}$$

$$\stackrel{a_{\ell} = 1}{=} -\frac{3}{8} \pi^{2} + \frac{7}{2}$$
(2.177)

2.5.2 Collinear integrals: initial state splittings

The subprocess qq(x)g

Using the variables defined in Section 2.4.4, the unaveraged qq(x)g splitting function is given by

$$v_{qqg}^2 = \frac{2\left(1 - x - y\right)}{\eta_a \eta_b y s} \left(d - 2 + \frac{4x}{(1 - x)^2}\right)$$
(2.178)

Including all prefactors, we obtain

$$\frac{4\pi\alpha_s}{2}C_F \mu^{2\epsilon} \int d\zeta_p v_{qqg}^2$$

$$= \frac{\alpha_s}{2\pi}C_F \left(\frac{4\pi\mu^2}{\eta_a \eta_b s}\right)^{\epsilon} \frac{1}{\Gamma(1-\epsilon)} \int_0^1 \frac{dx}{x} \left[\delta(1-x)\left(\frac{1}{\epsilon^2} + \frac{5}{2\epsilon}\right) - \frac{1}{\epsilon}\left(\frac{1+x^2}{1-x}\right)_+ + \mathfrak{B}(x,\epsilon^0)\right]$$
(2.179)

where

$$\mathfrak{B}(x,\epsilon^{0}) = -(1-x)\ln x + 2(1-x)\ln(1-x) + 4x\left(\frac{\ln(1-x)}{1-x}\right)_{+} \\ - \frac{2x(1+\ln x)}{(1-x)_{+}} + 2\delta(1-x)\left(1-\frac{\pi^{2}}{12}\right)$$
(2.180)

The leftover factor 1/x will be reabsorbed into the flux factor. Note that we should take the eikonal splitting function into account, when we consider the interference term. The integrated eikonal splitting function will be discussed later in more detail.

The subprocess qqg(x)

Using the variables defined in Section 2.4.4, the unaveraged qqg(x) splitting function is given by

$$v_{qqg}^{2} = \frac{2(d-2)x}{\eta_{a}\eta_{b}ys} + \frac{8x(1-x-y)}{\eta_{a}\eta_{b}sy(x+y)^{2}}$$
(2.181)

Including all prefactors, we obtain

$$\frac{4\pi\alpha_s}{2}C_F \mu^{2\epsilon} \int d\zeta_p v_{qqg}^2$$

$$= \frac{\alpha_s}{2\pi}C_F \left(\frac{4\pi\mu^2}{\eta_a \eta_b s}\right)^{\epsilon} \frac{1}{\Gamma(1-\epsilon)} \int_0^1 \frac{dx}{x} \left[-\frac{1}{\epsilon} \left(\frac{1+(1-x)^2}{x}\right) + \mathfrak{B}(x,\epsilon^0)\right] \quad (2.182)$$

where

$$\mathfrak{B}(x,\epsilon^0) = \frac{x^2 - 2(1-x)}{x} - x \ln x + 2 \ln(1-x) \left(\frac{1 + (1-x)^2}{x}\right) \qquad (2.183)$$

The leftover factor 1/x will be reabsorbed into the flux factor.

The subprocess $gqar{q}$

The unaveraged $gq\bar{q}$ splitting function is given by

$$v_{gq\bar{q}}^{2} = \frac{2(d-2)}{\eta_{a}\eta_{b}ys} + \frac{8(x+y)}{\eta_{a}\eta_{b}ys}(x+y-1)$$
(2.184)

Including all prefactors, the integrated splitting function is given by

$$\frac{4\pi\alpha_s}{2(1-\epsilon)}T_R\mu^{2\epsilon}\int d\zeta_p v_{gq\bar{q}}^2$$

$$=\frac{\alpha_s}{2\pi}T_R\left(\frac{4\pi\mu^2}{\eta_a\eta_b s}\right)^{\epsilon}\frac{1}{\Gamma(1-\epsilon)}\int_0^1\frac{dx}{x}\left[-\frac{1}{\epsilon}\left[x^2+(1-x)^2\right]+\mathfrak{B}(x,\epsilon^0)\right]$$
(2.185)

where

$$\mathfrak{B}(x,\epsilon^0) = 6x - 5x^2 - 1 + [2\ln(1-x) - \ln x] [x^2 + (1-x)^2] \qquad (2.186)$$

The leftover factor 1/x will be reabsorbed into the flux factor.

The subprocess ggg

The unaveraged ggg splitting function is given by

$$v_{ggg}^{2} = \frac{4x(1-x-y)}{y\eta_{a}\eta_{b}s}(d-2)\left[1 + \frac{1}{(1-x)^{2}} + \frac{1}{(x+y)^{2}}\right] - \frac{8x(1-x-y)}{\eta_{a}\eta_{b}s(1-x)(x+y)}$$
(2.187)

Including all prefactors, the integrated splitting function is given by

$$\frac{4\pi\alpha_s}{2(1-\epsilon)}C_A\mu^{2\epsilon}\int d\zeta_p \, v_{ggg}^2 = \frac{\alpha_s}{2\pi}C_A \left(\frac{4\pi\mu^2}{\eta_a \eta_b s}\right)^\epsilon \frac{1}{\Gamma(1-\epsilon)}$$

$$\times \int_0^1 \frac{dx}{x} \left[\frac{1}{\epsilon^2}\delta(1-x) + \frac{1}{\epsilon}\left[\delta(1-x) - 2\left(\frac{x}{(1-x)_+} + x(1-x) + \frac{1-x}{x}\right)\right] + \mathfrak{B}(x,\epsilon^0)\right]$$
(2.188)

where

$$\mathfrak{B}(x,\epsilon^{0}) = 4x \left(\frac{\ln(1-x)}{1-x}\right)_{+} - 2x(1-x)\ln x + 4(1-x)\ln(1-x)\left(\frac{1+x^{2}}{x}\right) + 2\left(x^{2} - \frac{1-x}{x} - \frac{x}{(1-x)_{+}}\right) + 2\left(1 - \frac{\pi^{2}}{12}\right)\delta(1-x)$$
(2.189)

The leftover factor 1/x will be reabsorbed into the flux factor.

The subprocess ggg with eikonal splitting function

Now we consider the eikonal factor. Using the variables defined in Section 2.4.4, the averaged eikonal splitting function is given by (see also Section 2.4.2)

$$\bar{v}_{\text{eik}}^2 = \frac{4 x (1 - x - y)}{(1 - x)^2 \eta_a \eta_b y s}$$
(2.190)

Integrating it out, we obtain

$$4 \pi \alpha_s C_i \mu^{2\epsilon} \int d\zeta_p \, \bar{v}_{eik}^2 = \frac{\alpha_s}{2 \pi} C_i \left(\frac{4 \pi \mu^2}{\eta_a \eta_b s} \right)^{\epsilon} \frac{1}{\Gamma(1-\epsilon)} \\ \times \int_0^1 dx \left[\frac{1}{\epsilon^2} \,\delta(1-x) + \frac{1}{\epsilon} \left(\delta(1-x) - \frac{2}{(1-x)_+} \right) + 2 \left(1 - \frac{\pi^2}{12} \right) \delta(1-x) \\ -2 \frac{1+\ln x}{(1-x)_+} + 4 \left(\frac{\ln(1-x)}{1-x} \right)_+ \right]$$
(2.191)

Here, C_i is given by Eq. (2.170). Hence the collinear ggg splitting including eikonal factor is given by

$$\frac{4\pi\alpha_s}{2(1-\epsilon)} C_A \mu^{2\epsilon} \int d\zeta_p \left[v_{ggg}^2 - v_{eik}^2 \right] = \frac{\alpha_s}{2\pi} C_A \left(\frac{4\pi\mu^2}{\eta_a \eta_b s} \right)^{\epsilon} \frac{1}{\Gamma(1-\epsilon)} \\ \times \int_0^1 \frac{dx}{x} \left[-\frac{2}{\epsilon} \left(\frac{x}{(1-x)_+} + x(1-x) + \frac{1-x}{x} \right) + \frac{1}{\epsilon} \frac{2x}{(1-x)_+} + \mathfrak{A}(x,\epsilon^0) \right]$$
(2.192)

where

$$\mathfrak{A}(x,\epsilon^{0}) = -2x(1-x)\ln x + 4(1-x)\ln(1-x)\left(\frac{1+x^{2}}{x}\right) + 2\left(x^{2} - \frac{1-x}{x}\right) + 2x\frac{\ln x}{(1-x)_{+}}$$
(2.193)

The complete integrated ggg splitting function including collinear and interference terms (see Section 2.5.3) is then given by

$$\frac{4\pi\alpha_s}{2(1-\epsilon)}C_A\mu^{2\epsilon}\int d\zeta_p \left[v_{ggg}^2 - v_{eik}^2\right] + 4\pi\alpha_s C_A\mu^{2\epsilon}\int d\zeta_p \ \Delta W_{ab} = \frac{\alpha_s}{2\pi}C_A \left(\frac{4\pi\mu^2}{\eta_a \eta_b s}\right)^{\epsilon} \frac{1}{\Gamma(1-\epsilon)} \times \int_0^1 \frac{dx}{x} \left[\left(\frac{1}{\epsilon^2} - \frac{\pi^2}{6}\right)\delta(1-x) - \frac{2}{\epsilon}\left(\frac{x}{(1-x)_+} + x(1-x) + \frac{1-x}{x}\right) + \mathfrak{J}(x,\epsilon^0)\right]$$
(2.194)

where

$$\mathfrak{J}(x,\epsilon^{0}) = 4x \left(\frac{\ln(1-x)}{1-x}\right)_{+} - 2x(1-x)\ln x + 4(1-x)\ln(1-x)\left(\frac{1+x^{2}}{x}\right) + 2\left(x^{2} - \frac{1-x}{x}\right)$$
(2.195)

2.5.3 Interference between initial states

After adding back the eikonal factor, which we have subtracted in the collinear integrals, the interference part of the integrated splitting function (see Eq. (2.169)) including all prefactors is given by

$$4\pi \alpha_s C_i \mu^{2\epsilon} \int d\zeta_p \,\Delta W_{ab} = \frac{\alpha_s}{2\pi} C_i \left(\frac{4\pi \mu^2}{\eta_a \eta_b s}\right)^{\epsilon} \frac{1}{\Gamma(1-\epsilon)} \times \int_0^1 dx \left\{ \frac{1}{\epsilon^2} \delta(1-x) - \frac{2}{\epsilon} \frac{1}{(1-x)_+} - \frac{\pi^2}{6} \delta(1-x) - 2\frac{\ln x}{(1-x)_+} + 4\left(\frac{\ln(1-x)}{1-x}\right)_+ \right\}$$
(2.196)

where C_i is given by Eq. (2.171). Note that the individual part of the unaveraged soft splitting function is given by

$$v_{\text{soft}}^2 = -16 \frac{x (1 - x - y)^2}{\eta_a \eta_b s (1 - x)^2 (y^2 + (1 - x - y)^2)}$$
(2.197)

Integrating it out, we obtain

$$\frac{4\pi\alpha_s}{2}C_i\mu^{2\epsilon}\int d\zeta_p v_{\text{soft}}^2$$

$$=\frac{\alpha_s}{2\pi}C_i\left(\frac{4\pi\mu^2}{\eta_a\eta_b s}\right)^{\epsilon}\frac{1}{\Gamma(1-\epsilon)}\int_0^1 dx \left[\frac{1}{\epsilon}\delta(1-x)+2\delta(1-x)-\frac{2}{(1-x)_+}\right] \qquad (2.198)$$

2.5.4 Interference between initial and final states

The initial-final state interference splitting function is given by Eq. (2.167), which is derived from Eq. (2.166) by the replacement $\hat{p}_{\ell} \rightarrow \hat{p}_a$. Using the initial state integration measure given by Eq. (5.43), we obtain

$$4\pi \alpha_{s} C_{i} \mu^{2\epsilon} \int d\zeta_{p} \Delta W_{ak} = \frac{\alpha_{s}}{2\pi} C_{i} \left(\frac{4\pi\mu^{2}}{\eta_{a} \eta_{b} s}\right)^{\epsilon} \frac{1}{\Gamma(1-\epsilon)} \times \int_{0}^{1} dx \left\{\frac{1}{\epsilon^{2}} \delta(1-x) - \frac{1}{\epsilon} \left[\frac{2}{(1-x)_{+}} + \delta(1-x) \ln z_{0}\right] + 4\left(\frac{\ln(1-x)}{1-x}\right)_{+} - \frac{2\ln x}{(1-x)_{+}} + \ln(16) \ln z_{0} \delta(1-x) + \frac{2}{\pi} I_{\text{fin}}^{a}(x,\bar{z})\right\}$$

$$(2.199)$$

where C_i is given by Eq. (2.171). In Eq. (2.167), $\overline{z} = 0$ corresponds to a singularity in the *m*particle phase space. This singularity should be excluded by an appropriate jet function definition since we only consider infrared safe observables. The function $\overline{z} = \overline{z}(x, y')$ in Eq. (2.167) contains Lorentz-transformed variables. Only in the limit of $x \to 1$ (soft limit) or $y' \to 0$ (soft or collinear limit) this dependence disappears; in this case the \overline{z} is denoted by z_0 . Hence we have

$$\bar{z} = \bar{z}(x, y'), \quad z_0 = \bar{z}(1, y') = \bar{z}(x, 0), \quad 0 < z_0 < 1$$
(2.200)

The finite term I_{fin}^a is given by

$$I_{\rm fin}^{a} = I_{\rm fin}^{a}(x,\bar{z}) = \pi \,\delta(1-x) \left\{ -\ln 4 \ln z_{0} - \frac{1}{8} \left[2 \,\mathrm{Li}_{2} \left(\frac{z_{0}-1}{z_{0}} \right) - \ln^{2} z_{0} \right] \right. \\ \left. + \int_{0}^{1} dy \, \frac{z_{0}}{y\sqrt{4 \, y^{2} \, (1-z_{0}) + z_{0}^{2}}} \ln \frac{2\sqrt{4 \, y^{2} \, (1-z_{0}) + z_{0}^{2}} \sqrt{1-y}}{2 \, y \, (1-z_{0}) + z_{0} + \sqrt{4 \, y^{2} \, (1-z_{0}) + z_{0}^{2}}} \right\} \\ \left. + \frac{1}{(1-x)_{+}} \int_{0}^{1} \frac{dy}{y} \left\{ \left[\int_{0}^{1} \frac{dv}{\sqrt{v \, (1-v)}} \frac{\bar{z}(x,y)}{N(x,y,v,\bar{z})} \right] - \pi \right\}$$
(2.201)

with

$$N(x, y, v, \bar{z}) = (4v - 2)\sqrt{y(1 - y)\bar{z}(1 - \bar{z})} + 2y + \bar{z} - 2y\bar{z}$$
(2.202)

Here, v and y are dummy variables, which we denoted with v' and y' previously. So the function \overline{z} in Eq. (2.201) is now a function of x and y. Note that we can only use Eq. (2.200) in the

singular limits $(x \to 1 \text{ or } y' \to 0)$ where the dependence on v' disappears. For the finite parts, *i.e.* Eq. (2.201), we need to use the original definition of \bar{z} given by Eq. (2.136)

$$\bar{z} = \frac{\hat{p}_a \cdot \hat{p}_k}{\hat{p}_k \cdot \hat{Q}}$$

Additionally we use the original definition of v given by Eq. (2.135)

$$v = \frac{\hat{\eta}_a \eta_b s \hat{p}_j \cdot \hat{p}_k}{2 \hat{p}_k \cdot \hat{Q} \hat{p}_j \cdot \hat{Q}} = \frac{1}{1 - x} \frac{\hat{p}_j \cdot \hat{p}_k}{\hat{p}_k \cdot \hat{Q}}$$

where \hat{p}_k needs to be calculated using the Lorentz transformation: $\hat{p}_k^{\mu} = \Lambda(\hat{K}, K)^{\mu}{}_{\nu} p_k^{\nu}$ and \hat{p}_j is parametrized according to Sudhakov parametrization⁹. In the singular limits, we obtain $\hat{p}_k \rightarrow p_k$. For $\bar{z} \rightarrow 1$ (which means $\hat{p}_k \rightarrow p_b$), we get

$$I_{\rm fin}^a = -\frac{\pi^3}{12}\delta(1-x)$$
(2.203)

in which case the initial-final state integrated splitting function is reduced to initial-initial state interference term (see Section 2.5.3).

2.5.5 Interference between final (final and initial) states

The final-final and final-initial state interference terms have the same structure. The Eq. (2.166) corresponds to final-final state interference term. If we replace \hat{p}_k by \hat{p}_a in Eq. (2.166), then the interference splitting function corresponds to final-initial state interference term. For $a_\ell = 1$ (which corresponds to only two particles in the final state), there are two situations: $\tilde{a} = 0$ and $\tilde{a} \neq 0$, which corresponds to final-final and final-initial state splittings, respectively. Using the variables defined in Section 2.4.3, the interference splitting function Eq. (2.166) can be rewritten as

$$\Delta W = \frac{1}{y \ (p_{\ell} \cdot Q)} \left(\frac{2 \left(1 - v\right) \left(1 - z \left(1 - x_0\right)\right)}{v \left(1 - z \left(1 - x_0\right)\right) + x_0 \left[\lambda \frac{\tilde{a}}{a_{\ell}} + 1\right]} \right)$$
(2.204)

where $\tilde{a} = \tilde{a}(y)$ and the parameter v is given by Eq. (5.23). We can split this function into a singular and a non-singular part, leading to

$$\Delta W = \frac{1}{y (x_0 + z (1 - x_0))} \Delta W_{\text{fin}} = \frac{1}{x y} \Delta W_{\text{fin}}$$
(2.205)

with

$$z = \frac{x - x_0}{1 - x_0} \tag{2.206}$$

⁹See also the discussions in Section 5.2.1 and Section 5.2.2.

and

$$\Delta W_{\text{fin}} = \frac{2}{(p_{\ell} \cdot Q)} \left(\frac{(x_0 + z (1 - x_0)) (1 - v) (1 - z (1 - x_0))}{v (1 - z (1 - x_0)) + x_0 \left[\lambda \frac{\tilde{a}}{a_{\ell}} + 1\right]} \right)$$
$$= \frac{2}{(p_{\ell} \cdot Q)} (x_0 + z (1 - x_0)) \left[\frac{(1 - z (1 - x_0)) + x_0 \left[\lambda \frac{\tilde{a}}{a_{\ell}} + 1\right]}{v (1 - z (1 - x_0)) + x_0 \left[\lambda \frac{\tilde{a}}{a_{\ell}} + 1\right]} - 1 \right]$$
$$= \frac{2}{(p_{\ell} \cdot Q)} x \left[\frac{1 - x + x_0 \left[\lambda \frac{\tilde{a}}{a_{\ell}} + 2\right]}{v (1 + x_0 - x) + x_0 \left[\lambda \frac{\tilde{a}}{a_{\ell}} + 1\right]} - 1 \right]$$
(2.207)

In the following we will discuss $a_{\ell} = 1$ and $a_{\ell} \neq 1$ cases, respectively.

Simplified case: $a_\ell = 1$ and $\tilde{a} = 0$ (Interference between final states)

If \hat{p}_k is the final state spectator, it is straightforward to show that $\tilde{a} = 0$ from Eq. (2.112). Hence, $\tilde{a} = 0$ corresponds to final-final state splitting. The averaged splitting function is then given by

$$\Delta W = \frac{1}{ux^2} \frac{2}{p_\ell \cdot Q} x \left(\frac{(1-x) + ux \left[(1-ux)\tilde{a} + 2 \right]}{v(1+ux-x) + ux \left[(1-ux)\tilde{a} + 1 \right]} - 1 \right) = \frac{1}{ux^2} \Delta W_{\text{fin}} \quad (2.208)$$

where ΔW_{fin} satisfies the following limits

$$\lim_{u \to 0} \Delta W_{\text{fin}} = \frac{2}{p_{\ell} \cdot Q} (1 - x),$$

$$\lim_{x \to 0} \Delta W_{\text{fin}} = \frac{2}{p_{\ell} \cdot Q} \frac{1}{v_r (x = 0) + u (a_0 + 1)},$$

$$\lim_{u \to 0, x \to 0} \Delta W_{\text{fin}} = \frac{2}{p_{\ell} \cdot Q}$$
(2.209)

and

$$\lim_{x \to 0} \tilde{a} = \lim_{u \to 0} \tilde{a} = a_0$$
(2.210)

which for final-final state interference term equals zero ($a_0 = 0$). The function $v_r(x = 0)$ is given by

$$\lim_{x \to 0} v = x v_r(x=0) = x \left[(4v'-2)\sqrt{u(1-u)a_0} + 1 - u + u a_0 \right]$$
(2.211)

Including all prefactors, the integrated splitting function is given by

$$4\pi\alpha_s C_i \mu^{2\epsilon} \int d\zeta_p \,\Delta W = \frac{\alpha_s}{\pi} C_i \left(\frac{4\pi\mu^2}{2p_\ell \cdot Q}\right)^\epsilon \frac{1}{\Gamma(1-\epsilon)} \left(\frac{1}{2\epsilon^2} + \frac{1}{\epsilon} + 3 - \frac{\pi^2}{4} + \mathcal{O}(\epsilon)\right)$$
(2.212)

where C_i is given by Eq. (2.171).

Simplified case: $a_\ell = 1$ and $\tilde{a} \neq 0$ (Interference between final and initial states)

Now, if \hat{p}_k is the initial state parton (\hat{p}_k is replaced by \hat{p}_a), then

$$\tilde{a} = \frac{\hat{p}_a \cdot \hat{p}_k}{(1-y)\,\hat{p}_a \cdot (\hat{p}_\ell + \hat{p}_j)} \neq 0$$
(2.213)

It follows that, $\tilde{a} \neq 0$ corresponds to final-initial state interference term. Including all prefactors, the integrated splitting function is

$$4\pi\alpha_{s} C_{i} \mu^{2\epsilon} \int d\zeta_{p} \Delta W = \frac{\alpha_{s}}{\pi} C_{i} \left(\frac{4\pi\mu^{2}}{2p_{\ell} \cdot Q}\right)^{\epsilon} \frac{1}{\Gamma(1-\epsilon)} \left\{\frac{1}{2\epsilon^{2}} + \frac{1}{\epsilon} \left[1 + \frac{1}{2}\ln(1+a_{0})\right] - \frac{\pi^{2}}{6} + 3 - 2\ln 2\ln(1+a_{0}) + \frac{1}{\pi} \left(I_{\text{fin}}^{b} + I_{\text{fin}}^{c}\right)\right\}$$
(2.214)

where C_i is given by Eq. (2.171) and the finite terms are given by

$$\begin{split} I_{\rm fin}^{b}(a_{0}) &= \frac{\pi}{2} \int_{0}^{1} \frac{du}{u} \left[2\ln 2 + \frac{1}{\sqrt{1 + 4a_{0}(1 + a_{0})u^{2}}} \ln \left[\frac{(1 - u)}{\left(1 + 2a_{0}u + \sqrt{1 + 4a_{0}(1 + a_{0})u^{2}} \right)^{2}} \right] \right] \\ &+ \frac{\pi}{2} \left(2\ln 2\ln(1 + a_{0}) + \frac{1}{2}\ln^{2}(1 + a_{0}) + \frac{5}{2}\text{Li}_{2} \left(\frac{a_{0}}{a_{0} + 1} \right) - \frac{1}{2}\text{Li}_{2} \left[\left(\frac{a_{0}}{a_{0} + 1} \right)^{2} \right] \right) \\ I_{\rm fin}^{c}(\tilde{a}) &= \int_{0}^{1} \frac{du}{u} \int_{0}^{1} \frac{dx}{x} \int_{0}^{1} dv' \left[v'(1 - v') \right]^{-\frac{1}{2}} \\ &\times \left\{ \frac{x \left(1 - x + ux \left[(1 - ux) \tilde{a} + 2 \right] \right)}{v \left[1 + ux - x \right] + ux \left[(1 - ux) \tilde{a} + 1 \right]} - \frac{1}{1 + 2a_{0}u + (4v' - 2)\sqrt{u \left(1 - u \right)a_{0}}} \right\} \\ &= \pi \int_{0}^{1} \frac{du}{u} \int_{0}^{1} \frac{dx}{x} \left\{ \frac{x \left(1 - x + ux \left[(1 - ux) \tilde{a} + 2 \right] \right)}{k(u, x, \tilde{a})} - \frac{1}{\sqrt{1 + 4a_{0}u^{2} \left(1 + a_{0} \right)}} \right\} \end{aligned}$$
(2.215)

We have introduced

$$k^{2}(u, x, \tilde{a}) = \left[(1 + u x - x)(z - \bar{z}) + u x ((1 - u x) \tilde{a} + 1) \right]^{2} + 4 u x \bar{z} (1 - z) (1 + u x - x) ((1 - u x) \tilde{a} + 1)$$
(2.216)

and

$$z = \frac{x(1-u)}{1-ux}, \, \bar{z} = ux\,\tilde{a}, \, \tilde{a} = \frac{p_a \cdot n_\ell}{p_a \cdot p_\ell + y\,p_a \cdot n_\ell}, \, a_0 = \tilde{a}(y=0) = \frac{p_a \cdot n_\ell}{p_a \cdot p_\ell}$$
(2.217)

Here, the parameter v is given by Eq. (5.23). For $\tilde{a} = 0$, the finite terms I_{fin}^b and I_{fin}^c can be reduced to

$$I_{\rm fin}^b = -\frac{\pi^3}{12}, \qquad I_{\rm fin}^c = 0$$
 (2.218)

and we of course obtain the result in the last subsection. All leftover integrals are finite in the limits $u \to 0$ and $x \to 0$.

General case: $a_\ell \neq 1$

We will now consider the integrated splitting function for $a_{\ell} \neq 1$. We have again the factoriziation into a finite and a singular term (see Eq. (2.205)), with the limits of the finite term ΔW_{fin} being given by

$$\lim_{u \to 0} \Delta W_{\text{fin}} = \frac{2}{p_{\ell} \cdot \hat{Q}} (1 - x),$$

$$\lim_{x \to 0} \Delta W_{\text{fin}} = \frac{2}{p_{\ell} \cdot \hat{Q}} \frac{1}{v_r (x = 0) + u \left(\frac{a_0}{a_{\ell}} + 1\right)},$$

$$\lim_{u \to 0, x \to 0} \Delta W_{\text{fin}} = \frac{2}{p_{\ell} \cdot \hat{Q}}$$
(2.219)

Here, the parameter $v_r(x=0)$ is

$$v_r(x=0) = (4v'-2)\sqrt{u(1-u)\frac{a_0}{a_\ell}} + 1 + u\left(\frac{a_0}{a_\ell} - 1\right)$$
(2.220)

Including all prefactors, the integrated splitting function is

$$4\pi\alpha_{s} C_{i} \mu^{2\epsilon} \int d\zeta_{p} \Delta W = \frac{\alpha_{s}}{\pi} C_{i} \left(\frac{4\pi\mu^{2}}{2 p_{\ell} \cdot \hat{Q}}\right)^{\epsilon} \frac{1}{\Gamma(1-\epsilon)} \left\{\frac{1}{2\epsilon^{2}} + \frac{1}{\epsilon} \left[1 + \frac{1}{2}\ln\left(a_{\ell} + a_{0}\right)\right] - \frac{\pi^{2}}{6} + 3 - 2\ln\left(2\ln\left(a_{\ell} + a_{0}\right)\right) + \frac{1}{\pi} \left[I_{\text{fin}}^{b} \left(\frac{a_{0}}{a_{\ell}}\right) + I_{\text{fin}}^{d} + I_{\text{fin}}^{e}\right] + \ln\left(a_{\ell}\right) + \ln\left(2\ln\left(2-\frac{1}{4}\ln\left(a_{\ell} + a_{0}\right)\right) + \frac{1}{2}\ln\left(a_{\ell} + a_{0}\right)\right) + 1\right]\right\}$$
(2.221)

where C_i is given by Eq. (2.171) and the finite terms are

$$I_{\rm fin}^{d} = \int_{0}^{1} \frac{du}{u} \int_{0}^{1} \frac{dx}{x} \int_{0}^{1} dv' \left[v' \left(1 - v' \right) \right]^{-\frac{1}{2}} \\ \times \left\{ \gamma x \left[\frac{1 - x + x_0 \left[\lambda \frac{\tilde{a}}{a_{\ell}} + 2 \right]}{v \left[1 + x_0 - x \right] + x_0 \left[\lambda \frac{\tilde{a}}{a_{\ell}} + 1 \right]} - 1 \right] + x - \frac{1}{v_r (x = 0) + u \left(\frac{a_0}{a_{\ell}} + 1 \right)} \right\} \\ I_{\rm fin}^{e} = \pi \int_{0}^{1} dx \left(\frac{1 - x}{x} \right) \ln \left[\frac{\delta a_{\ell}}{x} \right]$$
(2.222)

with v now being given by

$$v = \frac{1}{1 - x_0} \left\{ (4v' - 2) \left[(1 - x) \left(x - x_0 \right) \bar{z} (1 - \bar{z}) \right]^{\frac{1}{2}} + x - x_0 + \bar{z} \left(1 - x_0 \right) - 2 \bar{z} \left(x - x_0 \right) \right\}$$
(2.223)

We use Eq. (2.113) for \tilde{a} and Eq. (2.111) for \bar{z} . The spectator \hat{p}_k needs to be calculated using the Lorentz transformation

$$\hat{p}_k^{\mu} = \Lambda(K, K)^{\mu}{}_{\nu} p_k^{\nu}$$

Note that for the initial-state splitting, we need to use the original definition of v. The problem for the initial state is that an additional angle appears in the Lorentz transformation; this is not the case for the final state, where we take P_{ℓ} as given by Eq. (2.22), without any reference to the additional angle and/or use of \hat{p}_j . Hence, we never need to use the Sudhakov parametrization for \hat{p}_j in the final state ¹⁰.

The finite term $I_{\text{fin}}^b(a_0/a_\ell)$ in Eq. (2.221) means that a_0 is now being replaced by a_0/a_ℓ in $I_{\text{fin}}^b(a_0)$, which is already given in the last subsection. For $a_\ell = 1$, $I_{\text{fin}}^e = 0$ and $I_{\text{fin}}^d = I_{\text{fin}}^c$. Hence, we obtain the result in the last subsection. All leftover integrals are finite in the limits $u \to 0$ and $x \to 0$. Note also that the treatment of interference terms significantly differs from [22]. Here, our choice of momentum mapping leads to more complicated integrated finite terms, which we choose to evaluate numerically.

¹⁰See also the discussions in Section 5.2.1 and Section 5.2.2.

Chapter 3

Applications

For Nagy-Soper scheme, all collinear as well as singular parts of the soft splitting functions have been tested. In this chapter, we give a numerical comparison for Drell-Yan process at NLO using [22] as well as the Nagy-Soper scheme. We also give the analytic result of our splitting functions when applied to dijet production at lepton colliders, as well as the Higgs production at hadron colliders and decay.

3.1 Single *W* production

3.1.1 Tree level

We start with a simple process: single W production at hadron collider. The W production provides one of the cleanest processes with a large cross section at the Tevatron and at the LHC. This process is not only suited for a precise determination of the W boson mass M_W , it also yields valuable information on the parton structure of the proton. The QCD NLO calculations have been available in the literatures for some time [37–40].

The cross section for W production at hadron collider is

$$A + B \to W^{\pm} + \text{anything}$$
 (3.1)

The parton level subprocess in this case is (Fig. 3.1)

$$q \,\bar{q}' \to W^+ \tag{3.2}$$

where q is a quark with charge 2/3 (or an antiquark with charge 1/3) from hadron A and \bar{q}' is an antiquark with charge 1/3 (or a quark with charge 2/3) from hadron B. Labelling the momenta

by $q(p_2) \, \bar{q}'(p_1) \to W^+(P)$, we have

$$\mathcal{M} = -i V_{qq'} \frac{g}{\sqrt{2}} \varepsilon_{\alpha}^{\lambda*}(P) \bar{v}(p_1) \gamma^{\alpha} \frac{1}{2} \left(1 - \gamma^5\right) u(p_2)$$
(3.3)

Neglecting the quark masses and squaring the matrix element gives

$$\frac{1}{4} \sum_{\text{spins}} |\mathcal{M}|^2 = \frac{g^2}{4} |V_{qq'}|^2 \left(p_1 \cdot p_2 + 2 \frac{p_1 \cdot P \, p_2 \cdot P}{M_W^2} \right)$$
(3.4)

where we have averaged over the incoming parton spins. For the massive on-shell vector boson, we use

$$\sum_{\sigma} \varepsilon_{\sigma}^{\mu*}(P) \varepsilon_{\sigma}^{\nu}(P) = -g^{\mu\nu} + \frac{P^{\mu} P^{\nu}}{M_W^2}$$
(3.5)

Using now $P = p_1 + p_2$ and $P^2 = M_W^2$, we get

$$\frac{1}{4} \sum_{\text{spins}} |\mathcal{M}|^2 = \frac{g^2}{4} |V_{qq'}|^2 M_W^2$$
(3.6)

If we average over the parton colours, we obtain an additional factor $3 \times \frac{1}{3} \times \frac{1}{3} = \frac{1}{3}$ leading to

$$|\mathcal{M}_B|^2 = \frac{1}{4} \frac{1}{9} \sum_{\text{spins, colours}} |\mathcal{M}|^2 = \frac{g^2}{12} |V_{qq'}|^2 M_W^2$$
 (3.7)

The one-particle phase space is given by

$$\int dP S_1 = \int \frac{d^3 \vec{P}}{(2\pi)^3 2 P^0} (2\pi)^4 \,\delta^4 (P - p_1 - p_2) = 2\pi \,\delta^+ (\hat{s} - M_W^2) \tag{3.8}$$

Here, \hat{s} is the center of mass energy of the colliding partons.

3.1.2 Real emission, virtual correction and dipole subtraction

There are two subprocesses needed to be considered (Fig. 3.2 and Fig. 3.3): $q\bar{q} \rightarrow gW$ and $qg \rightarrow qW$. For the subprocess $q\bar{q} \rightarrow gW$, a gluon can be emitted from either the incoming quark or the incoming antiquark. Labelling the momenta by $q(p) \bar{q}(p') \rightarrow g(k') W(k)$, we have

$$i \mathcal{M} = \frac{-i g g_s}{\sqrt{2}} t_{ij}^A \varepsilon_{\mu}^*(k') \tilde{\varepsilon}_{\nu}^*(k) V_{q\bar{q}} \\ \times \bar{v}(p') \left[\gamma^{\mu} \frac{(p'-k')}{(p'-k')^2 + i\epsilon} \gamma^{\nu} \frac{1}{2} (1-\gamma^5) - \gamma^{\nu} \frac{1}{2} (1-\gamma^5) \frac{(p'-k')}{(p-k')^2 + i\epsilon} \gamma^{\mu} \right] u(p) \quad (3.9)$$


Figure 3.1: LO and virtual diagrams.



Figure 3.2: Real emission diagrams: $q\bar{q} \rightarrow g\,W$



Figure 3.3: Real emission diagrams: $qg \rightarrow qW$



Figure 3.4: Collinear splittings.

Here we have neglected the quark/antiquark masses, the ε_{μ} and $\tilde{\varepsilon}_{\mu}$ denote the polarization vectors of gluon and W boson, respectively. They satisfy

$$\sum_{\sigma} \varepsilon_{\sigma}^{\mu*}(k') \varepsilon_{\sigma}^{\rho}(k') = -g^{\mu\rho}$$
$$\sum_{\tau} \tilde{\varepsilon}_{\tau}^{\nu*}(k) \tilde{\varepsilon}_{\tau}^{\sigma}(k) = -g^{\nu\sigma} + \frac{k^{\nu} k^{\sigma}}{M_W^2}$$
(3.10)

Squaring the amplitude and averaging over the parton spins and colours, we obtain the matrix element for NLO real emission process $q\bar{q} \rightarrow g W$

$$|\mathcal{M}_{R}|^{2} = \frac{1}{4} \frac{1}{9} \sum_{\text{spins, colours}} |\mathcal{M}|^{2} = \pi \alpha_{s} \sqrt{2} G_{F} M_{W}^{2} |V_{qq'}|^{2} \frac{32}{9} \frac{\hat{t}^{2} + \hat{u}^{2} + 2M_{W}^{2} \hat{s}}{\hat{t} \hat{u}}$$
$$= \frac{8}{9} g^{2} \pi \alpha_{s} |V_{qq'}|^{2} \frac{\hat{t}^{2} + \hat{u}^{2} + 2M_{W}^{2} \hat{s}}{\hat{t} \hat{u}}$$
(3.11)

where $\hat{s}, \hat{t}, \hat{u}$ are the Mandelstam variables. There are both soft and collinear singularities corresponding to $\hat{t} \to 0$ and $\hat{u} \to 0$. The coupling constant g and the mass of gauge boson M_W are related to the Fermi coupling constant G_F by

$$\frac{G_F}{\sqrt{2}} = \frac{g^2}{8\,M_W^2} \tag{3.12}$$

The two-particle phase space is given by

$$\int dPS_2 = \int \frac{d^3\vec{k}}{(2\pi)^3 2 k_0} \frac{d^3\vec{k}'}{(2\pi)^3 2 k_0'} (2\pi)^4 \delta^4 (Q-k-k')$$
(3.13)





Nagy-Soper dipoles

The calculations of the subtracted splitting functions contain two dipole contributions: \mathcal{D}_{qqg} and $\mathcal{D}_{\bar{q}\bar{q}g}$, each of which contains both collinear and soft splittings (Fig. 3.4 and Fig. 3.5). Their explicit expressions are given in Eq. (2.141) and Eq. (2.161). We find

$$\mathcal{D}_{qqg} = \frac{\hat{s}}{M_W^2} \left[\frac{4\,\hat{u}}{\hat{s}\,\hat{t}} + \frac{8\,(\hat{s}+\hat{u})}{\hat{t}\,(\hat{t}+\hat{u})} - \frac{8\,\hat{s}}{(\hat{t}+\hat{u})^2} \right] + \frac{16\,\hat{s}\,\hat{u}^2}{(\hat{t}^2+\hat{u}^2)\,(\hat{t}+\hat{u})^2} \tag{3.14}$$

The dipole contribution $\mathcal{D}_{\bar{q}\bar{q}g}$ can be obtained from Eq. (3.14) by the replacement $\hat{t} \leftrightarrow \hat{u}$. The final expression for the two-particle cross section is given by

$$\sigma^{NLO\left\{2\right\}} = \int_{2} \left[d\sigma_{\epsilon=0}^{R} - d\sigma_{\epsilon=0}^{A} \right]$$
$$= \frac{1}{2\hat{s}} \int dPS_{2} \left\{ |\mathcal{M}_{R}|^{2} - \left(\frac{4\pi\alpha_{s}}{2}\right) C_{F} \left(\mathcal{D}_{qqg} + \mathcal{D}_{\bar{q}\bar{q}g}\right) |\mathcal{M}_{B}|^{2} \right\}$$
(3.15)

Eq. (3.15) is completely finite after subtracting the dipoles. For the *m*-parton phase space we use the results of the integrations of the splitting functions over the emitted one-parton phase space. All the collinear and soft integrals are given in Section 2.5. Using Eq. (2.16), we have

$$\int_{1} d\sigma_{ab}^{B}(xp_{a}, p_{b}) \left\{ \frac{4\pi\alpha_{s}}{2} C_{F} \mu^{2\epsilon} \int d\zeta_{p} \left[v_{qqg}^{2} - v_{eik}^{2} \right] + 4\pi\alpha_{s} C_{F} \mu^{2\epsilon} \int d\zeta_{p} \Delta W_{ab} \right\}$$

$$+ \int_{1} d\sigma_{ab}^{B}(p_{a}, xp_{b}) \left\{ \frac{4\pi\alpha_{s}}{2} C_{F} \mu^{2\epsilon} \int d\zeta_{p} \left[v_{qqg}^{2} - v_{eik}^{2} \right] + 4\pi\alpha_{s} C_{F} \mu^{2\epsilon} \int d\zeta_{p} \Delta W_{ab} \right\}$$

$$+ \int_{1} d\sigma_{ab}^{C}(p_{a}, p_{b}, \mu_{F}^{2})$$

$$= \int_{1} d\sigma_{ab}^{B}(p_{a}, p_{b}) \otimes I(\epsilon) + \int_{0}^{1} dx \int_{1} d\sigma_{ab}^{B}(xp_{a}, p_{b}) \otimes \left[K^{a}(xp_{a}) + P(x, \mu_{F}^{2}) \right]$$

$$+ (a \leftrightarrow b) \qquad (3.16)$$

where the universal collinear counter terms for any finite hard scattering m-parton cross section are defined by Eq. (2.14). The corresponding I, K and P terms are given by

$$I(\epsilon) = \frac{\alpha_s}{2\pi} C_F \frac{1}{\Gamma(1-\epsilon)} \left(\frac{4\pi\mu^2}{Q^2} \right)^{\epsilon} \left(\frac{2}{\epsilon^2} + \frac{3}{\epsilon} - \frac{\pi^2}{3} + \mathcal{O}(\epsilon) \right)$$

$$K^a(xp_a) = \frac{\alpha_s}{2\pi} C_F \frac{1}{\Gamma(1-\epsilon)} \left[-(1-x)\ln x + 2(1-x)\ln(1-x) + 4x \left(\frac{\ln(1-x)}{1-x} \right)_+ \right]$$

$$- \frac{2x\ln x}{(1-x)_+} - \left(\frac{1+x^2}{1-x} \right)_+ \ln \left(\frac{4\pi\mu^2}{2xp_a \cdot p_b} \right) \right]$$

$$P(x, \mu_F^2) = \frac{\alpha_s}{2\pi} C_F \frac{1}{\Gamma(1-\epsilon)} \left(\frac{1+x^2}{1-x} \right)_+ \ln \left(\frac{4\pi\mu^2}{\mu_F^2} \right)$$
(3.17)

The plus prescription is defined by

$$\int_0^1 dx \, f(x) \, F(x)_+ = \int_0^1 dx \, \left[f(x) - f(1) \right] \, F(x) \tag{3.18}$$

or alternatively,

$$F(x)_{+} = \lim_{\epsilon \to 0} \left[\theta(1 - x - \epsilon) F(x) - \delta(1 - x) \int_{0}^{1 - \epsilon} dx' F(x') \right]$$

= $F(x) - \delta(1 - x) \int_{0}^{1} dx' F(x')$ with $\theta(1 - x) = 1$ (3.19)

From that it follows immediately that

$$\int_{0}^{1} dx F(x)_{+} = 0 \tag{3.20}$$

 $\quad \text{and} \quad$

$$P_{qq}(x) = C_F \left(\frac{1+x^2}{1-x}\right)_+ = C_F \left[\frac{1+x^2}{(1-x)_+} + \frac{3}{2}\delta(1-x)\right], \qquad C_F = \frac{4}{3}$$
(3.21)

appearing in the K terms is the famous Altarelli-Parisi splitting function. The virtual matrix element in the \overline{MS} renormalization scheme is given by

$$|\mathcal{M}_{V}|^{2} = |\mathcal{M}_{B}|^{2} \frac{\alpha_{s}}{2\pi} C_{F} \left(\frac{4\pi\mu^{2}}{Q^{2}}\right)^{\epsilon} \frac{1}{\Gamma(1-\epsilon)} \left\{-\frac{2}{\epsilon^{2}} - \frac{3}{\epsilon} - 8 + \pi^{2} + \mathcal{O}(\epsilon)\right\}$$
(3.22)

We see that the singularities cancel each other between $\mid \mathcal{M}_V \mid^2$ and $I(\epsilon)$ as they must.

Now we consider the gluon induced process $qg \rightarrow qW$. The matrix element can be obtained from Eq. (3.11) by crossing symmetry. If we include the colours and spins, we obtain

$$|\mathcal{M}_{R}|^{2} = \frac{1}{4} \frac{1}{24} \sum_{\text{spins, colours}} |\mathcal{M}|^{2} = \frac{1}{3} g^{2} \pi \alpha_{s} |V_{qq'}|^{2} \frac{\hat{s}^{2} + \hat{u}^{2} + 2M_{W}^{2}\hat{t}}{-\hat{s}\,\hat{u}}$$
(3.23)

It is worth mentioning that no soft singularity is present in the process $qg \rightarrow qW$; $\hat{u} \rightarrow 0$ corresponds to collinear singularity. There is only one dipole needed in this case $g \rightarrow q\bar{q}$, which in terms of Mandelstam variables, is given by

$$\mathcal{D}_{gq\bar{q}} = -\left(\frac{4}{M_W^2}\right) \frac{\hat{s}^2 + 2\,\hat{s}\,\hat{t} + 2\,\hat{t}^2}{\hat{s}\,\hat{u}} \tag{3.24}$$

We found that

$$\sigma^{NLO\{2\}} = \int_{2} \left[d\sigma^{R}_{\epsilon=0} - d\sigma^{A}_{\epsilon=0} \right] = \frac{1}{2\,\hat{s}} \int dPS_{2} \left\{ |\mathcal{M}_{R}|^{2} - \left(\frac{4\,\pi\,\alpha_{s}}{2}\right) T_{R} \mathcal{D}_{gq\bar{q}} |\mathcal{M}_{B}|^{2} \right\} \\ = \frac{1}{2\,\hat{s}} \int dPS_{2} \left\{ -\frac{1}{3} g^{2} \pi\,\alpha_{s} \frac{(2\,\hat{t}+\hat{u})}{\hat{s}} \right\}$$
(3.25)

The collinear singularity appearing in *m*-parton phase spaces will be absorbed into PDFs, when we combine the integrated splitting function with the collinear counter term Eq. (2.14).

$$\int_{1} d\sigma_{ab}^{B}(xp_{a}, p_{b}) \left\{ \frac{4\pi \alpha_{s}}{2(1-\epsilon)} T_{R} \mu^{2\epsilon} \int d\zeta_{p} v_{gq\bar{q}}^{2} \right\} + \int_{1} d\sigma_{ab}^{C}(p_{a}, p_{b}, \mu_{F}^{2}) = \int_{0}^{1} dx \int_{1} d\sigma_{ab}^{B}(xp_{a}, p_{b}) \otimes \left[K(xp_{a}) + P(x, \mu_{F}^{2}) \right]$$
(3.26)

where

$$K(xp_{a}) = \frac{\alpha_{s}}{2\pi} \frac{1}{\Gamma(1-\epsilon)} \left\{ T_{R}(6x-5x^{2}-1) + [2\ln(1-x)-\ln(x)] P_{qg}(x) - P_{qg}(x) \ln\left(\frac{4\pi\mu^{2}}{2xp_{a}\cdot p_{b}}\right) \right\}$$
(3.27)
$$P(x,\mu_{F}^{2}) = \frac{\alpha_{s}}{2\pi} \frac{1}{\Gamma(1-\epsilon)} P_{qg}(x) \ln\left(\frac{4\pi\mu^{2}}{\mu_{F}^{2}}\right) - P_{qg}(x) = T_{R} \left[x^{2} + (1-x)^{2}\right], \quad T_{R} = \frac{1}{2}$$

and $P_{qg}(x)$ is the standard Altarelli-Parisi splitting function.

Catani-Seymour dipoles

Using Catani-Seymour dipoles, the K and P terms for process $qg \rightarrow qW$ are given by

$$K(xp_{a}) = \frac{\alpha_{s}}{2\pi} \frac{1}{\Gamma(1-\epsilon)} \left\{ 2P_{qg}(x) \ln(1-x) + T_{R}2x(1-x) - P_{qg}(x) \ln\left(\frac{4\pi\mu^{2}}{2xp_{a}\cdot p_{b}}\right) - P_{qg}(x) \ln x \right\}$$

$$P(x,\mu_{F}^{2}) = \frac{\alpha_{s}}{2\pi} \frac{1}{\Gamma(1-\epsilon)} P_{qg}(x) \ln\left(\frac{4\pi\mu^{2}}{\mu_{F}^{2}}\right)$$
(3.28)

and the I, K and P terms for process $q\bar{q} \rightarrow gW$ are given by

$$I(\epsilon) = \frac{\alpha_s}{2\pi} C_F \frac{1}{\Gamma(1-\epsilon)} \left(\frac{4\pi\mu^2}{Q^2}\right)^{\epsilon} \left(\frac{2}{\epsilon^2} + \frac{3}{\epsilon} - \frac{\pi^2}{3} + \mathcal{O}(\epsilon)\right)$$

$$K^a(xp_a) = \frac{\alpha_s}{2\pi} C_F \frac{1}{\Gamma(1-\epsilon)} \left[4 \left(\frac{\ln(1-x)}{1-x}\right)_+ + (1-x) - 2(1+x)\ln(1-x) - \left(\frac{1+x^2}{1-x}\right)_+ \ln\left(\frac{4\pi\mu^2}{2xp_a \cdot p_b}\right) - \left(\frac{1+x^2}{1-x}\right)_+ \ln x\right]$$

$$P(x, \mu_F^2) = \frac{\alpha_s}{2\pi} C_F \frac{1}{\Gamma(1-\epsilon)} \left(\frac{1+x^2}{1-x}\right)_+ \ln\left(\frac{4\pi\mu^2}{\mu_F^2}\right)$$
(3.29)

Results

In summary, we see that the pole structures in $I(\epsilon)$ terms are equivalent between Catani-Seymour's scheme and Nagy-Soper's scheme as they should. The finite terms in K and P are shifted around due to different momentum mapping. However the final results are the same. For a comparison, we calculated single W production for a pp initial state at NLO, using both the scheme in [22] as well as Nagy-Soper scheme, including PDFs¹ (CTEQ6M [101]) and varying the hadronic center of mass energy of the process. Here we have used routines from the CUBA library (Vegas) [99] as a Monte Carlo algorithm to implement our numerical phase space calculations. Fig. 3.6 shows the relative difference between the two implemented schemes. We see, that the numerical differences are on the permill level and consistent with zero. Fig. 3.7 plots the NLO corrections to single W production at the LHC as a function of the hardonic center of mass energy.

3.2 Dijet production in e^+e^- annihilation

Next we consider dijet production at NLO. The LO and NLO diagrams of dijet production are shown in Fig. 3.8 and Fig. 3.9. The kinematics of two-jet production is defined as follows: The variables are $x_i = 2 p_i \cdot Q/Q^2$, $y_{ij} = s_{ij}/Q^2$ and $s_{ij} = (p_i + p_j)^2$. The squared expression of Q corresponds to the square of the center of mass energy and p_i is the momentum of any QCD parton in the final state. They are related by $x_1 = 1 - y_{23}$, $x_2 = 1 - y_{13}$ and $x_3 = 1 - y_{12}$. We can choose the center of mass frame in which $Q = (\sqrt{s}, \vec{0})$ and $p_i = (E_i, \vec{p_i})$, hence $\sum_i E_i = \sqrt{s}$ and $\sum_i \vec{p_i} = \vec{0}$. \sqrt{s} is the center of mass energy. It is straightforward to show that $x_1 + x_2 + x_3 = 2$.

¹c.f. [102].



Figure 3.6: Relative difference between NLO corrections to single W production using Catani-Seymour and Nagy-Soper dipoles respectively, as a function of the hardonic center of mass energy. The results agree on sub-permil level. Additionally the numerical integration errors are shown.



Figure 3.7: NLO corrections $(\sigma_{\text{NLO}} - \sigma_{\text{Born}})/\sigma_{\text{Born}}$ to single W production at the LHC as a function of the hardonic center of mass energy. The result was obtained using the CTEQ6M parton distribution function [101].



Figure 3.8: LO and virtual diagrams.

The LO contribution is the parton model process $e^+e^- \rightarrow q(p_1) \bar{q}(p_2)$ with matrix element \mathcal{M}_2 . We average over the event orientation in the LO process. In this case, the momentum dependence of the Born contribution vanishes. The NLO real emission process is

$$e^+ e^- \to \gamma^*(Q) \to q(p_1) \,\bar{q}(p_2) \,g(p_3)$$
 (3.30)

The scattering amplitude for $e^+(p')e^-(p) \rightarrow q(p_1) \bar{q}(p_2) g(p_3)$ is

$$i\mathcal{M} = -\frac{e^2 g_s}{Q^2} t_{ij}^A Q_q \,\varepsilon^{\lambda *}(p_3) \,\bar{v}(p') \,\gamma_\mu \,u(p) \,\left\{ \bar{u}(p_1) \,\gamma_\lambda \frac{1}{p_1' + p_3'} \,\gamma^\mu \,v(p_2) - \bar{u}(p_1) \,\gamma^\mu \frac{1}{p_2' + p_3'} \,\gamma_\lambda \,v(p_2) \right\}$$
(3.31)

where $Q = p_1 + p_2 + p_3$ and $\varepsilon^{\lambda}(p_3)$ is the polarization vector of the gluon. Squaring the scattering amplitude gives then

$$\frac{1}{4}\sum |\mathcal{M}|^2 = -\frac{1}{4}\frac{e^4 g_s^2}{Q^4} \left(t_{ij}^A\right)^2 Q_q^2 \operatorname{Tr}(p'' \gamma^\mu \not p \gamma^\nu) \operatorname{Tr}(p_1 \Lambda_{\lambda\mu} p_2 \Lambda_{\nu\lambda})$$
(3.32)

where

$$\Lambda_{\lambda\mu} = \left(\gamma_{\lambda} \frac{-1}{p_{1}^{\prime} + p_{3}^{\prime}} \gamma_{\mu} + \gamma_{\mu} \frac{1}{p_{2}^{\prime} + p_{3}^{\prime}} \gamma_{\lambda}\right)$$

$$= \left(\frac{-1}{Q^{2} (1 - x_{2})} \gamma_{\lambda} (p_{1}^{\prime} + p_{3}^{\prime}) \gamma_{\mu} + \frac{1}{Q^{2} (1 - x_{1})} \gamma_{\mu} (p_{2}^{\prime} + p_{3}^{\prime}) \gamma_{\lambda}\right)$$
(3.33)

After a straightforward calculation we find the matrix element $\mathcal{M}_3(p_1, p_2, p_3)$

$$\left|\mathcal{M}_{3}(p_{1}, p_{2}, p_{3})\right|^{2} = C_{F} \frac{8 \pi \alpha_{s}}{Q^{2}} \frac{x_{1}^{2} + x_{2}^{2}}{(1 - x_{1})(1 - x_{2})} \left|\mathcal{M}_{2}\right|^{2}$$
(3.34)

The final-state parton momenta are denoted by p_i , where,

$$s_{12} = (p_1 + p_2)^2, \quad s_{13} = (p_1 + p_3)^2, \quad s_{23} = (p_2 + p_3)^2$$

$$s_{123} = (p_1 + p_2 + p_3)^2 = s_{12} + s_{13} + s_{23} = Q^2$$

$$y_{12} = s_{12}/Q^2, \quad y_{13} = s_{13}/Q^2, \quad y_{23} = s_{23}/Q^2, \quad y_{12} + y_{13} + y_{23} = 1$$
(3.35)



Figure 3.9: Real emission diagrams: $e^+e^- \rightarrow q\bar{q}g$.

or equivalently we have

$$s_{ij} = (p_i + p_j)^2$$
, $s_{ijk} = (p_i + p_j + p_k)^2$, $y_{ij} = s_{ij}/Q^2$, $(i < j < k)$ (3.36)

For the two-parton phase space integration, we follow the Catani-Seymour's convention, which is slightly different from the standard convention up to a factor of 8π . Hence

$$\int_{CS} dPS_2 = \int dy_{12} \,\delta(1 - y_{12}) = 8 \,\pi \,\int dPS_2 \tag{3.37}$$

and the LO cross section is given by

$$\sigma^{LO} = \int_{CS} dPS_2 |\mathcal{M}_2|^2 F_J^{(2)}(p_1, p_2)$$

= $|\mathcal{M}_2|^2 \int dy_{12} \,\delta(1 - y_{12}) F_J^{(2)}(p_1, p_2) \rightarrow |\mathcal{M}_2|^2$ (3.38)

The three-parton phase space is given by

$$\int_{CS} dPS_3 = \frac{Q^2}{16\pi^2} \int_0^1 dx_1 \int_0^1 dx_2 \,\theta(x_1 + x_2 - 1) = 8\pi \int dPS_3$$
$$= \frac{Q^2}{16\pi^2} \int_0^1 dy_{23} \int_0^1 dy_{13} \,\theta(1 - y_{23} - y_{13})$$
(3.39)

The calculation of the subtracted splitting functions contains two dipole contributions: \mathcal{D}_{qqg} and $\mathcal{D}_{\bar{q}\bar{q}g}$, each of which contains both collinear and soft contributions (see Fig. 3.10). Their definition is given in Eq. (2.119) and Eq. (2.161). We find that

$$\mathcal{D}_{qqg} = \frac{4}{\hat{Q}^2} \left\{ \left(\frac{1}{x_2} \right) \left[2 \left(\frac{x_1}{2 - x_1 - x_2} - \frac{1 - x_2}{(2 - x_1 - x_2)^2} \right) + \frac{1 - x_1}{1 - x_2} \right] + 2 \left(\frac{x_1 + x_2 - 1}{1 - x_2} \right) \frac{x_1}{(1 - x_1) x_1 + (1 - x_2) x_2} \right\}$$
(3.40)



Figure 3.10: Soft and collinear diagrams.

where x_1 , x_2 and x_3 are defined by

$$x_1 = \frac{2\,\hat{p}_\ell \cdot \hat{Q}}{\hat{Q}^2}, \quad x_2 = \frac{2\,\hat{p}_k \cdot \hat{Q}}{\hat{Q}^2}, \quad x_3 = \frac{2\,\hat{p}_j \cdot \hat{Q}}{\hat{Q}^2}, \quad \hat{Q} = \hat{p}_\ell + \hat{p}_j + \hat{p}_k = Q \tag{3.41}$$

The dipole contribution $\mathcal{D}_{\bar{q}\bar{q}g}$ can be obtained from Eq. (3.40) by the replacement $x_1 \leftrightarrow x_2$. The final expression for the three-parton cross section is given by

$$\sigma^{NLO \{3\}} = \int_{3} \left[d\sigma_{\epsilon=0}^{R} - d\sigma_{\epsilon=0}^{A} \right]$$

$$= \int_{CS} dPS_{3} \left\{ |\mathcal{M}_{3}(p_{1}, p_{2}, p_{3})|^{2} F_{J}^{(3)}(p_{1}, p_{2}, p_{3}) - \left(\frac{4\pi\alpha_{s}}{2}\right) C_{F} \left(\mathcal{D}_{qqg} F_{J}^{(2)}(\tilde{p}_{13}, \tilde{p}_{2}) + \mathcal{D}_{\bar{q}\bar{q}g} F_{J}^{(2)}(\tilde{p}_{23}, \tilde{p}_{1}) \right) |\mathcal{M}_{2}|^{2} \right\}$$

$$= \left(\frac{\alpha_{s}}{2\pi} C_{F}\right) |\mathcal{M}_{2}|^{2} \int_{0}^{1} dx_{1} dx_{2} \theta(x_{1} + x_{2} - 1) \left\{\frac{x_{1}^{2} + x_{2}^{2}}{(1 - x_{1})(1 - x_{2})} F_{J}^{(3)}(p_{1}, p_{2}, p_{3}) - \left[\left(\frac{1}{x_{2}}\right) \left(2\left(\frac{x_{1}}{2 - x_{1} - x_{2}} - \frac{1 - x_{2}}{(2 - x_{1} - x_{2})^{2}}\right) + \frac{1 - x_{1}}{1 - x_{2}}\right)$$

$$+ 2\left(\frac{x_{1} + x_{2} - 1}{1 - x_{2}}\right) \frac{x_{1}}{(1 - x_{1})x_{1} + (1 - x_{2})x_{2}} \right] F_{J}^{(2)}(\tilde{p}_{13}, \tilde{p}_{2})$$

$$- \left[\left(\frac{1}{x_{1}}\right) \left(2\left(\frac{x_{2}}{2 - x_{1} - x_{2}} - \frac{1 - x_{1}}{(2 - x_{1} - x_{2})^{2}}\right) + \frac{1 - x_{2}}{1 - x_{1}}\right)$$

$$+ 2\left(\frac{x_{1} + x_{2} - 1}{1 - x_{1}}\right) \frac{x_{2}}{(1 - x_{1})x_{1} + (1 - x_{2})x_{2}} \right] F_{J}^{(2)}(\tilde{p}_{23}, \tilde{p}_{1}) \right\}$$

$$(3.42)$$

which for any infrared safe observable (implying that $F_J^{(3)} \to F_J^{(2)}$ as x_i approaches 1) is finite and which for $F_J^{(3)} = F_J^{(2)} = 1$ can be reduced to

$$\sigma^{NLO\{3\}} = \frac{\alpha_s}{2\pi} C_F \left(\frac{23}{2} - \frac{4}{3}\pi^2\right) \sigma^{LO}$$
(3.43)



Figure 3.11: Vertex and quark self energy diagrams.

where we have used the formulae

$$\int_{0}^{1} \int_{1-x_{1}}^{1} dx_{1} dx_{2} \frac{(1-x_{1})^{3}}{x_{2} (2-x_{1}-x_{2})^{2}} = \frac{1}{6} (-29+3\pi^{2})$$

$$\int_{0}^{1} \int_{1-x_{1}}^{1} dx_{1} dx_{2} \frac{(1-x_{1}) (1-x_{2})^{2}}{x_{2} (2-x_{1}-x_{2})^{2}} = \frac{1}{12} (-19+2\pi^{2})$$

$$\int_{0}^{1} \int_{1-x_{1}}^{1} dx_{1} dx_{2} \frac{2x_{1} (1-x_{1})}{(2-x_{1}-x_{2})^{2}} = \frac{2}{3}$$

Now we have to use the results of the integrated splitting functions for the *m*-parton phase space contributions which, in general, can be devided into two pieces: soft integral and collinear integral. In the end we have to combine the integrated splitting functions with the virtual cross section leading to finite result. The one-loop matrix element (Fig. 3.11) in the $\overline{\text{MS}}$ renormalization scheme is given by Eq. (3.22), and the collinear and soft integrals can be looked up in Section 2.5. Combining these contributions, we obtain a finite ($\epsilon \rightarrow 0$) expression for the two-parton cross section:

$$\sigma^{NLO \{2\}} = \int_{2} \left[d\sigma^{V} + \int_{1} d\sigma^{A} \right]_{\epsilon=0}$$

$$= \int_{CS} dPS_{2} \left\{ |\mathcal{M}_{V}|^{2} + 2|\mathcal{M}_{2}|^{2} \times \left[\frac{4\pi\alpha_{s}}{2} C_{F} \mu^{2\epsilon} \int d\zeta_{p} \left[v_{qqg}^{2} - v_{eik}^{2} \right] + 4\pi\alpha_{s} C_{F} \mu^{2\epsilon} \int d\zeta_{p} \Delta W \right] \right\} F_{J}^{(2)}(p_{1}, p_{2})$$

$$= \frac{\alpha_{s}}{2\pi} C_{F} \left(-10 + \frac{4\pi^{2}}{3} \right) |\mathcal{M}_{2}|^{2} \int dy_{12} \,\delta(1 - y_{12}) F_{J}^{(2)}(p_{1}, p_{2})$$

$$= \frac{\alpha_{s}}{2\pi} C_{F} \left(-10 + \frac{4}{3}\pi^{2} \right) \sigma^{LO}$$
(3.44)

$2E_g/\sqrt{s}$	$3 \cdot 10^{-4}$	$3 \cdot 10^{-8}$
σ_{real} [pb]	$-0.0051499031 \pm 0.0000051413$	$-0.0051496084 \pm 0.0000051484$

Table 3.1: Variation of the real radiation subtracted cross section σ_{real} for two different values of E_g . If $2 E_g / \sqrt{s} < 3 \cdot 10^{-4}$, the result is the same. Here $\sqrt{s} = 500$ GeV.

Results

Summing Eq. (3.42) and Eq. (3.44), it is straightforward to show that the total NLO cross section (for $F_J^{(3)} = F_J^{(2)} = 1$) agrees with the well known result [22, 109],

$$\sigma^{NLO} = \sigma^{NLO\{2\}} + \sigma^{NLO\{3\}} = \frac{3}{4} \frac{\alpha_s}{\pi} C_F \sigma^{LO}$$
(3.45)

where σ^{LO} is given by

$$\sigma^{LO} = \sigma_0 = \frac{4\pi\alpha^2}{3Q^2} \left(3\sum_{q=1}^{n_f} Q_q^2\right)$$
(3.46)

Note, that here we can set jet functions being equal to one safely. Non-unit jet functions need to account for the mappings from m + 1 to m-phase space.

We also show our numerical results for dijet production. For the numerical computation we used the following parameters²

$$\alpha_s = 0.118055085612548, \quad \alpha = 7.54677226134035754 \times 10^{-3}$$

and vary the center of mass energy. The Fig. 3.12 plots the relative difference between the two implemented schemes. We see that the schemes are equivalent with agreement on the permill level and consistent with zero³. Fig. 3.13 and Fig. 3.14 plot a comparison of analytical result and results using Nagy-Soper and Catani-Seymour dipoles, respectively. We also show a comparison of analytical and numerical results using Nagy-Soper dipoles for the real emission subtracted cross section (see Fig. 3.15). Here we used routines from the CUBA library [99] as a Monte Carlo algorithm to implement our numerical evaluations⁴.

²Here we set quark flavour to be d quark and hence $Q_q = -\frac{1}{3}$.

³We also found that the number of total iterations of Vegas Monte Carlo integrations for the Nagy-Soper scheme is less than the Catani-Seymour scheme under the same accuracy, *e.g.* 259 for NS scheme and 498 for CS scheme. So we are confident that for a multi-particle process in the final state the CPU run time for NS scheme would be much less than the CS scheme.

⁴We also performed the Lorentz boost using our routines to check the Lorentz invariance of the real emission matrix element and all the dipoles; they completely agree with each other before and after the boost.



Figure 3.12: Relative difference between NLO corrections to dijet production using Catani-Seymour and Nagy-Soper dipoles respectively, as a function of the center of mass energy. The results agree on sub-permil level, shown are the numerical integration errors.

We have also checked, that the final result is insensitive to variations of the gluon energy, if the ratio $2 E_g / \sqrt{s}$ is below $3 \cdot 10^{-4}$. The Table 3.1 shows the real radiation subtracted cross section σ_{real} , where we vary the ratio between $3 \cdot 10^{-4}$ and $3 \cdot 10^{-8}$. Here we choose $\sqrt{s} = 500$ GeV.

3.3 Higgs production in gluon-gluon fusion: $gg \rightarrow H$

One of the most crucial experimental challenges for present and future high-energy physics is to search for Higgs boson, which is a fundamental ingredient of the Standard Model (SM). The discovery of Higgs boson will also enable us to well establish the Higgs mechanism, one of the cornerstones of the SM. The Higgs mechanism can not only explain the source of Electroweak Symmetry Breaking (EWSB) but also the generation of elementary particle masses.

For Higgs boson masses up to 700 GeV the dominant production process in the Standard Model is gluon-gluon fusion $gg \to H$ via a virtual top quark loop. Higher mass values may also be realized in extensions of the Beyond Standard Model, *e.g.* Supersymmetry [100]. In the following we will re-calculate QCD radiative corrections of $\mathcal{O}(\alpha_s^3)$ to Higgs boson production in hadronic collisions using both the Catani-Seymour and Nagy-Soper dipoles. These include the one-loop virtual contributions (Fig. 3.16) to the lowest-order process $gg \to H$ as well as real



Figure 3.13: Analytic result vs Nagy-Soper scheme.



Figure 3.14: Analytic result vs Catani-Seymour scheme.



Figure 3.15: Nagy-Soper scheme: a comparison of analytical and numerical results for the real subtracted cross section.

gluon emissions (Fig. 3.17)

$$gg \to g H, \quad qg \to q H, \quad q\bar{q} \to g H$$
 (3.47)

The corrections increase the LO cross section by approximately 50 %. This correction is larger than the corresponding correction to the Drell-Yan process. This is related to the integer/fractional colour charges of gluons/quarks. The lowest-order cross section has been available for some time [44–47],

$$\hat{\sigma}_0(gg \to H) = \frac{\alpha_s^2}{\pi} \frac{M_H^2}{256 v^2} |A|^2 \,\delta\left(\hat{s} - M_H^2\right) \tag{3.48}$$

where

$$|A|^{2} = \left| \sum_{q} \tau_{q} \left(1 + (1 - \tau_{q}) f(\tau_{q}) \right) \right|^{2}$$

$$\tau_{q} = \frac{4 M_{q}^{2}}{M_{H}^{2}}, \quad v^{2} = \frac{4 M_{W}^{2}}{g^{2}} = \frac{1}{\sqrt{2} G_{F}} = (246 \,\text{GeV})^{2}$$
(3.49)

and

$$f(\tau_q) = \begin{cases} \left[\sin^{-1}\left(\sqrt{1/\tau_q}\right)\right]^2 & \text{if } \tau_q \ge 1\\ \\ -\frac{1}{4}\left[\ln\left(\frac{1+\sqrt{1-\tau_q}}{1-\sqrt{1-\tau_q}}\right) - i\pi\right]^2 & \text{if } \tau_q < 1 \end{cases}$$
(3.50)



Figure 3.16: Generic diagrams for the Higgs production in gluon-gluon collisions $gg \rightarrow H$ at LO and its one loop virtual corrections.



Figure 3.17: Generic diagrams for real corrections: $gg \rightarrow gH$, $qg \rightarrow qH$ and $q\bar{q} \rightarrow gH$.

The Lagrangian of ggH coupling has a closed form

$$\mathcal{L}_{ggH} = \frac{\alpha_s}{12\pi} G^A_{\mu\nu} G^{\mu\nu}_A \ln\left(1 + \frac{H}{v}\right)$$
(3.51)

with

$$\ln(1+x) = -\sum_{n=1}^{\infty} \frac{(-x)^n}{n}$$
(3.52)

where $G^A_{\mu\nu}$ is the gluon field strength tensor. Eq. (3.51) is very convenient for simple calculations, but for gg + jets production it only holds in the limit that all jet momenta are much smaller than top quark mass. It also becomes problematic in the $gg \rightarrow HH$ or $gg \rightarrow HHH$ processes close to threshold, where the momenta of slow-moving Higgs bosons lead to an additional scale in the process.

In the limit that the top quark mass is infinitely large, $\tau_q \to \infty$, $A \to \frac{2}{3}$ and

$$\hat{\sigma}_0(gg \to H) \to \frac{\alpha_s^2}{\pi} \frac{M_H^2}{576 v^2} \delta\left(\hat{s} - M_H^2\right)$$
(3.53)

In this thesis we will consider only heavy quark limit (Fig. 3.18, Fig. 3.19 and Fig. 3.20). When the momentum transfer to the Higgs boson is small, or equivalently in the limit where $M_{\text{top}} \gg M_H$, the cross section to $\mathcal{O}(\alpha_s^3)$ can be obtained from the effective Lagrangian [48–50]

$$\mathcal{L}_{\text{eff}} = \frac{\alpha_s}{12 \pi v} H G^A_{\mu\nu} G^{\mu\nu}_A$$
(3.54)

The full NLO QCD cross section contains

$$\int d\sigma^{NLO} = \int d\sigma_{q\bar{q}\to gH} + \int d\sigma_{qg\to qH} + \int d\sigma_{gg\to gH} + \int d\sigma^V$$
(3.55)

In the following we will discuss each subprocess and its corresponding dipoles. The NLO QCD calculations have already been available in the literatures for some time [51–55].

3.3.1 The subprocess $q\bar{q} \rightarrow gH$

Using the effective Lagrangian Eq. (3.54), the matrix element for NLO real emission process $q\bar{q} \rightarrow gH$ in four dimensions is found to be

$$|\mathcal{M}(q\bar{q} \to gH)|^2 = \frac{16}{9} \frac{\alpha_s^3}{\pi v^2} \frac{\hat{t}^2 + \hat{u}^2}{\hat{s}}$$
(3.56)

The cross section for this process is completely finite. The spin and colour averages yield an additional factor which equals $1/2 \times 1/2 \times 1/3 \times 1/3 = 1/36$. It is straightforward to integrate



Figure 3.18: $gg \rightarrow H$: Heavy quark limit.



Figure 3.19: Heavy quark limit for $gg \rightarrow gH$, $qg \rightarrow qH$ and $q\bar{q} \rightarrow gH$.

over the phase space to obtain the cross section.

$$\sigma(q\bar{q} \to gH) = \frac{1}{2\hat{s}} \int dP S_2 |\mathcal{M}(q\bar{q} \to gH)|^2$$
$$= \frac{1}{486} \left(\frac{\alpha_s^3}{\pi^2 v^2}\right) \left(1 - \frac{M_H^2}{\hat{s}}\right)^3$$
(3.57)

The two-particle phase space integration is given by

$$dPS_2 = \frac{1}{32\pi^2} \left(1 - \frac{M_H^2}{\hat{s}} \right) d\Omega$$
(3.58)

provided that

$$\hat{s} > M_H^2, \qquad \hat{s} = \eta_a \eta_b s \tag{3.59}$$

Here, η_a and η_b are the momentum fractions of the incoming partons. The parameter \hat{s} is the center of mass energy of the colliding partons, while s is the center of mass energy of incoming hadrons.

3.3.2 The subprocess $qg \rightarrow qH$

Nagy-Soper dipoles

The matrix element for NLO real emission process $qg \rightarrow qH$ in four dimensions can be obtained by crossing from that for $q\bar{q} \rightarrow gH$. We find

$$\left|\mathcal{M}(qg \to qH)\right|^{2} = -\frac{16}{9} \frac{\alpha_{s}^{3}}{\pi v^{2}} \frac{\hat{s}^{2} + \hat{u}^{2}}{\hat{t}}$$
(3.60)

which has a singularity when $\hat{t} \to 0$ (collinear singularity). The spin and colour averages yield an additional factor $1/2 \times 1/2/(1-\epsilon) \times 1/3 \times 1/8 = 1/96/(1-\epsilon)$. This singularity can be regularized by subtracting the dipole (defined in Section 2.4) which, in terms of the Mandelstam variables, is given by

$$\mathcal{D}_{qqg} = -\frac{4}{\hat{t}} \frac{\hat{s}^2 + \hat{u}^2}{(\hat{s} + \hat{u})^2}$$
(3.61)

Hence we obtain

$$\frac{4\pi\alpha_s}{2}C_F \mathcal{D}_{qqg} \left|\mathcal{M}_{LO}\right|^2 = -\frac{1}{54} \left(\frac{\alpha_s^3}{\pi v^2}\right) \frac{\hat{s}^2 + \hat{u}^2}{\hat{t}} - \frac{1}{54} \left(\frac{\alpha_s^3}{\pi v^2}\right) \frac{(\hat{s}^2 + \hat{u}^2) \left[\hat{t} + 2\left(\hat{s} + \hat{u}\right)\right]}{(\hat{s} + \hat{u})^2}$$
(3.62)

where the lowest-order matrix element is given by

$$\left|\mathcal{M}_{LO}\right|^{2} = \frac{\alpha_{s}^{2} M_{H}^{4}}{576 \pi^{2} v^{2}} \frac{1}{(1-\epsilon)}$$
(3.63)

The subtracted cross section is

$$\sigma(qg \to qH) = \frac{1}{2\hat{s}} \int dP S_2 \left\{ \left| \mathcal{M}(qg \to qH) \right|^2 - \frac{4\pi\alpha_s}{2} C_F \mathcal{D}_{qqg} \left| \mathcal{M}_{LO} \right|^2 \right\}_{NS}$$
$$= \frac{1}{1728\hat{s}} \left(\frac{\alpha_s^3}{\pi^2 v^2} \right) \left(1 - \frac{M_H^2}{\hat{s}} \right) \left(3M_H^2 + \hat{s} + 4\hat{s} \ln\left(\frac{\hat{s}}{M_H^2}\right) \right)$$
(3.64)

The collinear singularity appearing in *m*-parton phase spaces will be absorbed into PDFs when we combine the integrated splitting function with the collinear counter term Eq. (2.14).

$$\int_{1} d\sigma_{ab}^{B}(xp_{a}, p_{b}) \left\{ \frac{4\pi\alpha_{s}}{2} C_{F} \mu^{2\epsilon} \int d\zeta_{p} v_{qqg}^{2} \right\} + \int_{1} d\sigma_{ab}^{C}(p_{a}, p_{b}, \mu_{F}^{2})$$

$$= \int_{0}^{1} dx \int_{1} d\sigma_{ab}^{B}(xp_{a}, p_{b}) \otimes \left[K(xp_{a}) + P(x, \mu_{F}^{2}) \right]$$
(3.65)

where

$$K(xp_{a}) = \frac{\alpha_{s}}{2\pi} \frac{1}{\Gamma(1-\epsilon)} \left\{ \left[\frac{x^{2}-2(1-x)}{x} - x \ln x \right] C_{F} + 2\ln(1-x)P_{gq}(x) - P_{gq}(x) \ln \left(\frac{4\pi\mu^{2}}{2xp_{a} \cdot p_{b}} \right) \right\}$$

$$P(x,\mu_{F}^{2}) = \frac{\alpha_{s}}{2\pi} \frac{1}{\Gamma(1-\epsilon)} P_{gq}(x) \ln \left(\frac{4\pi\mu^{2}}{\mu_{F}^{2}} \right)$$

$$P_{gq}(x) = C_{F} \frac{1+(1-x)^{2}}{x}, \qquad C_{F} = \frac{4}{3}$$
(3.66)

Here $P_{gq}(x)$ denotes the standard Altarelli-Parisi splitting function.

The collinear integrated splitting function is given in Section 2.5. It is worth mentioning that no soft singularity is present in the process $qg \rightarrow qH$. To this order, the appropriate scale at which to calculate α_s is not determined. We can take $\alpha_s = \alpha_s(\mu^2)$, where μ is an arbitrary renormalization scale. The hadronic cross section is independent of μ to $\mathcal{O}(\alpha_s^3)$.

Catani-Seymour dipoles

Using Catani-Seymour's scheme the dipole subtraction term is

$$\frac{4\pi\,\alpha_s}{2}\,C_F\,\mathcal{D}_{qqg}\,\left|\mathcal{M}_{LO}\right|^2 = -\frac{1}{54}\,\left(\frac{\alpha_s^3}{\pi\,v^2}\right)\,\frac{M_H^4 + 2\,\hat{s}\,(\hat{s} - M_H^2)}{\hat{t}} \tag{3.67}$$

and the subtracted cross section is given by

$$\sigma(qg \to qH) = \frac{1}{2\hat{s}} \int dP S_2 \left\{ |\mathcal{M}(qg \to qH)|^2 - \frac{4\pi\alpha_s}{2} C_F \mathcal{D}_{qqg} |\mathcal{M}_{LO}|^2 \right\}_{CS}$$
$$= -\frac{1}{576} \left(\frac{\alpha_s^3}{\pi^2 v^2}\right) \left(1 - \frac{M_H^2}{\hat{s}}\right)^2$$
(3.68)

The K and P terms are

$$K(xp_{a}) = \frac{\alpha_{s}}{2\pi} \frac{1}{\Gamma(1-\epsilon)} \left\{ C_{F}x + \left[2\ln(1-x) - \ln x\right] P_{gq}(x) - P_{gq}(x)\ln\left(\frac{4\pi\mu^{2}}{2xp_{a}\cdot p_{b}}\right) \right\}$$

$$P(x,\mu_{F}^{2}) = \frac{\alpha_{s}}{2\pi} \frac{1}{\Gamma(1-\epsilon)} P_{gq}(x)\ln\left(\frac{4\pi\mu^{2}}{\mu_{F}^{2}}\right)$$
(3.69)

3.3.3 The subprocess $gg \rightarrow gH$

To calculate the QCD corrections to the inclusive production of the Higgs boson from $gg \rightarrow H$, we also need the real contributions from $gg \rightarrow gH$. The matrix element for NLO real emission process in four dimensions is given by [56, 57]

$$\left|\mathcal{M}(gg \to gH)\right|^{2} = \frac{\alpha_{s}^{3}}{v^{2}} \frac{32}{3\pi} \frac{M_{H}^{8} + \hat{s}^{4} + \hat{t}^{4} + \hat{u}^{4}}{\hat{s}\,\hat{t}\,\hat{u}}$$
(3.70)

Collinear and soft singularities come from $\hat{t} \to 0$ or $\hat{u} \to 0$. The spin and colour averages yield an additional factor $1/2/(1-\epsilon) \times 1/2/(1-\epsilon) \times 1/8 \times 1/8 = 1/256/(1-\epsilon)^2$.

Nagy-Soper dipoles

The subtracted dipole term contains both t-channel and u-channel contributions. Their explicit expressions are defined in Section 2.4. We find

$$\mathcal{D}_{ggg} = \frac{8\,\hat{u}}{\hat{t}} \left\{ \frac{\hat{s}}{\left(\hat{t}+\hat{u}\right)^2} + \frac{1}{\hat{s}} \left[\frac{\left(\hat{s}+\hat{u}\right)^2 + \hat{s}^2}{\left(\hat{s}+\hat{u}\right)^2} - \frac{\hat{s}\,\hat{t}}{\left(\hat{s}+\hat{u}\right)\left(\hat{t}+\hat{u}\right)} \right] \right\} + \left(\hat{t}\leftrightarrow\hat{u}\right) + \frac{16\,\hat{s}}{\left(\hat{t}+\hat{u}\right)^2} \tag{3.71}$$

It is straightforward to integrate over the phase space to obtain the subtracted cross section

$$\sigma(gg \to gH) = \frac{1}{2\,\hat{s}} \int dPS_2 \left\{ |\mathcal{M}(gg \to gH)|^2 - \frac{4\,\pi\,\alpha_s}{2} C_A \mathcal{D}_{ggg} |\mathcal{M}_{LO}|^2 \right\}_{NS}$$

$$= \frac{1}{384\,\hat{s}} \left(\frac{\alpha_s^3}{\pi^2 \,v^2}\right) \left(1 - \frac{M_H^2}{\hat{s}}\right) \left\{\frac{4\,\hat{s}\,(2\,M_H^4 - 2\,M_H^2\,\hat{s} + \hat{s}^2)}{(M_H^2 - \hat{s})^2} \ln\left(\frac{\hat{s}}{M_H^2}\right) + \frac{M_H^4 + 34\,M_H^2\,\hat{s} + \hat{s}^2}{3\,\hat{s}} + \frac{4\,M_H^2\,\hat{s}}{M_H^2 - \hat{s}}\right\}$$
(3.72)

Next we have to consider the m-parton phase space contributions. All collinear and soft integrals can be looked up in Section 2.5. Using Eq. (2.16), we find

$$\int_{1} d\sigma_{ab}^{B}(xp_{a}, p_{b}) \left\{ \frac{4\pi\alpha_{s}}{2(1-\epsilon)} C_{A} \mu^{2\epsilon} \int d\zeta_{p} \left[v_{ggg}^{2} - v_{eik}^{2} \right] + 4\pi\alpha_{s} C_{A} \mu^{2\epsilon} \int d\zeta_{p} \Delta W_{ab} \right\}$$
$$+ \int_{1} d\sigma_{ab}^{B}(p_{a}, xp_{b}) \left\{ \frac{4\pi\alpha_{s}}{2(1-\epsilon)} C_{A} \mu^{2\epsilon} \int d\zeta_{p} \left[v_{ggg}^{2} - v_{eik}^{2} \right] + 4\pi\alpha_{s} C_{A} \mu^{2\epsilon} \int d\zeta_{p} \Delta W_{ab} \right\}$$
$$+ \int_{1} d\sigma_{ab}^{C}(p_{a}, p_{b}, \mu_{F}^{2})$$
$$= \int_{1} d\sigma_{ab}^{B}(p_{a}, p_{b}) \otimes I(\epsilon) + \frac{\alpha_{s}}{2\pi} \frac{1}{\Gamma(1-\epsilon)} \int_{1} d\sigma_{ab}^{B}(p_{a}, p_{b}) \left(\frac{4\pi\mu^{2}}{\mu_{F}^{2}} \right)^{\epsilon} \left(\frac{2}{\epsilon} \right) \left(\frac{11}{6} C_{A} - \frac{2}{3} n_{f} T_{R} \right)$$
$$+ \int_{0}^{1} dx \int_{1} d\sigma_{ab}^{B}(xp_{a}, p_{b}) \otimes \left[K^{a}(xp_{a}) + P(x, \mu_{F}^{2}) \right] + (a \leftrightarrow b)$$
(3.73)

Here, we always keep in mind that the factor $1/(1-\epsilon)$ is already included in the soft terms. The universal collinear counter terms are defined by Eq. (2.14). The standard Altarelli-Parisi splitting function $P_{gg}(x)$ is now

$$P_{gg}(x) = 2C_A \left(\frac{x}{(1-x)_+} + x(1-x) + \frac{1-x}{x}\right) + \delta(1-x) \left(\frac{11}{6}C_A - \frac{2}{3}n_f T_R\right), \quad C_A = 3$$
(3.74)

The corresponding I, K and P terms are given by

$$I(\epsilon) = \frac{\alpha_s}{2\pi} C_A \frac{1}{\Gamma(1-\epsilon)} \left(\frac{4\pi\mu^2}{Q^2}\right)^{\epsilon} \left(\frac{2}{\epsilon^2} - \frac{\pi^2}{3} + \mathcal{O}(\epsilon)\right)$$

$$K^a(xp_a) = \frac{\alpha_s}{2\pi} C_A \frac{1}{\Gamma(1-\epsilon)} \left\{ 4x \left(\frac{\ln(1-x)}{1-x}\right)_+ - 2x(1-x)\ln x + 4(1-x)\ln(1-x) \left(\frac{1+x^2}{x}\right)_+ - 2\left(x^2 - \frac{1-x}{x}\right) - 2\left(\frac{x}{(1-x)_+} + x(1-x) + \frac{1-x}{x}\right)\ln\left(\frac{4\pi\mu^2}{2xp_a \cdot p_b}\right) \right\}$$

$$P(x, \mu_F^2) = \frac{\alpha_s}{2\pi} C_A \frac{1}{\Gamma(1-\epsilon)} \left\{ 2\left(\frac{x}{(1-x)_+} + x(1-x) + \frac{1-x}{x}\right)\ln\left(\frac{4\pi\mu^2}{\mu_F^2}\right) \right\}$$
(3.75)

Catani-Seymour dipoles

Using Catani-Seymour's scheme the dipole subtraction term is

$$\frac{4\pi\alpha_s}{2}C_A\mathcal{D}_{ggg}\left|\mathcal{M}_{LO}\right|^2 = -\frac{1}{12}\left(\frac{\alpha_s^3}{\pi v^2}\right)\left(\frac{1}{\hat{t}} + \frac{1}{\hat{u}}\right)\left\{\hat{s}\left(\hat{s} - M_H^2\right) + \frac{\hat{s}M_H^4}{\hat{s} - M_H^2} + \frac{M_H^4\left(\hat{s} - M_H^2\right)}{\hat{s}}\right\}$$
(3.76)



Figure 3.20: Heavy quark limit for virtual corrections.

and the subtracted cross section is given by

$$\sigma(gg \to gH) = \frac{1}{2\,\hat{s}} \int dP S_2 \left\{ |\mathcal{M}(gg \to gH)|^2 - \frac{4\,\pi\,\alpha_s}{2} \,C_A \,\mathcal{D}_{ggg} \,|\mathcal{M}_{LO}|^2 \right\}_{CS}$$
$$= -\frac{11}{1152} \left(\frac{\alpha_s^3}{\pi^2 \,v^2}\right) \left(1 - \frac{M_H^2}{\hat{s}}\right)^3 \tag{3.77}$$

The I, K and P terms are

$$I(\epsilon) = \frac{\alpha_s}{2\pi} C_A \frac{1}{\Gamma(1-\epsilon)} \left(\frac{4\pi\mu^2}{Q^2} \right)^{\epsilon} \left(\frac{2}{\epsilon^2} - \frac{\pi^2}{3} + \mathcal{O}(\epsilon) \right)$$

$$K^a(xp_a) = \frac{\alpha_s}{2\pi} C_A \frac{1}{\Gamma(1-\epsilon)} \left\{ 4 \left(\frac{\ln(1-x)}{1-x} \right)_+ + 4 \left(-1 + x(1-x) + \frac{1-x}{x} \right) \ln(1-x) - 2 \left(\frac{x}{(1-x)_+} + x(1-x) + \frac{1-x}{x} \right) \ln \left(\frac{4\pi\mu^2}{2xp_a \cdot p_b} \right) - 2 \left(\frac{x}{(1-x)_+} + x(1-x) + \frac{1-x}{x} \right) \ln x \right\}$$

$$P(x, \mu_F^2) = \frac{\alpha_s}{2\pi} C_A \frac{1}{\Gamma(1-\epsilon)} \left\{ 2 \left(\frac{x}{(1-x)_+} + x(1-x) + \frac{1-x}{x} \right) \ln \left(\frac{4\pi\mu^2}{\mu_F^2} \right) \right\}$$
(3.78)

3.3.4 One-loop virtual corrections

Now we compute the one-loop virtual matrix element (Fig. 3.20) in the $\overline{\text{MS}}$ renormalization scheme. In the heavy quark limit $M_{\text{top}} \rightarrow \infty$ the Higgs boson couples to the trace of the energy momentum tensor [48–50, 58, 59]

$$\theta \equiv \theta^{\mu}{}_{\mu} = \partial^{\mu}s_{\mu} = \frac{\beta(g_s)}{2\,g_s} G^A_{\mu\nu} G^{\mu\nu}_A + (1+\delta) M_{\rm top} \,\bar{t} \,t$$
(3.79)

where s_{μ} is the scale current. The $(1 + \delta)$ term arises from the low energy theorem [53, 60–62]. Since the Higgs coupling to heavy fermions is $M_{\text{top}} (1 + H/v) \bar{t} t$, the counterterm for the Higgs Yukawa interaction is fixed in terms of the fermion mass and wave function renormalization. We have $\delta = 2 \alpha_s / \pi$. Hence, in the heavy quark limit $M_{\text{top}} \to \infty$ we have

$$\mathcal{L}_{\text{eff}} = \frac{H}{2v} \frac{\beta(g_s)}{g_s \left(1+\delta\right)} G^A_{\mu\nu} G^{\mu\nu}_A$$
(3.80)

Since the ggH coupling results from heavy quark loops, only processes with heavy fermions contribute to the β function. The heavy fermion contribution to the QCD β function is [63, 64]

$$\frac{\beta(g_s)}{g_s}\Big|_{\text{heavy fermions}} = N_H \frac{\alpha_s}{2\pi} \left(\frac{1}{3} + \frac{\alpha_s}{\pi} \frac{19}{12}\right)$$
(3.81)

Here, N_H denotes the number of heavy fermions. Therefore, to second order

$$\mathcal{L}_{\text{eff}} = \frac{\alpha_s}{12 \pi v} H G^A_{\mu\nu} G^{\mu\nu}_A \left(1 + \frac{11}{4} \frac{\alpha_s}{\pi}\right)$$
(3.82)

As a consequence of the non-abelian gauge invariance, the Lagrangian Eq. (3.82) can not only describe the Hgg coupling, but also the Hggg and Hgggg interactions (see Fig. 3.21). After a tedious calculation the one-loop virtual matrix element is given by [51,53]

$$\left|\mathcal{M}_{V}\right|^{2} = \left|\mathcal{M}_{LO}\right|^{2} \frac{\alpha_{s}}{2\pi} C_{A} \left(\frac{4\pi\mu^{2}}{M_{H}^{2}}\right)^{\epsilon} \Gamma(1+\epsilon) \left(-\frac{1}{\epsilon^{2}} + \frac{2}{3}\pi^{2} + \mathcal{O}(\epsilon)\right) \times 2$$
(3.83)

Hence, we obtain

$$\int_{1} d\sigma_{ab}^{B}(p_{a}, p_{b}) \otimes I(\epsilon) + \int d\sigma^{V} = \int_{1} d\sigma_{ab}^{B}(p_{a}, p_{b}) \frac{\alpha_{s}}{2\pi} C_{A}\left(\frac{2\pi^{2}}{3}\right)$$
(3.84)

The leftover $1/\epsilon$ pole can be regularized by performing charge renormalization. The charge counterterm in the $\overline{\text{MS}}$ renormalization scheme is (see *e.g.* [104, 106])

$$\sigma_{\rm ch} = (4 Z_g) \ \hat{\sigma}_0(gg \to H) \tag{3.85}$$

where

$$Z_g = -\frac{\alpha_s}{2\epsilon} \left(\frac{4\pi\mu^2}{\mu_F^2}\right)^{\epsilon} b_0 \Gamma(1+\epsilon) \left(\frac{\mu_F^2}{\mu^2}\right)^{\epsilon}$$

$$= -\left(\frac{11}{6}C_A - \frac{2}{3}n_f T_R\right) \frac{\alpha_s}{4\pi} \left(\frac{4\pi\mu^2}{\mu_F^2}\right)^{\epsilon} \Gamma(1+\epsilon) \left(\frac{1}{\epsilon} + \ln\frac{\mu_F^2}{\mu^2}\right),$$

$$b_0 = \frac{1}{2\pi} \left(\frac{11}{6}C_A - \frac{2}{3}n_f T_R\right)$$
(3.86)

and n_f is the number of light quarks. So we write for the charge renormalization cross section

$$\sigma_{\rm ch} = \frac{\alpha_s}{2\pi} \frac{1}{\Gamma(1-\epsilon)} \int_1 d\sigma_{ab}^B(p_a, p_b) \left(\frac{4\pi\mu^2}{\mu_F^2}\right)^{\epsilon} \left(-\frac{2}{\epsilon} - 2\ln\frac{\mu_F^2}{\mu^2}\right) \left(\frac{11}{6}C_A - \frac{2}{3}n_f T_R\right) \quad (3.87)$$

Now if we combine real contributions, virtual contributions, charge renormalization and collinear counter terms the physical cross section for $gg \rightarrow H$ is completely finite.

In summary, we see, that the pole structures in $I(\epsilon)$ terms are equivalent between Catani-Seymour's scheme and Nagy-Soper's scheme as they should. The finite terms in K and P are shifted around due to different momentum mapping. However the final numerical/analytical results are the same.

Result

In this subsection, we again summarize the results we obtain using Nagy-Soper dipoles from the different subprocesses of the Higgs production. The results shown here include one-particle phase space cross sections, two-particle phase space subtracted cross sections, collinear counterterms, virtual contributions, charge renormalization as well as the effective Lagrangian correction⁵.

$$\begin{aligned} \sigma(gg \to H) &= \sigma(q\bar{q} \to gH) \\ &+ \Delta\sigma(qg \to qH) + \sigma^{C}(qg \to qH) \\ &+ \Delta\sigma(gg \to gH) + \sigma^{C}(gg \to gH) + \sigma^{V} + \sigma_{ch} + \sigma_{eff} \end{aligned} \qquad \begin{array}{l} \text{quark induced} \\ \text{gluon induced} \end{aligned} \tag{3.88}$$

where

$$\begin{split} \Delta \sigma_{q\bar{q}} &= \sigma(q\bar{q} \to gH) = \frac{1}{486} \left(\frac{\alpha_s^3}{\pi^2 v^2}\right) (1-z)^3 \\ \Delta \sigma_{qg} &= \Delta \sigma(qg \to qH) + \sigma^C(qg \to qH) \\ &= \frac{\alpha_s}{\pi} \sigma_0 \left\{ -1 + 2z - \frac{1}{3}z^2 - \frac{1}{2}z P_{gq}(z) \left[\ln \frac{Q^2}{\hat{s}} - 2\ln(1-z) \right] \right\} \\ \Delta \sigma_{gg} &= \Delta \sigma(gg \to gH) + \sigma^C(gg \to gH) + \sigma^V + \sigma_{ch} + \sigma_{eff} \\ &= \frac{\alpha_s}{\pi} \sigma_0 \left\{ \delta(1-z) \left[\frac{11}{2} + \pi^2 + \left(\frac{11}{6} C_A - \frac{2}{3} n_f T_R \right) \ln \left(\frac{\mu^2}{\mu_F^2} \right) \right] \right\} \\ &+ \frac{\alpha_s}{\pi} \sigma_0 \left\{ 12 \left[\left(\frac{\ln(1-z)}{1-z} \right)_+ - z \left[2 - z \left(1 - z \right) \right] \ln(1-z) \right] \\ &- \frac{11}{2} \left(1 - z \right)^3 - z P_{gg}(z) \ln \frac{Q^2}{\hat{s}} \right\} \end{split}$$
(3.89)

⁵See talks given by F.J. Petriello at PSI Zuoz Summerschool 2010: *http://ltpth.web.psi.ch/zuoz2010/index.htm* and Fourth Graduate School in Physics at Colliders (Torino) 2009: *http://www.ph.unito.it/dft/scuola09/*

with

$$\sigma_{\rm eff} = \hat{\sigma}_0 \, \frac{\alpha_s}{\pi} \, \frac{11}{2} \,, \quad \hat{\sigma}_0 = \sigma_0 \, \delta(1-z) \,, \quad \sigma_0 = \frac{\alpha_s^2}{576 \, \pi \, v^2} \quad \text{and} \quad z = \frac{M_H^2}{\hat{s}}$$

Here, $\Delta\sigma(qg \to qH)$ and $\Delta\sigma(gg \to gH)$ in Eq. (3.88) and Eq. (3.89) mean the sum of oneparticle and two-particle phase space contributions. We find total agreement with the results in [51,53]. In the following sections we will show that the results of Nagy-Soper scheme, Catani-Seymour scheme and literature [51,53] for the process $gg \to H$ are identical.

3.3.5 Proof: Nagy-Soper scheme and Catani-Seymour scheme

In this section, we prove that the results of Nagy-Soper scheme and Catani-Seymour scheme for the process $gg \rightarrow H$ are equivalent. We only compare with real emission subtracted terms and K terms. The remaining I and P terms are exactly the same in both schemes.

The subprocess $gg \rightarrow gH$

The real emission subtracted cross section including PDFs for Catani-Seymour scheme is given by

$$R_{CS} = \int_{0}^{1} d\eta \int_{0}^{1} d\bar{\eta} \, g(\eta, Q^{2}) \, g(\bar{\eta}, Q^{2}) \, \theta(\eta \, \bar{\eta} \, s - M_{H}^{2}) \left[-\frac{11}{1152} \left(\frac{\alpha_{s}^{3}}{\pi^{2} \, v^{2}} \right) \left(1 - \frac{M_{H}^{2}}{\hat{s}} \right)^{3} \right]$$
$$= \int_{\tau_{H}}^{1} d\tau \, \frac{d\mathcal{L}^{gg}}{d\tau} \left[-\frac{11}{1152} \left(\frac{\alpha_{s}^{3}}{\pi^{2} \, v^{2}} \right) \left(1 - \frac{M_{H}^{2}}{\hat{s}} \right)^{3} \right], \quad \tau = \eta \, \bar{\eta}, \, \hat{s} = \tau \, s > M_{H}^{2}, \, \tau_{H} = \frac{M_{H}^{2}}{s}$$

where the differential parton luminosities are defined by

$$\frac{d\mathcal{L}^{gg}}{d\tau} = g \otimes g(\tau, Q^2)$$

$$\frac{d\mathcal{L}^{gq}}{d\tau} = g \otimes q(\tau, Q^2) + q \otimes g(\tau, Q^2)$$

$$\frac{d\mathcal{L}^{q\bar{q}}}{d\tau} = q \otimes \bar{q}(\tau, Q^2) + \bar{q} \otimes q(\tau, Q^2)$$
(3.90)

and the notation \otimes is given by

$$f \otimes g\left(x,\mu^2\right) = \int_x^1 \frac{dz}{z} f\left(z,\mu^2\right) g\left(\frac{x}{z},\mu^2\right)$$
(3.91)

The variables η and $\bar{\eta}$ denote momentum fractions of the partons. Furthermore, \hat{s} and s are the center-of-mass energies of partons and hadrons, respectively, while $q(\bar{q})$ and g in Eq. (3.90) are

PDFs of quark (antiquark) and gluon. The PDFs belong to the long-distance physics process of the scattering and hence belong to the non-perturbative part. But their evolutions follow the master equation of QCD: DGLAP equation, by which we can predict how PDFs evolve from one scale to another scale.

$$t\frac{d}{dt}q_a(x,t) = \frac{\alpha_s(t)}{2\pi} \int_x^1 \frac{d\xi}{\xi} P_{ab}\left(\frac{x}{\xi}, \alpha_s(t)\right) q_b(\xi,t)$$
(3.92)

where $t = \mu^2$ and P_{ab} is the splitting function. More generally, the DGLAP equation in $(2 n_f + 1)$ -dimension in the flavour space (flavour = quarks, antiquarks and gluons) is

$$t\frac{\partial}{\partial t}\left(\begin{array}{c}q_i(x,t)\\g(x,t)\end{array}\right) = \frac{\alpha_s(t)}{2\pi}\sum_{q_j,\bar{q}_j}\int_x^1\frac{d\xi}{\xi}\left(\begin{array}{c}P_{q_iq_j}\left(\frac{x}{\xi},\alpha_s(t)\right) & P_{q_ig}\left(\frac{x}{\xi},\alpha_s(t)\right)\\P_{gq_j}\left(\frac{x}{\xi},\alpha_s(t)\right) & P_{gg}\left(\frac{x}{\xi},\alpha_s(t)\right)\end{array}\right)\left(\begin{array}{c}q_j(\xi,t)\\g(\xi,t)\end{array}\right)$$
(3.93)

The real emission subtracted cross section including PDFs for Nagy-Soper scheme is given by

$$R_{NS} = \int_{\tau_H}^{1} d\tau \, \frac{d\mathcal{L}^{gg}}{d\tau} \left[\frac{1}{384\,\hat{s}} \left(\frac{\alpha_s^3}{\pi^2 \, v^2} \right) \left(1 - \frac{M_H^2}{\hat{s}} \right) \left\{ \frac{4\,\hat{s}\,(2\,M_H^4 - 2\,M_H^2\,\hat{s} + \hat{s}^2)}{(M_H^2 - \hat{s})^2} \ln\left(\frac{\hat{s}}{M_H^2}\right) + \frac{M_H^4 + 34\,M_H^2\,\hat{s} + \hat{s}^2}{3\,\hat{s}} + \frac{4\,M_H^2\,\hat{s}}{M_H^2 - \hat{s}} \right\} \right]$$
(3.94)

Define $\Delta K = K_{CS} - K_{NS}$ and we found

$$\Delta K^a = \frac{\alpha_s}{2\pi} C_A \left(-2\right) \left\{ \left(\frac{x}{1-x} + \frac{1-x}{x}\right) \ln x + \left(x^2 - \frac{1-x}{x}\right) \right\}$$

Convoluting former expression with PDFs and an additional one-dimensional integration yield

$$\int \Delta K^{a} + \int \Delta K^{b} = \int_{\tau_{H}}^{1} d\tau \, \frac{d\mathcal{L}^{gg}}{d\tau} \left[-\frac{6}{576} \left(\frac{\alpha_{s}^{3}}{\pi^{2} v^{2}} \right) \left(\frac{M_{H}^{2}}{\hat{s}} \right) \left\{ \left(\frac{M_{H}^{2}}{\hat{s}} \right)^{2} - \frac{\hat{s}}{M_{H}^{2}} \left(1 - \frac{M_{H}^{2}}{\hat{s}} \right) - \left(\frac{\hat{s}}{M_{H}^{2}} \left(1 - \frac{M_{H}^{2}}{\hat{s}} \right) + \frac{M_{H}^{2}}{\hat{s} \left(1 - M_{H}^{2}/\hat{s} \right)} \right) \ln \frac{\hat{s}}{M_{H}^{2}} \right\} \right]$$
(3.95)

Hence, it is straightforward to show that $\int \Delta K^a + \int \Delta K^b + R_{CS} - R_{NS} = 0$, where the symbolic notation $\int \Delta K$ simply means the convolution with PDFs and an additional one-dimensional integration.

The subprocess qg ightarrow qH

The real emission subtracted cross section including PDFs for Catani-Seymour scheme is given by

$$R_{CS} = \sum_{q,\bar{q}} \int_{\tau_H}^1 d\tau \, \frac{d\mathcal{L}^{gq}}{d\tau} \left[-\frac{1}{576} \left(\frac{\alpha_s^3}{\pi^2 v^2} \right) \left(1 - \frac{M_H^2}{\hat{s}} \right)^2 \right]$$
(3.96)

The real emission subtracted cross section including PDFs for Nagy-Soper scheme is given by

$$R_{NS} = \sum_{q,\bar{q}} \int_{\tau_H}^1 d\tau \, \frac{d\mathcal{L}^{gq}}{d\tau} \left[\frac{1}{1728\,\hat{s}} \left(\frac{\alpha_s^3}{\pi^2 \, v^2} \right) \left(1 - \frac{M_H^2}{\hat{s}} \right) \left(3 \, M_H^2 + \hat{s} + 4 \, \hat{s} \ln \left(\frac{\hat{s}}{M_H^2} \right) \right) \right]$$
(3.97)

From the definition of $\Delta K = K_{CS} - K_{NS}$ it follows that

$$\Delta K = \frac{\alpha_s}{2\pi} C_F \, \frac{2\left(1-x\right)}{x} \left(1-\ln x\right)$$

Convolution with PDFs and an additional one-dimensional integration yield

$$\int \Delta K = \sum_{q,\bar{q}} \int_{\tau_H}^1 d\tau \, \frac{d\mathcal{L}^{gq}}{d\tau} \left[\frac{1}{576} \frac{4}{3} \left(\frac{\alpha_s^3}{\pi^2 v^2} \right) \left(1 - \frac{M_H^2}{\hat{s}} \right) \left(1 + \ln \frac{\hat{s}}{M_H^2} \right) \right]$$
(3.98)

Hence, it is straightforward to show that $\int \Delta K + R_{CS} - R_{NS} = 0$.

3.3.6 **Proof: Catani-Seymour scheme and literature results**

In this section, we show that the results of Catani-Seymour scheme for the process $gg \rightarrow H$ are identical to literature results [51,53] and hence we complete our proofs that CS, NS and literature results agree with each other. Here, we only compare with the real emission part, K term and P term; terms proportional to δ function are straightforward and hence we will not list the proofs.

The subprocess $gg \rightarrow gH$

In the limit that the top quark mass is infinitely large, the parton level cross section is given by⁶

$$\hat{\sigma}_0(gg \to H) = \frac{\alpha_s^2}{\pi} \frac{M_H^2}{576 v^2} \,\delta\left(\hat{s} - M_H^2\right) = \frac{\alpha_s^2}{576 \pi v^2} \,\delta\left(1 - z\right) = \sigma_0 \,\delta\left(1 - z\right) \tag{3.99}$$

with

$$\sigma_0 = \frac{\alpha_s^2}{576 \pi v^2} \quad \text{and} \quad z = \frac{M_H^2}{\hat{s}}$$
 (3.100)

The real emission subtracted cross section is then

$$\sigma(gg \to gH) = -\frac{11}{1152} \left(\frac{\alpha_s^3}{\pi^2 v^2}\right) (1-z)^3 = \frac{\alpha_s}{\pi} \sigma_0 \left[-\frac{11}{2} (1-z)^3\right]$$
(3.101)

⁶Here we put a hat on cross section σ_0 in order to indicate that we only work on parton level.

Next we consider the K term and the P term contributions. They can be written as

$$\int_{0}^{1} dx \int d\sigma^{B}(x) \otimes [K(x) + P(x)] \times 2 = \int_{0}^{1} dx \left[\sigma_{0} M_{H}^{2} \delta \left(x \hat{s} - M_{H}^{2} \right) \right] \otimes [K(x) + P(x)] \times 2$$

= $\sigma_{0} \frac{M_{H}^{2}}{\hat{s}} \left[K(x) + P(x) \right] \times 2$, with $x = \frac{M_{H}^{2}}{\hat{s}}$
= $\frac{\alpha_{s}}{\pi} \sigma_{0} \left\{ 12 \left[x \left(\frac{\ln(1-x)}{1-x} \right)_{+} - x \left[2 - x \left(1 - x \right) \right] \ln(1-x) + \ln(1-x) \right] - x P_{gg}(x) \ln \frac{Q^{2}}{\hat{s}} \right\}$
(3.102)

where we have added an extra term

$$0 = -\delta(1-x) \left(\frac{11}{6}C_A - \frac{2}{3}n_f T_R\right) \ln x$$
(3.103)

in Eq. (3.102) and we have also used the plus prescription in such a way that

$$x\left(\frac{\ln(1-x)}{1-x}\right)_{+} + \ln(1-x) \quad \rightarrow \quad \left(\frac{\ln(1-x)}{1-x}\right)_{+} \tag{3.104}$$

Here, the test function is simply the LO cross section. Combining Eq. (3.101) and Eq. (3.102), we obtain Eq. (10) of [53].

The subprocess qg ightarrow qH

The real emission subtracted cross section is

$$\sigma(qg \to qH) = \frac{\alpha_s}{\pi} \sigma_0 \left(-1 + 2z - z^2\right)$$
(3.105)

The K term and the P term contributions are given by

$$\int_{0}^{1} dx \int d\sigma^{B}(x) \otimes \left[K(x) + P(x)\right] = \frac{\alpha_{s}}{\pi} \sigma_{0} \left\{ \frac{2}{3} z^{2} - \frac{1}{2} z P_{gq}(z) \left[\ln \frac{Q^{2}}{\hat{s}} - 2 \ln(1-z) \right] \right\}$$
(3.106)

Here, $x = z = \frac{M_H^2}{\hat{s}}$. Combining Eq. (3.105) and Eq. (3.106), we obtain Eq. (11) of [53].

3.4 Higgs decay: $H \rightarrow gg$

A closely related problem to the process $gg \rightarrow H$ at NLO is the QCD radiative corrections to the gluonic decay modes of the Higgs boson. Two different emissions appear at NLO: either the emission of an additional gluon or the splitting of one gluon into a $q\bar{q}$ pair.

$$H \to g g g, \qquad H \to g q \bar{q}$$
 (3.107)

for which we can test the final pieces of splitting processes in the final state using Nagy-Soper dipoles: $g \rightarrow gg$ and $g \rightarrow q\bar{q}$. The lowest-order, real emissions and virtual diagrams are shown in Fig. 3.21 and Fig. 3.22. The NLO QCD calculations have already been available in the literatures for some time [53–55].

3.4.1 The subprocess $H ightarrow gqar{q}$

We consider the heavy quark limit $M_{top} \gg M_H$. Using again the effective Lagrangian shown in Eq. (3.54), the lowest-order matrix element for $H(Q) \rightarrow g(p_1)g(p_2)$, which includes a symmetry factor 1/2! for identical gluons in the final state, is given by

$$\left|\mathcal{M}_{LO}\right|^{2} = \frac{2}{9} \frac{\alpha_{s}^{2} M_{H}^{4}}{\pi^{2} v^{2}} (1-\epsilon)$$
(3.108)

Here, we calculate in $d = 4 - 2 \epsilon$ dimensions. The matrix element for $H(Q) \rightarrow g(p_1)q(p_2)\bar{q}(p_3)$ is

$$|\mathcal{M}(H \to gq\bar{q})|^{2} = \frac{16}{9} \frac{\alpha_{s}^{3}}{\pi v^{2}} \frac{(p_{1} + p_{2})^{4} + (p_{1} + p_{3})^{4}}{(p_{2} + p_{3})^{2}}$$
$$= \frac{16}{9} \frac{\alpha_{s}^{3}}{\pi v^{2}} \hat{Q}^{2} \frac{(x_{1} + x_{2} - 1)^{2} + (1 - x_{2})^{2}}{(1 - x_{1})}$$
(3.109)

where x_1 , x_2 and x_3 are defined by

$$\begin{aligned}
x_1 &= \frac{2\hat{p}_k \cdot \hat{Q}}{\hat{Q}^2}, \qquad \hat{p}_k \to p_1, \qquad \hat{Q}^2 = Q^2 = M_H^2 \\
x_2 &= \frac{2\hat{p}_\ell \cdot \hat{Q}}{\hat{Q}^2}, \qquad \hat{p}_\ell \to p_2 \\
x_3 &= \frac{2\hat{p}_j \cdot \hat{Q}}{\hat{Q}^2}, \qquad \hat{p}_j \to p_3
\end{aligned} (3.110)$$

The collinear singularity arises from $x_1 \to 1$. Introducing the dipole $\mathcal{D}_{gq\bar{q}}$, defined in Section 2.4, we find

$$\frac{4\pi\alpha_s}{2} T_R \mathcal{D}_{gq\bar{q}} \left| \mathcal{M}_{LO} \right|^2 \times 2$$

$$= \frac{4\pi\alpha_s}{2} T_R \left| \mathcal{M}_{LO} \right|^2 \frac{2}{(\hat{p}_\ell \cdot \hat{p}_j)^2} \left\{ \frac{\hat{p}_\ell \cdot Q \, \hat{p}_j \cdot p_\ell + \hat{p}_j \cdot Q \, \hat{p}_\ell \cdot p_\ell}{p_\ell \cdot Q} - \frac{Q^2 \, \hat{p}_\ell \cdot p_\ell \, \hat{p}_j \cdot p_\ell}{(p_\ell \cdot Q)^2} \right\} \times 2$$

$$= \frac{8}{9} \frac{\alpha_s^3}{\pi v^2} \, \hat{Q}^2 \frac{1}{1-x_1} \left\{ \frac{x_2 \left(x_1 + x_2 - 1\right) + \left(2 - x_1 - x_2\right) \left(1 - x_2\right)}{x_1} - 2 \, \frac{\left(1 - x_1\right) \left(1 - x_2\right) \left(x_1 + x_2 - 1\right)}{x_1^2} \right\} + (x_2 \leftrightarrow x_3) \tag{3.111}$$

Here, the dot products $\hat{p}_j \cdot p_\ell$ and $\hat{p}_\ell \cdot p_\ell$ are related to λ, x, y and a_ℓ via

$$p_{\ell} \cdot \hat{p}_{j} = \frac{1}{\lambda} \left(y - \frac{1 - \lambda + y}{2 a_{\ell}} x \left(1 + y \right) \right) p_{\ell} \cdot Q$$

$$p_{\ell} \cdot \hat{p}_{\ell} = \frac{1}{\lambda} \left(y - \frac{1 - \lambda + y}{2 a_{\ell}} \left(1 - x \right) \left(1 + y \right) \right) p_{\ell} \cdot Q$$
(3.112)

where λ , x, y and a_{ℓ} are defined in Section 2.3 and Section 2.4. It is worth mentioning that the second term in Eq. (3.111) is exactly the same as the first term, so we can just simply multiply by a factor of 2 in the end. Integrating over the three particle phase space and summing over final state quark flavours yields the subtracted decay rate

$$d\Gamma(H \to gq\bar{q}) = \frac{1}{2M_H} \int dPS_3 \sum_q \left(\left| \mathcal{M}(H \to gq\bar{q}) \right|^2 - \frac{4\pi\alpha_s}{2} T_R \mathcal{D}_{gq\bar{q}} \left| \mathcal{M}_{LO} \right|^2 \times 2 \right)$$
$$= \frac{G_F M_H^3 \alpha_s^2}{36\sqrt{2}\pi^3} \left(-\frac{5}{18} n_f \right) \frac{\alpha_s}{\pi}$$
(3.113)

Next we use the results of the integrated splitting functions for the *m*-parton phase space contributions (Section 2.5), which in the case of $gq\bar{q}$ splitting only involves collinear integral. Hence, the integrated $gq\bar{q}$ splitting function, which sums over final state quark flavours, is given by

$$\frac{4\pi\alpha_s}{2(1-\epsilon)}T_R\,\mu^{2\epsilon}\sum_q \int d\zeta_p v_{gq\bar{q}}^2 \times 2 = \frac{\alpha_s}{2\pi}T_R\,\left(\frac{4\pi\mu^2}{2p_\ell\cdot\hat{Q}}\right)^\epsilon \frac{1}{\Gamma(1-\epsilon)}n_f\,\left(-\frac{2}{3\epsilon}-\frac{16}{9}\right) \times 2 \quad .$$
(3.114)

3.4.2 The subprocess $H \rightarrow ggg$

The matrix element for $H(Q) \rightarrow g(p_1)g(p_2)g(p_3)$, which includes a symmetry factor 1/3! for identical gluons in the final state, is given by

$$\begin{aligned} \left|\mathcal{M}(H \to ggg)\right|^{2} \\ &= \frac{\alpha_{s}^{3}}{\pi v^{2}} \frac{32}{3} \frac{1}{3!} \frac{\left[(p_{1} + p_{2})^{2} + (p_{1} + p_{3})^{2} + (p_{2} + p_{3})^{2}\right]^{4} + (p_{1} + p_{2})^{8} + (p_{1} + p_{3})^{8} + (p_{2} + p_{3})^{8}}{(p_{1} + p_{2})^{2}(p_{1} + p_{3})^{2}(p_{2} + p_{3})^{2}} \\ &= \frac{\alpha_{s}^{3}}{\pi v^{2}} \frac{32}{3} \frac{1}{3!} \left\{ \frac{2s_{123}^{2}s_{12}}{s_{13}s_{23}} + \frac{2s_{123}^{2}s_{13}}{s_{12}s_{23}} + \frac{2s_{123}^{2}s_{23}}{s_{12}s_{13}} + \frac{2s_{12}s_{13}}{s_{23}} + \frac{2s_{12}s_{23}}{s_{13}} + \frac{2s_{12}s_{23}}{s_{12}} + \frac{2s_{12}s_{23}}{s_{13}} + \frac{2s_{12}s_{23}}{s_{12}} + \frac{2s_{12}s_{23}}{s_{12}} + \frac{2s_{12}s_{23}}{s_{13}} + \frac{2s_{12}s_{23}}{s_{13}} + \frac{2s_{12}s_{23}}{s_{13}} + \frac{2s_{12}s_{23}}{s_{13}} + \frac{2s_{12}s_{23}}{s_{12}} + \frac{2s_{12}s_{23}}{s_{12}} + \frac{2s_{12}s_{23}}{s_{12}} + \frac{2s_{12}s_{23}}{s_{12}} + \frac{2s_{12}s_{23}}{s_{12}} + \frac{2s_{12}s_{23}}{s_{13}} + \frac{2s_{12}s_{23}}{s_{12}} + \frac{2s_{12}s_{23}}{s_{$$

where the first three terms correspond to soft singularities (interference terms) and the next three terms correspond to collinear singularities. Since there are three identical gluons in the final



Figure 3.21: Decays of the Higgs boson: LO diagrams and $H \rightarrow ggg$, $H \rightarrow gq\bar{q}$ diagrams.



Figure 3.22: Decays of the Higgs boson $H \rightarrow gg$: virtual diagrams.

state, the labelling of the gluon, denoted with \hat{p}_j (j = 1, 2, 3), is arbitrary. So we have to take all combinations into account:

A:	$\hat{p}_j = p_1 \rightarrow \text{emitted gluon}$	$p_2/p_3 \rightarrow \text{emitter/spectator}$ (or	spectator/emitter)
B:	$\hat{p}_j = p_2 \rightarrow \text{emitted gluon}$	$p_1/p_3 \rightarrow \text{emitter/spectator}$ (or	spectator/emitter)
C:	$\hat{p}_j = p_3 \rightarrow \text{emitted gluon}$	$p_1/p_2 \rightarrow \text{emitter/spectator}$ (or	spectator/emitter)

so the dipoles corresponding to soft part can be written as three different configurations A, B, C

$$\mathcal{D}_{s} = \frac{4\pi\alpha_{s}}{2}C_{A} |\mathcal{M}_{LO}|^{2} \frac{1}{3} \left(v_{s}^{2}(A) + v_{s}^{2}(B) + v_{s}^{2}(C)\right) \times 2$$

$$= \frac{\alpha_{s}^{3}}{\pi v^{2}} \frac{32}{3} \frac{1}{3!} \left\{ \frac{2s_{123}^{2}s_{12}}{s_{13}s_{23}} + \frac{2s_{123}^{2}s_{13}}{s_{12}s_{23}} + \frac{2s_{123}^{2}s_{23}}{s_{12}s_{13}} \right\}$$
(3.116)

where

$$v_s^2 = v_{\text{eik}}^2 - v_{\text{soft}}^2 = \frac{4\,\hat{p}_\ell \cdot \hat{p}_k\,\hat{p}_\ell \cdot \hat{Q}}{\left(\hat{p}_j \cdot \hat{p}_k\,\hat{p}_\ell \cdot \hat{Q} + \hat{p}_\ell \cdot \hat{p}_j\,\hat{p}_k \cdot \hat{Q}\right)\,\hat{p}_\ell \cdot \hat{p}_j}, \quad (\ell, j, k = 1, 2, 3) (3.117)$$

and factor of 2 in Eq. (3.116) is present due to the fact that emitters and spectators are interchangeable. The dipole corresponding to collinear part is

$$\mathcal{D}_{c} = \frac{4\pi\alpha_{s}}{2}C_{A}\left|\mathcal{M}_{LO}\right|^{2}\frac{1}{3}\left(v_{c}^{2}(A) + v_{c}^{2}(B) + v_{c}^{2}(C)\right) \times 2$$

$$= \frac{4}{3}\frac{\alpha_{s}^{3}}{\pi v^{2}}\hat{Q}^{4}\frac{1}{3}\left\{\frac{4s_{12}s_{13}}{s_{23}(s_{12} + s_{13})^{2}}\left[2 - \frac{4s_{23}\hat{Q}^{2}}{(s_{12} + s_{23})(s_{13} + s_{23})} + \left[\frac{2s_{23}\hat{Q}^{2}}{(s_{12} + s_{23})(s_{13} + s_{23})}\right]^{2}\right]$$

$$+ \frac{4s_{12}s_{23}}{s_{13}(s_{12} + s_{23})^{2}}\left[2 - \frac{4s_{13}\hat{Q}^{2}}{(s_{12} + s_{13})(s_{13} + s_{23})} + \left[\frac{2s_{13}\hat{Q}^{2}}{(s_{12} + s_{13})(s_{13} + s_{23})}\right]^{2}\right]$$

$$+ \frac{4s_{13}s_{23}}{s_{12}(s_{13} + s_{23})^{2}}\left[2 - \frac{4s_{12}\hat{Q}^{2}}{(s_{12} + s_{23})(s_{12} + s_{13})} + \left[\frac{2s_{12}\hat{Q}^{2}}{(s_{12} + s_{23})(s_{12} + s_{13})}\right]^{2}\right]\right\} \quad (3.118)$$

where

$$v_{c}^{2} = \tilde{v}_{ggg}^{2} - v_{eik}^{2}$$

$$= \frac{(\hat{p}_{j} \cdot Q - \hat{p}_{j} \cdot p_{\ell}) (\hat{p}_{\ell} \cdot Q - \hat{p}_{\ell} \cdot p_{\ell})}{(p_{\ell} \cdot Q)^{2} \hat{p}_{\ell} \cdot \hat{p}_{j}} \left\{ 2 - 2 \frac{\hat{p}_{j} \cdot \hat{p}_{\ell} Q^{2}}{\hat{p}_{\ell} \cdot Q \hat{p}_{j} \cdot Q} + \left(\frac{\hat{p}_{j} \cdot \hat{p}_{\ell} Q^{2}}{\hat{p}_{\ell} \cdot Q \hat{p}_{j} \cdot Q} \right)^{2} \right\}$$

$$(\ell, j, k = 1, 2, 3)$$
(3.119)

Here, factor of 2 is present for the same reason as explained above. Integrating over the three particle phase space yields the subtracted decay rate

$$d\Gamma(H \to ggg) = \frac{1}{2M_H} \int dP S_3 \left(|\mathcal{M}(H \to ggg)|^2 - \mathcal{D}_c - \mathcal{D}_s \right) = \frac{G_F M_H^3 \alpha_s^2}{36\sqrt{2}\pi^3} \left(\frac{-214 + 27\pi^2}{24} \right) \frac{\alpha_s}{\pi}$$
(3.120)

Next we have to consider the m-parton phase space contributions, the integrated splitting functions in the case of ggg splitting involve both the collinear and soft integrals (see Section 2.5) which are given by

$$\left(\frac{4\pi\alpha_s}{2(1-\epsilon)}C_A\mu^{2\epsilon}\int d\zeta_p \left[\tilde{v}_{ggg}^2 - v_{eik}^2\right] + 4\pi\alpha_s C_A\mu^{2\epsilon}\int d\zeta_p\,\Delta W\right) \times 2$$
$$= \frac{\alpha_s}{2\pi}C_A \left(\frac{4\pi\mu^2}{2p_\ell\cdot\hat{Q}}\right)^\epsilon \frac{1}{\Gamma(1-\epsilon)} \left(\frac{2}{\epsilon^2} + \frac{11}{3\epsilon} + \frac{163}{9} - \frac{7}{4}\pi^2\right)$$
(3.121)

Here, the $1/(1-\epsilon)$ is already included in the soft term. The virtual matrix element for $H \to gg$ in the $\overline{\text{MS}}$ scheme is given in Eq. (3.83). Now, if we combine the real emission contributions $H \to gq\bar{q}, H \to ggg$ and virtual contribution, the $1/\epsilon^2$ pole is cancelled

$$d\Gamma(H \to gg(g) + gq\bar{q}) = \frac{1}{2M_H} \int dP S_2 \left\{ |\mathcal{M}_V|^2 + \sum_q \int d\zeta_p \,\mathcal{D}(gq\bar{q}) \, |\mathcal{M}_{LO}|^2 + \int d\zeta_p \,\mathcal{D}(ggg) \, |\mathcal{M}_{LO}|^2 \right\} + d\Gamma(H \to ggg) + d\Gamma(H \to gq\bar{q}) = \Gamma_{LO} \left(\frac{4\pi\mu^2}{\mu_F^2}\right)^{\epsilon} \frac{1}{\Gamma(1-\epsilon)} \left(\frac{\alpha_s}{2\pi}\right) \,\beta_0 \,\frac{2}{\epsilon} + \Gamma_{LO} \left(\frac{73}{4} - \frac{7}{6}n_f\right) \left(\frac{\alpha_s}{\pi}\right)$$
(3.122)

and the leftover $1/\epsilon$ pole has to be renormalized by performing charge renormalization. Here, $\mathcal{D}(gq\bar{q})$ and $\mathcal{D}(ggg)$ are symbolic notations for splitting functions with appropriate prefactor neglected, and their explicit expressions are given in Eq. (3.114) and Eq. (3.121). The charge counterterm in the $\overline{\text{MS}}$ renormalization scheme is

$$\Gamma_{\rm ch} = (4 Z_g) \ \Gamma_{LO}(H \to gg) \tag{3.123}$$

where

$$Z_g = -\frac{\alpha_s}{2\epsilon} (4\pi)^{\epsilon} b_0 \Gamma(1+\epsilon), \quad b_0 = \frac{1}{2\pi} \left(\frac{11}{6}C_A - \frac{2}{3}n_f T_R\right) = \frac{1}{2\pi}\beta_0$$
(3.124)

and n_f is the number of light quarks. Hence

$$\Gamma_{\rm ch} = -\Gamma_{LO} \frac{1}{\Gamma(1-\epsilon)} \left(\frac{4\pi\mu^2}{\mu_F^2}\right)^{\epsilon} \left(\frac{\alpha_s}{\pi}\right) \beta_0 \frac{1}{\epsilon} - \Gamma_{LO} \left(\frac{\alpha_s}{\pi}\right) \beta_0 \ln \frac{\mu_F^2}{\mu^2}$$
(3.125)

If we treat everything equivalently in the effective theory, we need to add the effective Lagrangian correction term, which is given by⁷

$$\Gamma_{LO} \frac{\alpha_s}{\pi} \frac{11}{2} \tag{3.126}$$

Result

Combining all contributions to the total decay rate for $H \to gg$ at NLO, the result is completely finite

$$\Gamma_{LO} + \Gamma(H \to gg(g) + gq\bar{q}) + \Gamma_{ch}$$

$$= \Gamma_{LO} \left[1 + \left[\left(\frac{95}{4} - \frac{7}{6} n_f \right) + \left(\frac{11}{6} C_A - \frac{2}{3} n_f T_R \right) \ln \frac{\mu^2}{\mu_F^2} \right] \frac{\alpha_s \left(M_H^2 \right)}{\pi} \right] \quad (3.127)$$

where

$$\Gamma_{LO}(H \to gg) = \frac{G_F M_H^3 \alpha_s^2}{36 \sqrt{2} \pi^3}$$
(3.128)

The result shown here is in agreement with [53].

⁷See talks given by F.J. Petriello at PSI Zuoz Summerschool 2010: *http://ltpth.web.psi.ch/zuoz2010/index.htm* and Fourth Graduate School in Physics at Colliders (Torino) 2009: *http://www.ph.unito.it/dft/scuola09/*

Chapter 4

Conclusions

In this thesis we have proposed an alternative subtraction method at NLO QCD calculations. The traditional way of regularizing the infrared singularities in both one-loop diagrams and real radiation corrections in the context of dimensional regularization scheme is straightforward for the analytic cancellation of the respective divergences. However, numerical implementations for multi-particle processes prove to be challenging.

The subtraction method provides a way of achieving this. The goal of subtraction formalism is to extract infrared divergences from real radiation matrix elements in order to combine them with the one-loop virtual contributions. The key observation for the subtraction scheme is that a (m + 1)-parton matrix element can be factorized into a *m*-parton matrix element multiplied by the generalised splitting functions (which contain the singularity structure of the (m + 1)-parton matrix element) in the soft and collinear limits based on the factorization theorems (see Eq. (4.1)). An important message is that the splitting functions are universal and process independent; this means we calculate them once and for all. The symbolic expression is

$$\mathcal{M}_{m+1}(\{\hat{p}\}_{m+1}) \approx v_{\ell}(\{\hat{p}\}_{m+1}) \cdot \mathcal{M}_m(\{p\}_m) .$$
(4.1)

In this new scheme, dipoles ($\approx |v_\ell|^2$) are based on the momentum mappings and on the splitting functions derived from an improved parton shower formulation with quantum interference [28]. Momentum mappings must guarantee total energy momentum conservation as well as the on-shell condition for all external partons both before and after the mappings. One important feature of our scheme is that we use a global momentum mapping in which the mapping takes all the partons into account at once when going from (m + 1)- to m-particle phase space, instead of separately summing over all possible emitter/spectator pairs like Catani-Seymour scheme. As a result, the number of dipole terms is greatly smaller than the standard subtraction schemes. Another essential point of our formalism is that we split the collinear and soft (based on the eikonal approximation) splitting functions according to Eq. (2.160) in such a way that the two
terms $\overline{W}_{\ell\ell} - \overline{W}_{\ell\ell}^{\text{eikonal}}$ and $\overline{W}_{\ell\ell}^{\text{eikonal}} - \overline{W}_{\ell k}$ are positive and hence we can use these splitting functions as dipole subtraction terms to construct a parton shower Monte Carlo program. The use of the shower splitting functions as dipoles also significantly facilitates the matching of NLO calculations with the corresponding parton shower.

We have also achieved the analytical integrations of the splitting functions over an unresolved one parton integration measure, obtaining the correct soft and collinear singularities in ϵ that are necessary to cancel the soft divergences in the virtual diagrams.

To establish our formalism we have investigated some simple processes at colliders with up to two massive/massless particles in the final state. We have presented all subtraction terms and their integrated splitting functions and have applied our scheme to a variety of well-known processes at NLO, showing that the singular behaviours of the shower splitting functions indeed match the behaviours of real radiation matrix elements and one-loop contributions in the soft and collinear limits. In more detail¹, we have investigated single W production at hadron colliders (initial-state qq(x)g and $gq\bar{q}$ collinear splittings, interference between initial states), dijet production at lepton colliders (final-state qqg collinear splittings, interference between final states), Higgs production at hadron colliders (initial-state qqg collinear splittings, interference between initial states), Higgs decay to two gluons (final-state $gq\bar{q}$ and ggg collinear splittings, interference between initial states). The discussions about interference between initial and final states or interference between final and initial states can be found in [87] in which we have used deep-inelastic scattering (DIS) process. In all cases, we have reproduced the results from the literature and have shown that our implementation agrees with results obtained using the Catani-Seymour scheme.

In this thesis, we have demonstrated that the global momentum mappings combined with the shower splitting functions as dipoles indeed can be used as the subtraction terms for some simple processes; the advantages of the two features will become apparent when applying to more involved multi-parton processes at NLO or matching the NLO calculations with the corresponding improved parton shower ². Due to the different momentum mapping prescription, our scheme leads to more complicated finite parts of the integrated splitting functions when considering processes with three or more final-state partons, and an example of the general case ($a_{\ell} \neq 1$) for the final-state splitting function $g \rightarrow q \bar{q}$ has been presented in [86]. Nevertheless all these finite parts can be integrated numerically in the respective Monte Carlo program. A generic application to a more non-trivial scattering process is still work in progress. However, we are confident that we will obtain some interesting results for multi-parton final states using our new scheme.

¹*c.f.* Table 4.1

²We note that the work [89] is to implement the new scheme into the Helac Event Generator [90].

Splitting function	Process
Initial state: qq(x)g $gq\bar{q}$ qqg(x) ggg interference between initial states interference between initial and final states	single W production, DIS single W production Higgs production Higgs production single W production, Higgs production DIS
Final state: qqg $gq\bar{q}$ ggg interference between final states interference between final and initial states	Dijet production, DIS Higgs decay Higgs decay Dijet production, Higgs decay DIS

Table 4.1: List of all splitting functions presented in Chapter 2 and test processes used for the scheme validation in Chapter 3 and [87].

Chapter 5

Appendix

5.1 Useful mathematical formulae

Here are some formulae that I find useful from time to time.

5.1.1 Gamma function, Beta function and Hypergeometric function

Gamma function

We make frequent use of the Euler Γ -function, which can be defined by the convergent integral

$$\Gamma(z) = \int_0^\infty dt \, t^{z-1} \, e^{-t}, \qquad \mathcal{R}e\{z\} > 0 \tag{5.1}$$

Integration by parts can confirm the identity

$$\Gamma(1+z) = z \,\Gamma(z) \tag{5.2}$$

Hence, for positive, integer values of z, we have

$$\Gamma(z) = (z-1)! \tag{5.3}$$

Eq. (5.2) can also be used to shift the argument and define the Γ -function when $\mathcal{R}e\{z\} < 0$. This shows that there are simple poles at $z = 0, -1, -2, \cdots$. The following expansion is also useful

$$\Gamma(1+\epsilon) = 1 - \gamma_E \epsilon + \left(\frac{\pi^2}{12} + \frac{1}{2}\gamma_E^2\right)\epsilon^2 + \mathcal{O}(\epsilon^3)$$
(5.4)

where $\gamma_E = 0.577\,215\,664\,901\,53\cdots$ is the Euler-Mascheroni constant.

Beta function

We will also sometimes use the β -function integral, which is defined by

$$B(1+m,1+n) = \int_0^1 dx \, x^m \, (1-x)^n = \frac{\Gamma(1+m) \, \Gamma(1+n)}{\Gamma(2+m+n)}, \qquad \mathcal{R}e\{m,n\} > -1 \quad (5.5)$$

Hypergeometric function $_{p}F_{q}(a_{1}, \cdots, a_{p}; b_{1}, \cdots, b_{q}; x)$

Next we list a couple of useful relations for the Hypergeometric functions; further details can be found in [119, 120].

$${}_{p}F_{q}(a_{1},\cdots,a_{k-1},a_{k},a_{k+1},\cdots,a_{p};b_{1},\cdots,b_{m-1},a_{k},b_{m+1},\cdots,b_{q};x)$$

$$={}_{p-1}F_{q-1}(a_{1},\cdots,a_{k-1},a_{k+1},\cdots,a_{p};b_{1},\cdots,b_{m-1},b_{m+1},\cdots,b_{q};x)$$
(5.6)

$$\int_{0}^{1} dx \, x^{m-1} \, (1-x)^{n-1} \, _{p}F_{q}(a_{1}, \cdots, a_{p}; b_{1}, \cdots, b_{q}; xt)$$

$$= \frac{\Gamma(m) \, \Gamma(n)}{\Gamma(m+n)} \, _{p+1}F_{q+1}(m, a_{1}, \cdots, a_{p}; m+n, b_{1}, \cdots, b_{q}; t)$$
(5.7)

$${}_{2}F_{1}(a_{1}, a_{2}; b; x) = \frac{\Gamma(b)}{\Gamma(a_{2})\Gamma(b - a_{2})} \int_{0}^{1} dt \, t^{a_{2}-1} (1 - t)^{b-a_{2}-1} (1 - xt)^{-a_{1}}$$

$${}_{2}F_{1}(a_{1}, a_{2}; b; 1) = \frac{\Gamma(b)\Gamma(b - a_{1} - a_{2})}{\Gamma(b - a_{1})\Gamma(b - a_{2})}$$
(5.8)

5.1.2 Dilogarithm function

The dilogarithm function can be defined by the sum

$$\operatorname{Li}_{2}(z) = \frac{z^{1}}{1^{2}} + \frac{z^{2}}{2^{2}} + \frac{z^{3}}{3^{2}} + \dots = \sum_{n=1}^{\infty} \frac{z^{n}}{n^{2}}$$
(5.9)

or the integral

$$\operatorname{Li}_{2}(z) = -\int_{0}^{1} dt \, \frac{\ln(1-t\,z)}{t} = -\int_{0}^{z} dt \, \frac{\ln(1-t)}{t}$$
(5.10)

The derivative of the dilogarithm function is given by

$$\frac{d}{dz}\operatorname{Li}_2(z) = -\frac{\ln(1-z)}{z}$$
(5.11)

The major functional equations for the dilogarithm function are given by

$$\begin{aligned} \operatorname{Li}_{2}(1-z) &= -\operatorname{Li}_{2}(z) - \ln(z) \ln(1-z) + \zeta(2) \\ \operatorname{Li}_{2}\left(\frac{1}{z}\right) &= -\operatorname{Li}_{2}(z) - \frac{1}{2} \ln^{2}(-z) - \zeta(2) \\ \operatorname{Li}_{2}\left(\frac{1}{1-z}\right) &= \operatorname{Li}_{2}(z) + \ln(1-z) \ln(-z) - \frac{1}{2} \ln^{2}(1-z) + \zeta(2) \\ \operatorname{Li}_{2}\left(-\frac{1-z}{z}\right) &= \operatorname{Li}_{2}(z) + \ln(z) \ln(1-z) - \frac{1}{2} \ln^{2}(z) - \zeta(2) \\ \operatorname{Li}_{2}\left(-\frac{z}{1-z}\right) &= -\operatorname{Li}_{2}(z) - \frac{1}{2} \ln^{2}(1-z) \\ \operatorname{Li}_{2}(z^{2}) &= 2 \left[\operatorname{Li}_{2}(z) + \operatorname{Li}_{2}(-z)\right] \end{aligned}$$
(5.12)

where the Riemann zeta function is defined by

$$\zeta(n) = \sum_{k=1}^{\infty} \frac{1}{k^n} \tag{5.13}$$

The values of $\zeta(n)$ for small positive integer values of n are

$$\begin{aligned} \zeta(2) &= \frac{\pi^2}{6} \\ \zeta(3) &= 1.202\,056\,903\,159\,59\cdots \\ \zeta(4) &= \frac{\pi^4}{90} \end{aligned}$$
(5.14)

5.1.3 The volume element in *d* dimensions

We consider general formula of the volume element in d dimensions

$$dV_d = d^d r = r^{d-1} dr \, d\Omega_d \qquad \text{(Euclidean space)} \tag{5.15}$$

where the area element in Euclidean space is

$$d\Omega_d = \prod_{\ell=1}^{d-1} \sin^{d-1-\ell} \theta_\ell \, d\theta_\ell = \int_0^\pi d\theta_1 \sin^{d-2} \theta_1 \cdots \int_0^\pi d\theta_{d-2} \sin \theta_{d-2} \int_0^{2\pi} d\theta_{d-1}$$
$$= \int_0^{2\pi} d\phi \prod_{\ell=1}^{d-2} \int_0^\pi d\theta_\ell \sin^\ell \theta_\ell \quad \text{where} \quad \theta \in [0,\pi], \quad \phi \in [0,2\pi]$$
(5.16)

and the following relations hold

$$\int_0^{\pi} d\theta \,\sin^\ell \theta = \sqrt{\pi} \,\frac{\Gamma\left(\frac{\ell+1}{2}\right)}{\Gamma\left(\frac{\ell+2}{2}\right)}, \qquad \prod_{\ell=1}^{d-2} \int_0^{\pi} d\theta_\ell \,\sin^\ell \theta_\ell = \frac{\pi^{\frac{d}{2}-1}}{\Gamma\left(d/2\right)} \tag{5.17}$$

d	$\Gamma(d/2)$	$\int d \Omega_d$
1 2 3 4	$\sqrt{\pi}$ $\frac{1}{\sqrt{\pi}/2}$ 1	2 2π 4π $2\pi^2$

Table 5.1: The values of $\Gamma(d/2)$ and $d \Omega_d$.

The relation between $d \Omega_{d-1}$ and $d \Omega_{d-2}$ is given by

$$d\Omega_{d-1} = d\Omega_{d-2} \int_0^{\pi} d\theta \sin^{d-3} \theta = d\Omega_{d-2} \int_{-1}^1 d\cos\theta \left(1 - \cos^2\theta\right)^{\frac{d-4}{2}} = \int_0^{2\pi} d\phi \prod_{\ell=1}^{d-3} \int_0^{\pi} d\theta_\ell \sin^\ell \theta_\ell$$
(5.18)

Finally, we give the values of $\Gamma(d/2)$ and $d \Omega_d$ for $d = 1, \dots, 4$ in Table 5.1.

5.2 Integration measures

In this subsection we will first derive the integration measures for both the initial state and final state splittings. In order to extract the correct pole structures we have to define good parametrizations of the kinematics. Then we will also list master integrals and general formulae that are used to extract singularities and finite terms in Section 2.5.

5.2.1 Final state splitting

Integration measure: $a_\ell = 1$

With the kinematics as defined in Section 2.4.3, we obtain the integration measure

$$d\zeta_p = \frac{(2\,p_\ell \cdot Q)^{1-\epsilon}}{16\,\pi^2} \frac{(4\,\pi)^{\epsilon}}{\Gamma(1-\epsilon)} \int_0^{y_{\text{max}}} dy\, y^{-\epsilon}\,\lambda^{1-2\,\epsilon} \int_0^1 dz\, \left[z\,(1-z)\right]^{-\epsilon} \int \frac{d^{d-2}\,\Omega}{\Omega_{d-2}}$$
(5.19)

for the integration in $d = 4 - 2\epsilon$ dimensions. Here,

$$y_{\max} = \left(\sqrt{a_{\ell}} - \sqrt{a_{\ell} - 1}\right)^2 = 2 a_{\ell} - 1 - 2 \sqrt{a_{\ell} (a_{\ell} - 1)}$$
(5.20)

For the integration of the interference terms, we need to consider the additional azimuthal angle of the emitted particle. Hence we keep the second angular integration,

$$d\Omega_{d-2} = d\Omega_{d-3} \int_{-1}^{1} d\cos\varphi \left(1 - \cos^{2}\varphi\right)^{-\frac{(1+2\epsilon)}{2}} = \frac{2^{1-2\epsilon}\pi^{\frac{1}{2}-\epsilon}}{\Gamma\left(\frac{1}{2}-\epsilon\right)} \int_{0}^{1} dv' \left[v'(1-v')\right]^{-\frac{(1+2\epsilon)}{2}}, \qquad \int d\Omega_{d} = \frac{2\pi^{d/2}}{\Gamma\left(d/2\right)}$$
(5.21)

where φ is now the azimuthal angle of the emitted parton. So the integration measure becomes

$$d\zeta_p = \frac{(2\,p_\ell \cdot \hat{Q})^{1-\epsilon}}{16} \frac{\pi^{-\frac{5}{2}+\epsilon}}{\Gamma\left(\frac{1}{2}-\epsilon\right)} \int_0^{y_{\text{max}}} dy \, y^{-\epsilon} \, \lambda^{1-2\,\epsilon} \int_0^1 dz \, [z\,(1-z)]^{-\epsilon} \int_0^1 dv' \, [v'\,(1-v')]^{-\frac{(1+2\epsilon)}{2}}$$
(5.22)

In the integration, we will use v' as a variable; however, the splitting functions are written in terms of $v = v[\bar{z}(y), z, v']$

$$v = (4v'-2) \left[z(1-z)\bar{z}(1-\bar{z}) \right]^{\frac{1}{2}} + z + \bar{z} - 2z\,\bar{z}$$
(5.23)

Note that Eq. (5.23) has been derived in a specific frame which generally does not coincide with the frame following in the inverse transform from m- to (m+1)-particle phase space. In general, v is defined in Eq. (2.111), *i.e.*

$$v = \frac{\hat{p}_j \cdot \hat{p}_k}{\hat{p}_k \cdot P_\ell}$$

where \hat{p}_k needs to be calculated using the Lorentz transformation

$$\hat{p}^{\mu}_k = \Lambda(\hat{K}, K)^{\mu}{}_{\nu} p^{\nu}_k$$

We now consider the special case in which $a_{\ell} = 1$ (which corresponds to $\lambda = 1 - y$). We start with the integral

$$I = \int_0^1 dy \, y^{-\epsilon} \, (1-y)^{1-2\epsilon} \, \int_0^1 dz \, [z \, (1-z)]^{-\epsilon}$$
(5.24)

We now make a change of variable such that

$$x = y + z (1 - y) \tag{5.25}$$

thus we have

$$I = \int_0^1 dy \, y^{-\epsilon} \int_y^1 dx \, \left[(x - y) \, (1 - x) \right]^{-\epsilon}$$
(5.26)

We now change the integration order

$$\int_0^1 dy \int_y^1 dx \quad \longrightarrow \quad \int_0^1 dx \int_0^x dy$$

and obtain

$$I = \int_0^1 dx \int_0^x dy \, y^{-\epsilon} \, \left[(x - y) \, (1 - x) \right]^{-\epsilon} \tag{5.27}$$

We now make a change of variable once again

$$u = \frac{y}{x}$$

finally leads to

$$I = \int_0^1 dx \, x^{1-2\epsilon} \, (1-x)^{-\epsilon} \int_0^1 du \, u^{-\epsilon} \, (1-u)^{-\epsilon}$$
(5.28)

Now the integration measure becomes

$$d\zeta_p = \frac{(2p_\ell \cdot \hat{Q})^{1-\epsilon}}{16} \frac{\pi^{-\frac{5}{2}+\epsilon}}{\Gamma\left(\frac{1}{2}-\epsilon\right)} \int_0^1 du \, u^{-\epsilon} \, (1-u)^{-\epsilon} \int_0^1 dx \, x^{1-2\epsilon} \, (1-x)^{-\epsilon} \int_0^1 dv' \, \left[v'\left(1-v'\right)\right]^{-\frac{(1+2\epsilon)}{2}}$$
(5.29)

Here u and x can be expressed in terms of momenta, we have

$$x = \frac{\hat{p}_j \cdot Q}{\gamma \, p_\ell \cdot Q}, \qquad u = \gamma \frac{\hat{p}_\ell \cdot \hat{p}_j}{\hat{p}_j \cdot Q}$$
(5.30)

For $a_{\ell} = 1$ case we have $\gamma = 1$. It follows immediately that x is purely soft variable and u is purely collinear variable. Note that the collinear integrations do not depend on v' (or the azimuthal angle φ), while the soft integrations do depend on v' or φ .

Integration measure: general case $a_\ell \neq 1$

Again, we start with the integral

$$I = \int_{0}^{y_{\text{max}}} dy \, y^{-\epsilon} \, \lambda^{1-2\epsilon} \, \int_{0}^{1} dz \, \left[z \, (1-z) \right]^{-\epsilon} \tag{5.31}$$

The first variable transformation is defined via

$$z = \frac{x - x_0}{1 - x_0} \tag{5.32}$$

leading to

$$I = \int_{0}^{y_{\text{max}}} dy \, y^{-\epsilon} \, \gamma^{1-2\,\epsilon} \, \int_{x_0}^{1} dx \, \left[(1-x) \, (x-x_0) \right]^{-\epsilon} \tag{5.33}$$

Next we use the rescaling parameter δ defined as

$$1 + \delta = \frac{a_{\ell} - \sqrt{a_{\ell}^2 - a_{\ell} \frac{4x}{(1+x)^2}}}{2x} (1+x)^2 \ge 1$$
(5.34)

and redefine

$$y = \delta u \tag{5.35}$$

We then obtain

$$I = \int_0^1 du \, u^{-\epsilon} \int_0^1 dx \, \delta^{1-\epsilon} \, \gamma^{1-2\epsilon} \, \left[(1-x) \, (x-x_0) \right]^{-\epsilon} \tag{5.36}$$

Finally the complete integration measure is

$$d\zeta_{p} = \frac{(2 p_{\ell} \cdot \hat{Q})^{1-\epsilon}}{16} \frac{\pi^{-\frac{5}{2}+\epsilon}}{\Gamma\left(\frac{1}{2}-\epsilon\right)} \times \int_{0}^{1} du \, u^{-\epsilon} \int_{0}^{1} dx \, \delta^{1-\epsilon} \, \gamma^{1-2\epsilon} \, \left[(1-x) \, (x-x_{0})\right]^{-\epsilon} \int_{0}^{1} dv' \, \left[v'\left(1-v'\right)\right]^{-\frac{1+2\epsilon}{2}}$$
(5.37)

For $a_{\ell} = 1$ we of course obtain the result in the last subsection.

5.2.2 Initial state splitting

Integration measure

For the initial state splitting, the integration measure is given by

$$d\zeta_p = \frac{d^d \hat{p}_j}{(2\pi)^d} 2\pi \,\delta^+(\hat{p}_j^2) \,\frac{\alpha}{\hat{\alpha}} = \frac{d^d \hat{p}_j}{(2\pi)^d} 2\pi \,\delta^+(\hat{p}_j^2) \frac{d\hat{\eta}_a}{d\eta_a}$$
(5.38)

where $\alpha = \eta_b s$ and $\hat{\alpha} = \eta_b s - 2 p_A \cdot \hat{p}_j$. The factor $\alpha/\hat{\alpha}$ is just the derivative $d\hat{\eta}_a/d\eta_a$ calculated from the relation $\hat{K}^2 = K^2$. Using the identity

$$d^{d}p\,\delta^{+}(p^{2}) = \frac{1}{2}\,(p^{0})^{d-3}\,dp^{0}\,d\Omega_{d-2}\,d\cos\theta\,\left(1-\cos^{2}\theta\right)^{\frac{d-4}{2}}$$
(5.39)

We then obtain

$$d\zeta_p = dx \, dy \frac{(\eta_a \, \eta_b \, s)^{1-\epsilon} x^{\epsilon-1}}{\Gamma(1-\epsilon)(4 \, \pi)^{2-\epsilon}} \, (1-x)^{-2\epsilon} \left[\frac{y}{1-x} \left(1 - \frac{y}{1-x} \right) \right]^{-\epsilon} \, \frac{1}{x} \, \theta(x-\eta_a)$$
$$\times \, \theta(x \, (1-x)) \, \theta(y) \, \theta\left(1 - \frac{y}{1-x} \right) \tag{5.40}$$

where we calculated in $d=4-2\,\epsilon$ dimensions. However, note that the change of variables in η_a corresponds to

$$\int_{0}^{1} d\hat{\eta}_{a} = \frac{1}{x} \int_{0}^{x} d\eta_{a}$$
(5.41)

so we have to impose the condition $x > \eta_a$; alternatively, we can not make the change of variables and keep $\hat{\eta}_a = \eta_a/x$ as the integration variable; this is the approach followed in [22]. For comparison, we will instead use

$$d\zeta_p \to d\zeta_p \frac{1}{x} \theta(x - \eta_a)$$
 (5.42)

For the integration of the interference terms, we need to keep the additional azimuthal angle φ of the emitted particle in $d\Omega_{d-2}$; see Eq. (5.21). Using the variables as defined in Section 2.4.4, we obtain for the integration measure including the additional angle,

$$d\zeta_{p} = dx \, dy \, dv' \, \frac{(\eta_{a} \, \eta_{b} \, s)^{1-\epsilon} x^{\epsilon-1}}{(4 \, \pi)^{2}} (1-x)^{-2\epsilon} \left[\frac{y}{1-x} \left(1 - \frac{y}{1-x} \right) \right]^{-\epsilon} \\ \times \frac{\pi^{\epsilon-\frac{1}{2}}}{\Gamma\left(\frac{1-2\epsilon}{2}\right)} \left(v' \, (1-v') \right)^{-\frac{(1+2\epsilon)}{2}} \frac{1}{x} \, \theta(x-\eta_{a}) \\ \times \, \theta(x \, (1-x)) \, \theta(y) \, \theta\left(1 - \frac{y}{1-x} \right) \, \theta(v' \, (1-v'))$$
(5.43)

In the integration, we will use v' as a variable; however, the splitting functions are written in terms of $v = v(\bar{z}, y', v')$

$$v = (4v'-2) \left[y'(1-y') \,\bar{z} \,(1-\bar{z}) \right]^{\frac{1}{2}} + y' + \bar{z} - 2\,y'\,\bar{z}$$
(5.44)

hence we have

$$v_{\max} = y' + \bar{z} - 2y'\bar{z} + 2\left[y'(1-y')\bar{z}(1-\bar{z})\right]^{\frac{1}{2}}$$

$$v_{\min} = y' + \bar{z} - 2y'\bar{z} - 2\left[y'(1-y')\bar{z}(1-\bar{z})\right]^{\frac{1}{2}}$$
(5.45)

Note, however, that we can only use Eq. (5.44) in the singular limits $(x \to 1 \text{ or } y' \to 0)$ where the dependence on v' disappears; for the finite parts, we need to use the original definition given by Eq. (2.135)

$$v = \frac{\hat{\eta}_{a} \eta_{b} s \hat{p}_{j} \cdot \hat{p}_{k}}{2 \hat{p}_{k} \cdot \hat{Q} \hat{p}_{j} \cdot \hat{Q}} = \frac{1}{1 - x} \frac{\hat{p}_{j} \cdot \hat{p}_{k}}{\hat{p}_{k} \cdot \hat{Q}}$$

where \hat{p}_k needs to be calculated using the Lorentz transformation

$$\hat{p}_{k}^{\mu} = \Lambda(x, y')^{\mu}{}_{\nu} p_{k}^{\nu} = \Lambda(\hat{K}, K)^{\mu}{}_{\nu} p_{k}^{\nu} \quad .$$
(5.46)

In the limits $x \to 1$ or $y' \to 0$, we obtain $\hat{p}_k \to p_k$; and \hat{p}_j is parametrized according to Sudhakov parametrization (e.g. [109]). The change of variables defined in Section 2.4.4 always requires that the integration over v' is performed before the integration over y in Eq. (5.43). It is worth mentioning that \bar{z} is not integrated out and therefore still depends on x, y' and v' through the Lorentz transformation of \hat{p}_k . Also note that the collinear integrations do not depend on v'(or the azimuthal angle φ), while the soft integrations do depend on v' or φ .

5.2.3 Master integrals

Interference between initial and final states

The general master integrals appearing in Section 2.5.4 are

$$\int_{0}^{1} \frac{dy}{y} \left\{ \frac{z_0}{\sqrt{4y^2(1-z_0)+z_0^2}} - 1 \right\} = \ln z_0$$
(5.47)

and

$$\int_{0}^{1} dv \frac{\left[v\left(1-v\right)\right]^{-\frac{1+2\epsilon}{2}}}{\left(4v-2\right)\left[y\left(1-y\right)z\left(1-z\right)\right]^{\frac{1}{2}}+2y+z-2yz}$$
$$=\frac{\pi}{\sqrt{4y^{2}(1-z)+z^{2}}} \left\{1-2\epsilon \ln\left[\frac{\sqrt{4y^{2}(1-z)+z^{2}}}{2\left(2y+z-2yz+\sqrt{4y^{2}(1-z)+z^{2}}\right)}\right]\right\} + \mathcal{O}(\epsilon^{2})$$
(5.48)

Interference between final (final and initial) states

The general master integrals appearing in Section 2.5.5 are

$$\int_{0}^{1} \frac{du}{u} \frac{1}{\sqrt{1 + 4a_0(1 + a_0)u^2}} \ln\left(1 + 4a_0(1 + a_0)u^2\right)$$

= $-\left[\frac{\pi^2}{12} + \ln(1 + 2a_0)\ln\left[2(1 + a_0)\right] + \text{Li}_2\left(-(1 + 2a_0)\right) + \text{Li}_2\left(-2a_0\right)\right]$ (5.49)

$$\int_{0}^{1} dv \, \frac{\ln[v(1-v)]}{\sqrt{v(1-v)}(av+b)} = \frac{2\pi}{\sqrt{b}\sqrt{a+b}} \ln \frac{\sqrt{b}\sqrt{a+b}}{\left(\sqrt{a+b}+\sqrt{b}\right)^{2}}$$
$$\int_{\epsilon}^{1} du \, \frac{\ln u}{u\sqrt{1+bu^{2}}} = \frac{1}{2} \ln^{2} \left(\frac{2}{\sqrt{1+b}+1}\right) + \frac{1}{2} \text{Li}_{2} \left(\frac{\sqrt{1+b}-1}{\sqrt{1+b}+1}\right) - \frac{1}{2} \ln^{2} \epsilon$$
$$\int_{\epsilon}^{1} du \, \frac{\ln u}{u} = -\frac{1}{2} \ln^{2} \epsilon$$
(5.50)

and

$$\int_{0}^{1} dx \, \frac{\ln\left(1+ax^{2}\right)}{x\sqrt{1+ax^{2}}} = -\left[\frac{\pi^{2}}{12} + \frac{1}{2}\ln(a+1)\ln\left[\sqrt{a+1}+1\right] + \text{Li}_{2}\left(-\sqrt{a+1}\right) + \text{Li}_{2}\left(1-\sqrt{a+1}\right)\right]$$
(5.51)

Here, ϵ in Eq. (5.50) is just an infinitesimal parameter.

5.2.4 Pole extractions

Single poles

For a function having a single pole, e.g. x = 0, we use

$$\int_0^1 dx \, x^{-(1+\epsilon)} g(x) = \int_0^1 dx \, x^{-(1+\epsilon)} \left[g(x) - g(0) \right] - \frac{g(0)}{\epsilon} \tag{5.52}$$

where g(x) is nonsingular function at x = 0. For a general case where we integrate only up to x_{max} instead of 1, the equation above becomes

$$\int_{0}^{x_{\max}} dx \, x^{-(1+\epsilon)} g(x) = \int_{0}^{x_{\max}} dx \, x^{-(1+\epsilon)} \left[g(x) - g(0) \right] - \frac{g(0)}{\epsilon} + g(0) \ln x_{\max} \tag{5.53}$$

Double poles

In order to extract the double poles, e.g. x = 0 and y = 0, we consider the following integral

$$I = \int_0^1 dx \int_0^1 dy \, \frac{1}{x^{1+\epsilon}} \frac{1}{y^{1+\epsilon'}} g(x,y)$$
(5.54)

where g(x, y) is nonsingular function at x = 0 and y = 0. We use Eq. (5.52) twice and obtain the general formula

$$I = \int_{0}^{1} dx \int_{0}^{1} dy \frac{1}{x^{1+\epsilon}} \frac{1}{y^{1+\epsilon'}} [g(x,y) - g(x,0) - g(0,y) + g(0,0)] - \frac{1}{\epsilon'} \int_{0}^{1} dx \frac{1}{x^{1+\epsilon}} [g(x,0) - g(0,0)] - \frac{1}{\epsilon} \int_{0}^{1} dy \frac{1}{y^{1+\epsilon'}} [g(0,y) - g(0,0)] + \frac{g(0,0)}{\epsilon\epsilon'}$$
(5.55)

5.3 Colour algebra

Notations

In this section we will give a brief descriptions about the manipulations of colour algebra; we will follow Catani-Seymour's notations very closely. Here we will only consider the processes that involve the final state QCD partons; in the case of processes that involve the initial state QCD partons please refer to [22]. First, it will be convenient to introduce a basis $\{ | c_1 \cdots c_m > \otimes | s_1 \cdots s_m > \}$ in colour + helicity space in such a way that

$$\mathcal{M}_m^{c_1 \cdots c_m, s_1 \cdots s_m}(p_1, \cdots, p_m) = \{ \langle c_1 \cdots c_m \mid \otimes \langle s_1 \cdots s_m \mid \} \mid 1, \cdots, m \rangle_m$$
(5.56)

where $| 1, \dots, m \rangle_m$ is a vector in colour + helicity space; $\{c_1 \dots c_m\}$ and $\{s_1 \dots s_m\}$ are colour indices (for gluons the values take $1 \dots N^2 - 1$, while the values take $1 \dots N$ for quarks or antiquarks) and spin indices (the values take $\mu = 1, \dots, d-2$ for gluons and s = 1, 2 for massless fermions), respectively. According to this notation, the matrix element squared (summed over final-state colours and spins) can be written as

$$|\mathcal{M}_{m}|^{2} = {}_{m} < 1, \cdots, m | 1, \cdots, m >_{m}$$
 (5.57)

It is useful to define the square of colour-correlated tree-amplitudes according to

$$|\mathcal{M}_{m}^{i,k}|^{2} = {}_{m} < 1, \cdots, m | \mathbf{T}_{i} \cdot \mathbf{T}_{k} | 1, \cdots, m >_{m}$$

=
$$\left[\mathcal{M}_{m}^{A_{1} \cdots B_{i} \cdots B_{k} \cdots A_{m}}(p_{1}, \cdots, p_{m})\right]^{*} (T^{C})_{B_{i}A_{i}} (T^{C})_{B_{k}A_{k}} \mathcal{M}_{m}^{A_{1} \cdots A_{i} \cdots A_{k} \cdots A_{m}}(p_{1}, \cdots, p_{m})$$
(5.58)

where we have associated a colour charge T_i with the emission of a gluon from each parton *i*. Here we follow the notations in Chapter 1 where $(T^A)_{BC} \equiv -i f^{ABC}$ (colour-charge matrix in the adjoint representation) if the emitting parton *i* is a gluon and $T^A_{ab} \equiv t^A_{ab}$ (colour-charge matrix in the fundamental representation) if the emitting parton *i* is a quark; if the emitting parton *i* is a antiquark, then we have $T^A_{ab} \equiv \bar{t}^A_{ab} = -t^A_{ba}$). It is straightforward to check that the colour-charge algebra obeys

$$\mathbf{T}_i \cdot \mathbf{T}_j = \mathbf{T}_j \cdot \mathbf{T}_i \qquad \text{if } i \neq j, \qquad \mathbf{T}_i^2 = C_i$$
(5.59)

where C_i is the Casimir operator, *i.e.*, $C_i = C_F = (N^2 - 1)/(2N)$ if *i* is a quark or antiquark and $C_i = C_A = N$ if *i* is a gluon. Each vector $|1, \dots, m \rangle_m$ is a colour-singlet state, therefore colour conservation is

$$\sum_{i=1}^{m} \mathbf{T}_{i} \mid 1, \cdots, m >_{m} = 0$$
(5.60)

Examples

In this subsection we will practice with the simplest cases of colour algebra. For the cases with two or three partons, the colour algebra can be computed in factorized form. First, we consider the case with two partons. Using colour conservation relation, we have

$$\mathbf{T}_{1} \cdot \mathbf{T}_{2} | 1, 2 > = -\mathbf{T}_{1} \cdot \mathbf{T}_{1} | 1, 2 > = -\mathbf{T}_{1}^{2} | 1, 2 > = -\mathbf{T}_{2}^{2} | 1, 2 >$$
(5.61)

so that all the charge operators $\{T_1^2, T_2^2, -T_1 \cdot T_2\}$ are factorizable in terms of the Casimir operator. Now we consider the case with three partons, using colour conservation, we have

$$0 = \left(\sum_{i=1}^{3} \mathbf{T}_{i}\right)^{2} |1, 2, 3\rangle = \left(\mathbf{T}_{1}^{2} + \mathbf{T}_{2}^{2} + \mathbf{T}_{3}^{2} + 2\mathbf{T}_{1} \cdot \mathbf{T}_{2} + 2\mathbf{T}_{1} \cdot \mathbf{T}_{3} + 2\mathbf{T}_{2} \cdot \mathbf{T}_{3}\right) |1, 2, 3\rangle$$
(5.62)

and

$$(\mathbf{T}_1 \cdot \mathbf{T}_2 + \mathbf{T}_1 \cdot \mathbf{T}_3) |1, 2, 3\rangle = -\mathbf{T}_1^2 |1, 2, 3\rangle$$
(5.63)

Combining these two equations we obtain

$$2 \mathbf{T}_2 \cdot \mathbf{T}_3 | 1, 2, 3 \rangle = \left(\mathbf{T}_1^2 - \mathbf{T}_2^2 - \mathbf{T}_3^2 \right) | 1, 2, 3 \rangle$$
(5.64)

and similarly for $T_1 \cdot T_3$ and $T_1 \cdot T_2$. Hence, all the charge operators are factorizable in terms of linear combinations of the Casimir operators C_1 , C_2 and C_3 .

The colour algebra does not factorize any longer when the total number n of partons is $n \ge 4$, e.g. , if n = 4 we have

$$\mathbf{T}_{i}^{2} | 1, 2, 3, 4 \rangle = C_{i} | 1, 2, 3, 4 \rangle, \qquad i = 1, \cdots, 4$$
(5.65)

and

$$\mathbf{T}_i \cdot \sum_{j=1}^{4} \mathbf{T}_j | 1, 2, 3, 4 \rangle = 0, \qquad i = 1, \cdots, 4$$
 (5.66)

in order to single out two independent charge operators, we can write

$$T_{3} \cdot T_{4} | 1, 2, 3, 4 \rangle = \left[\frac{1}{2} (C_{1} + C_{2} - C_{3} - C_{4}) + T_{1} \cdot T_{2} \right] | 1, 2, 3, 4 \rangle,$$

$$T_{2} \cdot T_{4} | 1, 2, 3, 4 \rangle = \left[\frac{1}{2} (C_{1} + C_{3} - C_{2} - C_{4}) + T_{1} \cdot T_{3} \right] | 1, 2, 3, 4 \rangle,$$

$$T_{2} \cdot T_{3} | 1, 2, 3, 4 \rangle = \left[\frac{1}{2} (C_{4} - C_{1} - C_{2} - C_{3}) - T_{1} \cdot T_{2} - T_{1} \cdot T_{3} \right] | 1, 2, 3, 4 \rangle,$$

$$T_{1} \cdot T_{4} | 1, 2, 3, 4 \rangle = - \left[C_{1} + T_{1} \cdot T_{2} + T_{1} \cdot T_{3} \right] | 1, 2, 3, 4 \rangle$$
(5.67)

and express all the charge operators in terms of Casimir invariants and $T_1 \cdot T_2$ and $T_1 \cdot T_3$.

5.4 Phase space integration

In this section, we present the parameterization of the n-particle phase space that we use to evaluate the cross sections. The n-body phase space in d dimensions is

$$dPS_n = (2\pi)^d \,\delta^d \left(p_a + p_b - \sum_{i=1}^n p_i \right) \prod_{i=1}^n \frac{d^d p_i}{(2\pi)^{d-1}} \,\delta^+(p_i^2 - m_i^2) = (2\pi)^d \,\delta^d \left(p_a + p_b - \sum_{i=1}^n p_i \right) \prod_{i=1}^n \frac{d^{d-1}\vec{p_i}}{(2\pi)^{d-1} \, 2E_i}$$
(5.68)

where $\delta^+(q^2 - m^2) = \delta(q^2 - m^2) \theta(q^0)$ ensures that we only consider positive-energy particles. The case of n = 1 is particularly simple

$$dPS_1 = 2\pi \delta^+ (Q^2 - m^2)|_{Q = p_a + p_b}$$
(5.69)

For the two-body phase space (n = 2), Eq. (5.68) reduces to

$$dPS_{2}(Q \to p_{1} + p_{2}) = \int \frac{d^{d-1}\vec{p}_{1}}{(2\pi)^{d-1} 2E_{1}} \int \frac{d^{d-1}\vec{p}_{2}}{(2\pi)^{d-1} 2E_{2}} (2\pi)^{d} \delta^{d}(Q - p_{1} - p_{2})$$

$$= \frac{1}{4(2\pi)^{d-2}} \int \frac{d^{d-1}\vec{p}_{1}}{E_{1}E_{2}} \underbrace{\int d^{d-1}\vec{p}_{2} \,\delta^{d-1}(\vec{Q} - \vec{p}_{1} - \vec{p}_{2})}_{=1} \delta(Q^{0} - E_{1} - E_{2})$$

$$= \frac{1}{4(2\pi)^{d-2}} \int \frac{d^{d-1}\vec{p}_{1}}{E_{1}E_{2}} \frac{\delta(Q^{0} - E_{1} - E_{2})}_{=1} \delta(Q^{0} - E_{1} - E_{2}) \delta(Q^{0} - E_{1} - E_$$

$$= \frac{1}{4(2\pi)^{d-2}} \int \frac{d^{-p_1}}{E_1 E_2} \delta(Q^0 - E_1 - E_2)$$
(5.70)

where $E_i = \sqrt{\vec{p_i} \cdot \vec{p_i} + m_i^2}$. We now consider the massive case in which $p_i^2 = m_i^2$ (i = 1, 2). Using the identities

$$\int dp_2^0 \,\delta\left(p_2^0 - \sqrt{\vec{p}_2 \cdot \vec{p}_2 + m_2^2}\right) = 1 \quad \text{and} \quad \delta(p_2^2 - m_2^2) \,\theta(p_2^0) = \frac{1}{2 \, p_2^0} \,\delta\left(p_2^0 - \sqrt{\vec{p}_2 \cdot \vec{p}_2 + m_2^2}\right) \tag{5.71}$$

Hence we have

$$\frac{d^3 \vec{q}}{2 q^0} = d^4 q \,\delta(q^2 - m^2) \,\theta(q^0) \quad \text{with} \quad q = p_2 \tag{5.72}$$

Using

$$d^{3}\vec{p}_{1} = \vec{p}_{1} \cdot \vec{p}_{1} \, d|\vec{p}_{1}| \, d\Omega \,, \qquad d\Omega = d\cos\theta \, d\phi, \qquad |\vec{p}_{1}| \, d|\vec{p}_{1}| = p_{1}^{0} \, dp_{1}^{0} \tag{5.73}$$

then we obtain

$$dPS_{2} = \frac{1}{8\pi^{2}} |\vec{p_{1}}| dp_{1}^{0} d\Omega \theta \left(\sqrt{s} - p_{1}^{0}\right) \delta \left(p_{1}^{0} - \frac{s + m_{1}^{2} - m_{2}^{2}}{2\sqrt{s}}\right) \frac{1}{2\sqrt{s}}$$
$$= \frac{1}{32\pi^{2}} \frac{\sqrt{\lambda(s, m_{1}^{2}, m_{2}^{2})}}{s} d\Omega \theta \left(\sqrt{s} - \frac{s + m_{1}^{2} - m_{2}^{2}}{2\sqrt{s}}\right)$$
(5.74)

with

$$|\vec{p}_1| = \frac{\sqrt{\lambda(s, m_1^2, m_2^2)}}{2\sqrt{s}} \quad \text{and} \quad \lambda(x, y, z) = x^2 + y^2 + z^2 - 2(xy + yz + zx)$$
(5.75)

Here we work in the CM frame in which $Q = (\sqrt{s}, \vec{0})$. If only one of the two final state particles is massive with mass $m_1 = m (m_2 = 0)$, then the two-particle phase space integration is

$$dPS_2 = \frac{1}{32\pi^2} \left(1 - \frac{m^2}{s}\right) d\Omega, \quad \text{provided that} \quad s > m^2 \tag{5.76}$$

Multiparticle phase space decomposition

Now consider a system Z with particles of momenta p_1, p_2, \dots, p_n in the final state. We partition the final state particles into two subsystems X and Y:

$$Z \longrightarrow X(p_1 p_2 \cdots p_j) + Y(p_{j+1} \cdots p_n)$$
(5.77)

The *n*-body phase space can be decomposed as follows:

$$dPS_n(Z \to p_1 p_2 \cdots p_n) = dPS_2(Z \to X Y) \frac{dm_X^2}{2\pi} \frac{dm_Y^2}{2\pi}$$
$$\cdot dPS_j(X \to p_1 p_2 \cdots p_j) dPS_{n-j}(Y \to p_{j+1} \cdots p_n)$$
(5.78)

where m_X and m_Y are resonant masses of the decaying particles X and Y, respectively. The ranges of the invariant mass m_X and m_Y are

$$\sum_{i=1}^{j} m_i \le m_X, \qquad \sum_{i=j+1}^{n} m_i \le m_Y, \qquad m_X + m_Y \le m_Z$$
(5.79)

0

Hence a particle decays into another two particles (with resonance of masses m_X and m_Y , respectively) which later decay into j and n - j particles, respectively. The subsystems X and Y can be further reduced recursively until we obtain the products of two-particle phase spaces. In the special case where Y is a single particle p_n , then we have

$$dPS_n(Z \to p_1 p_2 \cdots p_n) = dPS_2(Z \to XY) \cdot \frac{dm_X^2}{2\pi} \cdot dPS_{n-1}(X \to p_1 p_2 \cdots p_{n-1})$$

$$m_1 + m_2 + \cdots + m_{n-1} \le m_X \le m_Z - m_n$$
(5.80)

We now consider the parametrization of the three-particle phase space [97]

$$Z \longrightarrow X(p_1 p_3) + Y(p_2) \tag{5.81}$$

In this case Y is a single particle p_2 . We decompose the three-particle phase space into two two-particle phase spaces as follows

$$dPS_3(Z \to p_1 p_2 p_3) = dPS_2(Z \to XY) \cdot \frac{ds_{13}}{2\pi} \cdot dPS_2(X \to p_1 p_3)$$
(5.82)

where $s_{ij} = (p_i + p_j)^2 = p_{ij}^2$ and $p_1 + p_2 + p_3 = Q$. We work in the CM frame in which $Q = (\sqrt{s}, \vec{0})$, hence we have $\vec{p}_{13} = -\vec{p}_2$ and $|\vec{p}_{13}| = 1/2 \sqrt{\lambda(s, s_{13}, m_2^2)/s}$. Using

$$dPS_2(Z \to X Y) = \frac{1}{32 \pi^2} \frac{\sqrt{\lambda(s, s_{13}, m_2^2)}}{s} d\cos\theta_{13} d\phi_{13} \theta \left(\sqrt{s} - \frac{s + s_{13} - m_2^2}{2\sqrt{s}}\right),$$

$$dPS_2(X \to p_1 p_3) = \frac{1}{(4 \pi)^2 |\vec{p}_{13}|} dp_1^0 d\phi_1$$
(5.83)

Thus we obtain

$$dPS_3(Z \to p_1 p_2 p_3) = \frac{1}{(2\pi)^5} \frac{s}{32} d\cos\theta_{13} d\phi_{13} d\phi_1 dx_1 dx_2$$
(5.84)

where θ_{13} , ϕ_{13} and ϕ_1 are Euler angles; θ_{13} and ϕ_{13} are the polar and azimuthal angles of \vec{p}_{13} , respectively, and ϕ_1 is the azimuthal angle of \vec{p}_1 with respect to the axis pointing along \vec{p}_{13} . If we integrate out the Euler angles then the phase space depends only on x_1 and x_2 . We now parametrize the four-particle phase space according to

$$Z \longrightarrow X(p_1 p_2) + Y(p_3 p_4) \tag{5.85}$$

We decompose the four-particle phase space into three two-particle phase spaces as follows

$$dPS_4(Z \to p_1 p_2 p_3 p_4) = dPS_2(Z \to XY) \cdot \frac{ds_{12}}{2\pi} \frac{ds_{34}}{2\pi} \cdot dPS_2(X \to p_1 p_2) dPS_2(Y \to p_3 p_4)$$
(5.86)

where $p_1 + p_2 + p_3 + p_4 = Q$. We work in the CM frame and choose the x axis arbitrarily, hence we have $\vec{p}_{12} = -\vec{p}_{34}$ and $|\vec{p}_{12}| = (1/2)\sqrt{\lambda(s, s_{12}, s_{34})/s}$. Using

$$dPS_{2}(Z \to XY) = \frac{1}{32 \pi^{2}} \frac{\sqrt{\lambda(s, s_{12}, s_{34})}}{s} d\Omega_{12} \theta \left(\sqrt{s} - \frac{s + s_{12} - s_{34}}{2\sqrt{s}}\right),$$

$$dPS_{2}(X \to p_{1}p_{2}) = \frac{1}{(4 \pi)^{2} |\vec{p}_{12}|} dp_{1}^{0} d\phi_{1},$$

$$dPS_{2}(Y \to p_{3}p_{4}) = \frac{1}{(4 \pi)^{2} |\vec{p}_{34}|} dp_{3}^{0} d\phi_{3}$$
(5.87)

where $d\Omega_{12} = d\cos\theta_{12} d\phi_{12}$; θ_{12} and ϕ_{12} are the polar and azimuthal angles of \vec{p}_{12} , respectively, and $\phi_i (i = 1, 3)$ is the azimuthal angle of \vec{p}_i with respect to the axis pointing along \vec{p}_{ij} , hence we obtain

$$\frac{dPS_4(Z \to p_1 p_2 p_3 p_4)}{(4\pi)^8 \sqrt{\lambda(s, s_{12}, s_{34})}} \frac{ds_{12} ds_{34} d\cos\theta_{12} d\phi_{12} dp_1^0 d\phi_1 dp_3^0 d\phi_3}{(5.88)}$$

The integration over ϕ_{12} is trivial due to the azimuthal symmetry. We can choose the coordinate system in such a way that \vec{p}_{12} points along the z axis and \vec{p}_1 lies in the x-z plane, hence we have $\theta = \theta_{12}$ and $\phi = \pi - \phi_1$. Introducing $y_{ij} = s_{ij}/Q^2$ and $x_i = 2 p_i \cdot Q/Q^2$, we thus have

$$dPS_4(Z \to p_1 p_2 p_3 p_4) = \frac{s^2}{(4\pi)^7 \sqrt{\lambda(1, y_{12}, y_{34})}} dy_{12} dy_{34} dx_1 dx_3 d\phi_3 d\cos\theta d\phi \qquad (5.89)$$

Here the limit of integration variable ϕ is $-\pi < \phi < \pi$; the scalar products we need are

$$p_{1} \cdot p_{3} = \frac{s}{4} x_{1} x_{3} [1 - (\sin \theta_{1} \sin \theta_{3} \cos \phi_{3} - \cos \theta_{1} \cos \theta_{3})],$$

$$p_{1} \cdot p_{4} = \frac{s}{2} (x_{1} - y_{12}) - p_{1} \cdot p_{3},$$

$$p_{2} \cdot p_{3} = \frac{s}{2} (x_{3} - y_{34}) - p_{1} \cdot p_{3},$$

$$p_{2} \cdot p_{4} = \frac{s}{2} (1 - x_{1} - x_{3}) + p_{1} \cdot p_{3}$$
(5.90)

which in terms of the dimensionless quantities y_{ij} can be rewritten as

$$y_{13} = \frac{1}{2} x_1 x_3 \left[1 - (\sin \theta_1 \sin \theta_3 \cos \phi_3 - \cos \theta_1 \cos \theta_3) \right],$$

$$y_{14} = x_1 - y_{12} - y_{13},$$

$$y_{23} = x_3 - y_{34} - y_{13},$$

$$y_{24} = 1 - x_1 - x_3 + y_{13}$$
(5.91)

where θ_i is the angle enclosed between \vec{p}_{ij} and \vec{p}_i . It is determined by

$$\cos \theta_i = \frac{x_i \left(1 + y_{ij} - y_{kl}\right) - 2 y_{ij}}{x_i \sqrt{\lambda(1, y_{ij}, y_{kl})}}$$
(5.92)

with (i, j), (k, l) = (1, 2), (3, 4) and $(i, j) \neq (k, l)$. Furthermore, we have

$$x_j = 1 - x_i + y_{ij} - y_{kl} (5.93)$$

The limits of integration boundary are

$$\begin{cases} 0 < y_{12} < 1 & x_i^- < x_i < x_i^+ \\ 0 < y_{34} < (1 - \sqrt{y_{12}})^2 & 0 < \phi_3 < 2\pi \end{cases}$$
(5.94)

where

$$x_{i}^{\pm} = \frac{1}{2} \left[\left(1 + y_{ij} - y_{kl} \right) \pm \sqrt{\lambda(1, y_{ij}, y_{kl})} \right]$$
(5.95)

The four-particle phase space with massive particles is discussed in [97]. Finally we summarize this subsection by writing the three-particle and four-particle phase spaces in terms of kinematic invariants $s_{ij} = 2 p_i \cdot p_j$ and the *d*-dimensional hypersphere $d\Omega_d$. Using Eq. (5.68), the three-particle phase space is [98]

$$dPS_3 = (2\pi)^{3-2d} 2^{-1-d} (Q^2)^{\frac{2-d}{2}} d\Omega_{d-1} d\Omega_{d-2} (s_{12} s_{13} s_{23})^{\frac{d-4}{2}} \cdot ds_{12} ds_{13} ds_{23} \delta (Q^2 - s_{12} - s_{13} - s_{23})$$
(5.96)

and the four-particle phase space is

$$dPS_4 = (2\pi)^{4-3d} (Q^2)^{1-\frac{d}{2}} 2^{1-2d} (-\Delta_4)^{\frac{d-5}{2}} \theta(-\Delta_4) \,\delta(Q^2 - s_{12} - s_{13} - s_{14} - s_{23} - s_{24} - s_{34}) \\ \cdot d\Omega_{d-1} \,d\Omega_{d-2} \,d\Omega_{d-3} \,ds_{12} \,ds_{13} \,ds_{14} \,ds_{23} \,ds_{24} \,ds_{34}$$
(5.97)

where the Gram determinant Δ_4 is given by $+\Delta_4 = \lambda(s_{12}s_{34}, s_{13}s_{24}, s_{14}s_{23})$ and λ is the Källen function $\lambda(x, y, z) = x^2 + y^2 + z^2 - 2(xy + yz + zx)$. If we take the unit matrix element and integrate out the three-particle and four-particle phase spaces, respectively, we obtain

$$\int_{CS} dPS_3 = 8\pi \int dPS_3 = \frac{Q^2}{32\pi^2}, \qquad \int_{CS} dPS_4 = 8\pi \int dPS_4 = \frac{Q^4S}{3072\pi^4}$$
(5.98)

An alternative parametrization of four-particle phase space is [15]

$$\int dPS_4 = A_0 \left(\frac{1}{8\pi^2}\right)^2 \frac{1}{16} S Q^2 \int dy_{123} \int dy_{134} \int dy_{13} \theta(y_{13}) \theta(y_{123}y_{134} - y_{13})$$
$$\cdot \theta(y_{13} + 1 - y_{123} - y_{134}) \cdot \int_0^1 dv \cdot \frac{1}{\pi} \int_0^\pi d\theta'$$
(5.99)

Here, $A_0 = Q^2/(2\pi)$, $y_{ijk} = s_{ijk}/Q^2$ and S is the symmetry factor. The variables v and θ' in Eq. (5.99) will be given in the next subsection. The lower limit of the y_{13} integration is specified by the θ functions

$$\theta(y_{13}) \,\theta(y_{13} + 1 - y_{123} - y_{134}) \to \,\theta(y_{13}) \,\theta(1 - y_{123} - y_{134}) \\ + \,\theta(y_{123} + y_{134} - 1) \,\theta(y_{13} + 1 - y_{123} - y_{134})$$
(5.100)

so the range of the y_{13} integration can be split so that

$$\int dy_{13} \to \theta (1 - y_{123} - y_{134}) \int_0^{y_{123} y_{134}} dy_{13} + \theta (y_{123} + y_{134} - 1) \int_{y_{123} + y_{134} - 1}^{y_{123} y_{134}} dy_{13} \quad (5.101)$$

Parametrizations of the four-momenta: four-particle phase space

In order to calculate the momentum mappings between (m + 1)- and *m*-particle phase space in the subtraction terms (dipoles), we need to find the explicit expressions of the four-momenta p_1 , p_2 , p_3 and p_4 in terms of the integration variables y_{12} , y_{34} , x_1 , x_3 and ϕ_3 . We can choose that \vec{p}_{12} points along the *z* axis and \vec{p}_1 lies in the *x*-*z* plane. Here we work in the CM frame, *i.e.* $\vec{p}_{12} = |\vec{p}_{12}| \hat{z}$, $\vec{p}_{34} = -|\vec{p}_{12}| \hat{z}$ and hence $\vec{p}_{12} = -\vec{p}_{34}$ is fulfilled.

$$Q = \begin{pmatrix} \sqrt{s} \\ 0 \\ 0 \\ 0 \end{pmatrix}, \quad p_1 = \frac{\sqrt{s}}{2} x_1 \begin{pmatrix} 1 \\ \sin \theta_1 \\ 0 \\ \cos \theta_1 \end{pmatrix}, \quad p_3 = \frac{\sqrt{s}}{2} x_3 \begin{pmatrix} 1 \\ \sin \theta_3 \cos \phi_3 \\ \sin \theta_3 \sin \phi_3 \\ -\cos \theta_3 \end{pmatrix}$$
(5.102)

where θ_i is the angle enclosed between \vec{p}_{ij} and \vec{p}_i , and ϕ_3 is the azimuthal angle of \vec{p}_3 with respect to the axis pointing along \vec{p}_{34} . Note that there is a minus sign in the z-component of \vec{p}_3 reflecting the fact that the z-component of \vec{p}_3 is in the direction of -z axis, while the x-component and y-component of \vec{p}_3 point along the +x axis and +y axis, respectively.

$$p_{12} = \begin{pmatrix} p_{12}^{0} \\ 0 \\ 0 \\ |\vec{p}_{12}| \end{pmatrix}, \quad p_{34} = \begin{pmatrix} p_{34}^{0} \\ 0 \\ 0 \\ -|\vec{p}_{12}| \end{pmatrix}$$
(5.103)

We can write $|\vec{p}_{12}|$ in terms of y_{ij} , *i.e.*

$$|\vec{p}_{12}| = \frac{\sqrt{s}}{2} \sqrt{\lambda(1, y_{12}, y_{34})}$$
(5.104)

Using Eq. (5.102), Eq. (5.103) and the identities

$$x_1 + x_2 = 1 - y_{34} + y_{12}, \qquad x_3 + x_4 = 1 - y_{12} + y_{34}$$
 (5.105)

we obtain

$$p_{2} = \frac{\sqrt{s}}{2} \begin{pmatrix} 1 - y_{34} + y_{12} - x_{1} \\ -x_{1} \sin \theta_{1} \\ 0 \\ \sqrt{\lambda(1, y_{12}, y_{34})} - x_{1} \cos \theta_{1} \end{pmatrix}, \quad p_{4} = \frac{\sqrt{s}}{2} \begin{pmatrix} 1 - y_{12} + y_{34} - x_{3} \\ -x_{3} \sin \theta_{3} \cos \phi_{3} \\ -x_{3} \sin \theta_{3} \sin \phi_{3} \\ -\sqrt{\lambda(1, y_{12}, y_{34})} + x_{3} \cos \theta_{3} \end{pmatrix}$$
(5.106)

Eq. (5.102) and Eq. (5.106) will be useful when we compute the dipoles. An alternative parametrizations of the four-momenta is given by $[15]^1$

$$p_{1} = \sqrt{s_{13}}/2 (1, \sin\theta \sin\theta', \sin\theta \cos\theta', \cos\theta), \qquad p_{2} = (s_{123} - s_{13})/(2\sqrt{s_{13}}) (1, 0, 0, 1)$$

$$p_{3} = \sqrt{s_{13}}/2 (1, -\sin\theta \sin\theta', -\sin\theta \cos\theta', -\cos\theta), \qquad p_{4} = (s_{134} - s_{13})/(2\sqrt{s_{13}}) (1, 0, \sin\beta, \cos\beta)$$
(5.107)

Parametrizations of the four-momenta: three-particle phase space

We will do the same thing for the momentum mappings in three-particle phase space. We need to find the explicit expressions of the four-momenta p_1 , p_2 and p_3 in terms of the integration

$$\frac{1}{2} \left(1 - \cos \beta \right) = \frac{s_{13} \left(Q^2 - s_{123} - s_{134} + s_{13} \right)}{\left(s_{123} - s_{13} \right) \left(s_{134} - s_{13} \right)}$$

¹Here, we set up the reference frame where $p_{13} = p_1 + p_3$ is at rest. We shall refer to this as the 1-3 system. The parameters v and θ are related by $v = 1/2 (1 - \cos \theta)$ and the variable $\cos \beta$ is determined by using energy-momentum conservation

variables y_{13} and y_{23} . We can choose that $\vec{p_3}$ points along the z axis and $\vec{p_1}$ lies in the x-z plane. Here we work in the CM frame, hence

$$p_{1} = \frac{\sqrt{s}}{2} x_{1} \begin{pmatrix} 1\\ \sin \theta_{13}\\ 0\\ \cos \theta_{13} \end{pmatrix}, \quad p_{2} = \frac{\sqrt{s}}{2} x_{2} \begin{pmatrix} 1\\ \sin \theta_{23} \cos \phi\\ \sin \theta_{23} \sin \phi\\ \cos \theta_{23} \end{pmatrix}, \quad p_{3} = \frac{\sqrt{s}}{2} x_{3} \begin{pmatrix} 1\\ 0\\ 0\\ 1\\ 1 \end{pmatrix}$$
(5.108)

where θ_{13} is the angle enclosed between $\vec{p_1}$ and $\vec{p_3}$, θ_{23} is the angle enclosed between $\vec{p_2}$ and $\vec{p_3}$ and ϕ is the azimuthal angle of $\vec{p_2}$ with respect to the $z \, axis^2$. We find

$$y_{12} = \frac{1}{2} x_1 x_2 (1 - \sin \theta_{13} \sin \theta_{23} \cos \phi - \cos \theta_{13} \cos \theta_{23})$$

$$y_{13} = \frac{1}{2} x_1 x_3 (1 - \cos \theta_{13}), \qquad y_{23} = \frac{1}{2} x_2 x_3 (1 - \cos \theta_{23})$$
(5.109)

Using the identity: $x_3 = y_{13} + y_{23} = 1 - y_{12}$, we then obtain

$$\cos \theta_{13} = 1 - \frac{2y_{13}}{(1 - y_{23})(y_{13} + y_{23})}, \qquad \cos \theta_{23} = 1 - \frac{2y_{23}}{(1 - y_{13})(y_{13} + y_{23})}$$
$$\cos \phi = \frac{1}{\sin \theta_{13} \sin \theta_{23}} \left[1 - \cos \theta_{13} \cos \theta_{23} - \frac{2(1 - y_{13} - y_{23})}{(1 - y_{13})(1 - y_{23})} \right]$$
(5.110)

5.5 Soft photon radiation

In this section, we will discuss the emission of a soft photon in the final state³. We will first derive the amplitude for a soft photon emitted from a single outgoing fermion; then we sum over all external fermions, obtaining the amplitude for a single soft photon emitted from all external particles. Finally we generalize to the emission of an arbitrary number of photons in the soft limit. The emission of a photon from an incoming particle (initial state radiation) can be dealt with in the same way. In this section, we only discuss final state radiation.

Final state radiation

The LO amplitude can be written as

$$\mathcal{M}_{Born}(p) = \bar{u}(p) \mathcal{A}(p) \tag{5.111}$$

$$-i e Q_f \gamma_\mu \rightarrow -i g_s t^A_{ab} \gamma_\mu$$

and hence $e Q_f \rightarrow g_s t^A_{ab}$ where t^A_{ab} are generators of $SU(3)_C$ in the fundamental representation.

²Here we have used slightly different notations for the angles.

³For the gluon radiation in QCD we just simply replace photon field with gluon field and take the colour factors and QCD coupling constant g_s into account, *i.e.* we make the following substitution for vertex:

where $\bar{u}(p)$ is the spinor of the fermion and $\mathcal{A}(p)$ the remaining part (hard part) of the amplitude. The amplitude for the emission of a photon with momentum k in final state (see Fig. 5.1) reads

$$i \mathcal{M}(p,k) = \bar{u}(p) \left[-i e Q_f \gamma_\mu \right] \varepsilon^{*\mu}(k) \frac{i(\not p + \not k + m)}{(p+k)^2 - m^2 + i \epsilon} \mathcal{A}(p+k)$$
(5.112)

In the soft-photon approximation, *i.e.* $k \rightarrow 0$, we have

$$i \mathcal{M}(p,k) \approx e Q_f \bar{u}(p) \frac{\gamma_\mu (\not p + m)}{2 \, p \cdot k + i \, \epsilon} \mathcal{A}(p) \, \varepsilon^{*\mu}(k)$$
(5.113)

Using the identity

$$\gamma_{\mu} (\not p + m) = 2 p_{\mu} - \not p \gamma_{\mu} + m \gamma_{\mu}$$
(5.114)

and Dirac equation $\bar{u}(p) (p - m) = 0$, Eq. (5.113) can be rewritten as⁴

$$i \mathcal{M}(p,k) \approx \frac{2 e Q_f p_\mu \varepsilon^{*\mu}(k)}{2 p \cdot k + i \epsilon} \bar{u}(p) \mathcal{A}(p) \approx \left[\frac{2 e Q_f p \cdot \varepsilon^{*}(k)}{2 p \cdot k + i \epsilon}\right] \mathcal{M}_{Born}(p)$$
(5.115)

For a photon emitted from an outgoing scalar particle, the result is the same, *i.e.* the result is independent of the spin of the charged particle. Spin-dependent terms are IR regular.

The emission of a photon from an incoming particle can be dealt with in the same way; for the initial state radiation the charged particle has momentum p - k instead of p + k.

If we sum over all external particles, we obtain the amplitude for the emission of a single soft photon, *i.e.* $k \rightarrow 0$

$$i \mathcal{M}(p_j, k) \stackrel{k \to 0}{\approx} \sum_{\ell} \left[\frac{2 e Q_{\ell} p_{\ell} \cdot \varepsilon^*(k)}{2 \eta_{\ell} p_{\ell} \cdot k + i \epsilon} \right] \mathcal{M}_{Born}(p_j)$$
(5.116)

where Q_{ℓ} and p_{ℓ} are the charges and momenta of the ℓ th external particle; $\eta_{\ell} = 1$ for outgoing particles (final state radiation) and $\eta_{\ell} = -1$ for incoming particles (initial state radiation). Here p_j means the emitter that emits a soft photon.

The generalization to multi-photon emissions is straightforward. Let us now consider the emission of an arbitrary number of photons; the corresponding result can be derived by induction and as a consequence, soft photons are emitted independently. The amplitude for the emission of nsoft photons with momenta k_1, k_2, \dots, k_n in the limit $k_i \to 0$ is given by

$$i \mathcal{M}(p_j, k_i) \stackrel{k_i \to 0}{\approx} \prod_{m=1}^n \left[\sum_{\ell} \frac{2 e Q_{\ell} p_{\ell} \cdot \varepsilon^*(k_m)}{2 \eta_{\ell} p_{\ell} \cdot k_m + i \epsilon} \right] \mathcal{M}_{Born}(p_j)$$
(5.117)

⁴For QCD scattering the prefactor of $\mathcal{M}_{Born}(p)$ is actually the eikonal factor, which corresponds to a soft gluon emission, *i.e.* $k \to 0$.

5.6 Collinear photon radiation

In this section, we will discuss the emission of a photon in the collinear limit⁵; we will derive the matrix elements of the emission of a collinear photon for both final state and initial state radiations.

5.6.1 Final state radiation

The Born matrix element is given by (see Fig. 5.2)

$$\left|\mathcal{M}_{Born}(p)\right|^{2} = \left|\bar{u}(p)\mathcal{A}(p)\right|^{2} = \overline{\mathcal{A}}(p)u(p)\bar{u}(p)\mathcal{A}(p) = \overline{\mathcal{A}}(p)\left(\not p + m\right)\mathcal{A}(p)$$
(5.118)

where the projection operator is

$$\sum_{\text{spins}} u(p) \,\bar{u}(p) = \not p + m \tag{5.119}$$

and $\bar{u}(p)$ is the spinor of the fermion; $\mathcal{A}(p)$ the remaining part (hard part) of the amplitude. Here the spin indices have been suppressed and sum over spins (polarizations) is also implicitly understood. Let us now consider the emission of a collinear photon with momentum k (see Fig. 5.3). The corresponding matrix element is given by

$$\begin{aligned} \left|\mathcal{M}(p',k)\right|^{2} &= \left|\bar{u}(p')\left[-i\,e\,Q_{f}\,\gamma_{\mu}\right]\,\varepsilon^{*\mu}(k)\,\frac{i(\not\!p+m)}{p^{2}-m^{2}+i\,\epsilon}\,\mathcal{A}(p)\right|^{2} \\ &= e^{2}\,Q_{f}^{2}\,\left|\bar{u}(p')\,\gamma_{\nu}\,\varepsilon^{*\nu}(k)\,\frac{(\not\!p+m)}{p^{2}-m^{2}+i\,\epsilon}\,\mathcal{A}(p)\right|^{2} \\ &= \frac{e^{2}\,Q_{f}^{2}}{(p^{2}-m^{2})^{2}}\,\left|\bar{u}(p')\,\gamma_{\mu}\,\varepsilon^{*\mu}(k)\,(\not\!p+m)\,\mathcal{A}(p)\right|^{2} \\ &= \frac{e^{2}\,Q_{f}^{2}}{(p^{2}-m^{2})^{2}}\,\overline{\mathcal{A}}(p)\,(\not\!p+m)\,\gamma_{\nu}\,\varepsilon^{\nu}(k)\,u(p')\,\bar{u}(p')\,\gamma_{\mu}\,\varepsilon^{*\mu}(k)\,(\not\!p+m)\,\mathcal{A}(p) \\ &= \frac{e^{2}\,Q_{f}^{2}}{(p^{2}-m^{2})^{2}}\,\overline{\mathcal{A}}(p)\,(\not\!p+m)\,\gamma_{\nu}\,u(p')\,\bar{u}(p')\,\gamma_{\mu}\,(\not\!p+m)\,\mathcal{A}(p)\,\varepsilon^{*\mu}(k)\,\varepsilon^{\nu}(k) \\ &= \frac{e^{2}\,Q_{f}^{2}}{(p^{2}-m^{2})^{2}}\,\overline{\mathcal{A}}(p)\,(\not\!p+m)\,\gamma_{\nu}\,(\not\!p'+m)\,\gamma_{\mu}\,(\not\!p+m)\,\mathcal{A}(p)\,d^{\mu\nu}(k) \end{aligned} \tag{5.120}$$

⁵Similarly for the gluon radiation we replace photon field with gluon field and make the following substitution for vertex:

$$i \, e \, Q_f \, \gamma_\mu \,
ightarrow - i \, g_s \, t^A_{ab} \, \gamma_\mu$$

and hence

$$e^2 Q_f^2 \rightarrow g_s^2 t_{ad}^A t_{db}^A$$

where we have used Eq. (5.119) and

$$d^{\mu\nu}(k) = \sum_{\text{pol}} \varepsilon^{*\mu}(k) \,\varepsilon^{\nu}(k) = -g^{\mu\nu} + \frac{k^{\mu}n^{\nu} + k^{\nu}n^{\mu}}{k \cdot n}$$
(5.121)

again the spin indices have been suppressed and sum over spins (polarizations) is also implicitly understood. The four vector n and $d^{\mu\nu}(k)$ satisfy the light-cone gauge conditions

$$n^{2} = 0, \quad -g^{\mu\nu} d_{\mu\nu}(k) = d - 2 = 2 - 2\epsilon, \quad k^{\mu} d_{\mu\nu}(k) = k^{\nu} d_{\mu\nu}(k) = 0$$

$$n^{\mu} d_{\mu\nu}(k) = n^{\nu} d_{\mu\nu}(k) = 0$$
(5.122)

Kinematics

We need a bit kinematics for the massless photon and massive fermions

$$p = p' + k$$

$$k^{2} = 0, \quad p'^{2} = m^{2}; \quad p^{2} \longrightarrow \mathcal{O}(m^{2}) \quad \text{in collinear limit}$$

$$k^{\mu} = z p^{\mu} + k^{\mu}_{\perp} + k^{\mu}_{r}, \quad z := k^{0}/p^{0}, \quad p \cdot k_{\perp} = 0, \quad \vec{k}_{r} = \vec{0}$$

$$p'^{\mu} = p^{\mu} - k^{\mu} = (1 - z) p^{\mu} - k^{\mu}_{\perp} - k^{\mu}_{r} \qquad (5.123)$$

From Eq. (5.123), we obtain immediately

$$k_r^0 = -k_\perp^0, \quad \vec{k}_\perp^2 = \mathcal{O}(m^2), \quad k_\perp^0 = \mathcal{O}(m^2/p^0)$$
 (5.124)

The matrix element with photon emission now reads

$$\left|\mathcal{M}(p',k)\right|^{2} = \frac{e^{2} Q_{f}^{2}}{(2 \, p' \cdot k)^{2}} \,\overline{\mathcal{A}}(p) \left(p' + k + m\right) \gamma_{\nu} \left(p' + m\right) \gamma_{\mu} \left(p' + k + m\right) \mathcal{A}(p) \, d^{\mu\nu}(k) \quad (5.125)$$

Using the identity

$$(p' + k + m) \gamma_{\nu} (p' + m) = k (-p' + m) \gamma_{\nu} + 2 p'_{\nu} (p' + k + m)$$
(5.126)

hence

Using the identities

$$\gamma_{\nu} \gamma_{\mu} d^{\mu\nu} = g_{\mu\nu} d^{\mu\nu} = -d+2$$
(5.128)

and

•

$$(p' + k + m) \gamma_{\mu} = 2 p'_{\mu} + 2 k_{\mu} + \gamma_{\mu} (-p' - k + m)$$
(5.129)

Hence we have

$$\begin{aligned} \left|\mathcal{M}(p',k)\right|^{2} &= \frac{e^{2}Q_{f}^{2}}{(2\,p'\cdot k)^{2}}\,\overline{\mathcal{A}}(p)\,\left[-2\,p'\cdot k\,\,kg_{\mu\nu} + 4\,p'_{\nu}\,p'_{\mu}\,(p'+k+m) - 4\,p'_{\nu}\,\gamma_{\mu}\,p'\cdot k\right]\,\mathcal{A}(p)\,d^{\mu\nu}(k) \\ &= \frac{e^{2}\,Q_{f}^{2}}{(2\,p'\cdot k)}\,\overline{\mathcal{A}}(p)\,\left[2\,(1-\epsilon)\,k-\frac{2\,m^{2}}{p'\cdot k}\,(p'+k+m) + 4\,\frac{p'\cdot n}{k\cdot n}\,(p'+k+m) \right. \\ &\left. + 2\,p'-2\,\frac{k\,p'\cdot n+p'\cdot k\,\,t}{k\cdot n}\right]\,\mathcal{A}(p) \end{aligned} \tag{5.130}$$

In high-energy limit, a charged fermion energy p^0 is much larger than its mass m and hence we can neglect the fermion mass. Now we consider the collinear limit: $k_r^{\mu} = k_{\perp}^{\mu} = 0$ and $p^2 \longrightarrow \mathcal{O}(m^2)$

$$k \approx z \not p$$
, $p' \approx (1-z) \not p$, $\frac{m^2}{2 p' \cdot k} = \mathcal{O}(1)$, $\frac{p' \cdot n}{k \cdot n} = \frac{1}{z} - 1 + \mathcal{O}(m/p^0)$ (5.131)

The photon emission matrix element in the collinear limit now becomes

$$\left|\mathcal{M}(p',k)\right|^{2} \approx \frac{e^{2} Q_{f}^{2}}{2 p' \cdot k} \overline{\mathcal{A}}(p) \left[2 \left(1-\epsilon\right) z \not p - \frac{2 m^{2}}{p' \cdot k} \not p + 4 \left(\frac{1}{z}-1\right) \not p + 2 \left(1-z\right) \not p - 2 z \not p \left(\frac{1}{z}-1\right) - 2 \frac{p' \cdot k}{k \cdot n} \not n \right] \mathcal{A}(p)$$

$$(5.132)$$

The last term in Eq. (5.132) vanishes since $p' \cdot k \approx \mathcal{O}(m^2)$. Finally we obtain

$$\begin{aligned} \left|\mathcal{M}(p',k)\right|^2 &\approx \frac{e^2 Q_f^2}{2 \, p' \cdot k} \,\overline{\mathcal{A}}(p) \,\left[2 \left(1-\epsilon\right) z \, \not{p} - \frac{2 \, m^2}{p' \cdot k} \, \not{p} + 4 \,\left(\frac{1-z}{z}\right) \, \not{p}\right] \, \mathcal{A}(p) \\ &= \frac{e^2 \, Q_f^2}{p' \cdot k} \,\overline{\mathcal{A}}(p) \,\left[\left(1-\epsilon\right) z \, - \frac{m^2}{p' \cdot k} + 2 \, \left(\frac{1-z}{z}\right)\right] \, \not{p} \, \mathcal{A}(p) \\ &= \frac{e^2 \, Q_f^2}{p' \cdot k} \,\overline{\mathcal{A}}(p) \,\left[-\epsilon \, z \, - \frac{m^2}{p' \cdot k} + \frac{1+(1-z)^2}{z}\right] \, \not{p} \, \mathcal{A}(p) \\ &= \frac{e^2 \, Q_f^2}{p' \cdot k} \,\left[P_{\gamma f}(z) - \frac{m^2}{p' \cdot k}\right] \, \left|\mathcal{M}_{Born}(p)\right|^2 \end{aligned} \tag{5.133}$$

where we have introduced the photon splitting function⁶ in d dimension

$$P_{\gamma f}(z) := \frac{1 + (1 - z)^2}{z} - \epsilon z$$
(5.134)

```
z\,\rightarrow\,1-x
```

⁶For the gluon radiation in QCD, this is exactly the Altarelli-Parisi splitting function $P_{qq}(x)$ with appropriate substitution [6]

5.6.2 Initial state radiation

The Born matrix element is given by (see Fig. 5.4)

$$\left|\mathcal{M}_{Born}(p)\right|^2 = \overline{\mathcal{A}}(p)\left(\not p + m\right)\mathcal{A}(p) \tag{5.135}$$

Kinematics

We need a bit kinematics for the photon emission in the initial state radiation (see Fig. 5.5)

$$x := \frac{k^0}{p^0}, \quad z = \frac{k^0}{p'^0}, \quad p = p' + k$$
 (5.136)

Substitutions from final state radiation, then we have

$$k \to -k, \quad p \leftrightarrow p'$$
 (5.137)

and

$$z \to -z = \frac{x}{x-1}, \quad P_{\gamma f}(-z) = \frac{1+(1+z)^2}{-z} + \epsilon z = \frac{1}{x-1}P_{\gamma f}(x)$$
 (5.138)

Hence the matrix element for an emission of a collinear photon with momentum k is given by

$$|\mathcal{M}(p,k)|^2 \approx \frac{e^2 Q_f^2}{p \cdot k} \frac{1}{1-x} \left[P_{\gamma f}(x) - (1-x) \frac{m^2}{p \cdot k} \right] \left| \mathcal{M}_{Born}((1-x) p) \right|^2$$
(5.139)

with

$$p' \approx (1-x) p, \quad k \approx -x p$$
 (5.140)

in the collinear limit.

5.7 One-loop calculations: examples

The quark self-energy contributions

For simplicity we consider the Feynman gauge in which $\alpha = 1$; so the expression for the quark self-energy term (see Fig. 5.6) to order α_s in the dimensional regularization scheme is⁷

$$-i\Sigma_{ab}(p) = \mu^{\epsilon} \int \frac{d^{d}k}{(2\pi)^{d}} \frac{-ig_{\mu\nu}}{k^{2} + i\epsilon} \left(-ig_{s} t_{ad}^{C} \gamma^{\mu}\right) \frac{i}{(\not p - \not k) - m_{q} + i\epsilon} \left(-ig_{s} t_{db}^{C} \gamma^{\mu}\right)$$
$$= -g_{s}^{2} \mu^{\epsilon} \int \frac{d^{d}k}{(2\pi)^{d}} \frac{g_{\mu\nu}}{k^{2} + i\epsilon} \gamma^{\mu} \frac{1}{(\not p - \not k) - m_{q} + i\epsilon} \gamma^{\nu} t_{ad}^{C} t_{db}^{C}$$
$$= -i\Sigma_{\text{QED}} t_{ad}^{C} t_{db}^{C}$$
(5.141)

⁷Here we work in $d = 4 - \epsilon$ dimensions.



Figure 5.1: Final state radiation: soft photon emission.



Figure 5.2: Final state radiation: Born diagram.



Figure 5.3: Final state radiation: collinear photon emission.



Figure 5.4: Initial state radiation: Born diagram.



Figure 5.5: Initial state radiation: collinear photon emission.



Figure 5.6: The one-loop contribution to the quark self-energy diagram in QCD.



Figure 5.7: The one-loop contribution to the gluon vertex diagram in QCD.

where

$$\sum_{A} t_{ab}^{A} t_{bc}^{A} = C_{F} \delta_{ac}, \quad C_{F} = \frac{N^{2} - 1}{2N}$$
(5.142)

and

$$-i\Sigma_{\text{QED}} = -g_s^2 \,\mu^\epsilon \,\int \frac{d^d k}{(2\,\pi)^d} \frac{\gamma_\mu \,[\not\!p - \not\!k + m_q] \,\gamma^\mu}{(k^2 + i\epsilon) \,[(p-k)^2 - m_q^2 + i\epsilon]}$$
(5.143)

Here we consider the massless and on mass-shell quark (antiquark), so we have

$$-i\Sigma_{\text{QED}} = -g_s^2 \mu^{\epsilon} \int \frac{d^d k}{(2\pi)^d} \frac{\gamma_{\mu} \left[\not p - \not k\right] \gamma^{\mu}}{(k^2 + i\epsilon) \left[(p - k)^2 + i\epsilon\right]}$$
(5.144)

The infrared divergence (mass singularity) comes from the massless quark and antiquark. After some tedious calculations, we find

$$-i\Sigma_{\text{QED}} = -ig_s^2 \mu^{\epsilon} (2-d) \not p \int_0^1 dz (1-z) \int \frac{d^d k'}{(2\pi)^d} \frac{1}{i} \frac{1}{(k')^4}$$

$$\approx \frac{1}{\epsilon_{UV}} - \frac{1}{\epsilon_{IR}} = 0$$

$$\uparrow$$

$$\epsilon_{UV} = \epsilon_{IR}$$
(5.145)

where k' = k - p z and we have used the scaleless integral

$$\int d^d k \, \frac{1}{\left[k^2\right]^n} \propto \frac{1}{\epsilon_{UV}} - \frac{1}{\epsilon_{IR}} = 0 \tag{5.146}$$

Here the ϵ_{UV} and ϵ_{IR} poles are used to regularize the ultraviolet and infrared divergences, respectively. So the final result is independent of any scale ($p^2 = 0$). Eq. (5.146) is very useful as long as we do not specifically care about the pole coefficients.

The vertex corrections

In this section we compute the order of α_s vertex corrections (see Fig. 5.7) to the quark and antiquark annihilation in QCD. Applying the Feynman rules, we find $\Gamma^{\mu} = \gamma^{\mu} + \delta \Gamma^{\mu}$, where

$$\delta\Gamma^{\mu}(p',p) = \mu^{\epsilon} \int \frac{d^{d}k}{(2\pi)^{d}} \left[\frac{-i g_{\nu\rho}}{(k-p)^{2} + i\epsilon} \right] \left[-i g_{s} \gamma^{\nu} t_{cj}^{A} \right] \frac{i}{\not{k}' - m_{\bar{q}} + i\epsilon} \\ \times \left[\frac{1}{\sqrt{2}} \gamma^{\mu} \frac{1}{2} \left(1 - \gamma^{5} \right) V_{ji} \right] \frac{i}{\not{k} - m_{q} + i\epsilon} \left[-i g_{s} \gamma^{\rho} t_{ib}^{A} \right]$$
(5.147)

Assume that $m_q \simeq m_{ar{q}} \,=\, 0,$ then $\delta \Gamma^\mu$ becomes

$$\delta\Gamma^{\mu}(p',p) = -i g_s^2 \frac{\mu^{\epsilon}}{\sqrt{2}} \left(t_{cj}^A V_{ji} t_{ib}^A \right) \times I^{\mu}$$
(5.148)

where

$$I^{\mu} = \int \frac{d^{d}k}{(2\pi)^{d}} \frac{\gamma^{\nu}(\not{k} + \not{q})\gamma^{\mu}\frac{1}{2}(1-\gamma^{5})\not{k}\gamma_{\nu}}{(k-p)^{2}(k+q)^{2}k^{2}} = \int \frac{d^{d}k}{(2\pi)^{d}} \frac{[\gamma^{\nu}\not{k}\gamma^{\mu}\not{k}\gamma_{\nu} + \gamma^{\nu}\not{q}\gamma^{\mu}\not{k}\gamma_{\nu}]}{(k-p)^{2}(k+q)^{2}k^{2}} \frac{1}{2}(1-\gamma^{5})$$
(5.149)

where k' = k + q. The gamma matrices in $d = 4 - \epsilon$ dimensions satisfy

$$\gamma^{\nu} \not k \gamma^{\mu} \not k \gamma_{\nu} = (2-d) \not k \gamma^{\mu} \not k \quad \text{and} \quad \gamma^{\nu} \not q \gamma^{\mu} \not k \gamma_{\nu} = -2 \not k \gamma^{\mu} \not q - (d-4) \not q \gamma^{\mu} \not k \quad (5.150)$$

So I^{μ} becomes

$$I^{\mu} = -2 \int \frac{d^{d}k}{(2\pi)^{d}} \frac{\left[\frac{d-2}{2} \not k \gamma^{\mu} \not k + \not k \gamma^{\mu} \not q + \frac{d-4}{2} \not q \gamma^{\mu} \not k\right]}{(k-p)^{2} (k+q)^{2} k^{2}} \frac{1}{2} \left(1 - \gamma^{5}\right)$$
(5.151)

Using the Feynman parametrization prescription, the denominator can be expressed as

$$\frac{1}{(k-p)^2 (k+q)^2 k^2} = \int_0^1 dx \, dy \, dz \, \delta(x+y+z-1) \frac{2}{D^3}$$
(5.152)

where

$$D = \ell^2 - \Delta + i\epsilon, \quad \ell = k + yq - zp \quad \text{and} \quad \Delta \equiv -xyq^2$$
 (5.153)

or more generally

$$\frac{1}{A_1^{n_1}A_2^{n_2}\cdots A_k^{n_k}} = \frac{\Gamma(n_1+n_2+\cdots+n_k)}{\Gamma(n_1)\Gamma(n_2)\cdots\Gamma(n_k)} \int_0^1 d\alpha_1\cdots d\alpha_k \frac{\alpha_1^{n_1-1}\alpha_2^{n_2-1}\cdots\alpha_k^{n_k-1}\delta\left(1-\sum_i\alpha_i\right)}{(\alpha_1A_1+\cdots+\alpha_kA_k)^{n_1+n_2+\cdots+n_k}}$$
(5.154)

Now let us calculate the numerator of Eq. (5.151)

$$\bar{v}(p') \left[\frac{d-2}{2} \not\!\!\!\! k \gamma^{\mu} \not\!\!\! k + \not\!\!\!\! k \gamma^{\mu} \not\!\!\! q + \frac{d-4}{2} \not\!\!\! q \gamma^{\mu} \not\!\!\! k \right] \frac{1}{2} \left(1 - \gamma^5 \right) u(p) \tag{5.155}$$

after some tedious calculations, we obtain

$$\bar{v}(p') \left[-\frac{(2-d)^2}{2d} \ell^2 + (1-x) \left(1 - y \frac{d-2}{2} \right) q^2 + \frac{d-4}{2} y q^2 \right] \gamma^\mu \frac{(1-\gamma^5)}{2} u(p) \quad (5.156)$$

where we have used

Finally, we have to compute the following expression

$$\bar{v}(p') I^{\mu} u(p) = -2 \int \frac{d^{d}\ell}{(2\pi)^{d}} \int_{0}^{1} dx \, dy \, dz \, \frac{2\,\delta(x+y+z-1)}{(\ell^{2}-\Delta+i\epsilon)^{3}} \times \left[\bar{v}(p')\,\gamma^{\mu}\,\frac{(1-\gamma^{5})}{2}\,u(p)\right] \\ \times \left[-\frac{(2-d)^{2}}{2\,d}\,\ell^{2} + (1-x)\,\left(1-y\,\frac{d-2}{2}\right)\,q^{2} + \frac{d-4}{2}\,y\,q^{2}\right] \\ = \bar{v}(p')\,\gamma^{\mu}\,\frac{(1-\gamma^{5})}{2}\,u(p) \times A(q^{2})$$
(5.158)

We can divide $A(q^2)$ into two terms $A = A^a(q^2) + A^b(q^2)$. In order to calculate $A(q^2)$, we could resort to the following integration formulae,

$$\int \frac{d^{d}\ell}{(2\pi)^{d}} \frac{\ell^{2}}{(\ell^{2} - \triangle + i\epsilon)^{3}} = \frac{i}{4} \frac{d}{(4\pi)^{d/2}} \Gamma(2 - d/2) \Delta^{d/2 - 2}$$

$$\int \frac{d^{d}\ell}{(2\pi)^{d}} \frac{1}{(\ell^{2} - \triangle + i\epsilon)^{3}} = \frac{1}{2} \frac{-i}{(4\pi)^{d/2}} (2 - d/2) \Gamma(2 - d/2) \Delta^{d/2 - 3}$$
(5.159)

hence we have

$$A^{a}(q^{2}) = \frac{i}{(4\pi)^{d/2}} (-q^{2})^{d/2-2} \frac{\Gamma(d/2-2)\Gamma(d/2)}{\Gamma(d-2)} \Gamma(2-d/2) (d/2-2)$$

$$A^{b}(q^{2}) = -2 \frac{i}{(4\pi)^{d/2}} (-q^{2})^{d/2-2} (2-d/2) \Gamma(2-d/2) \frac{\Gamma(d/2-2)\Gamma(d/2)}{\Gamma(d-2)}$$

$$\times \left[-\frac{2}{\epsilon} - \frac{d}{(d-2)^{2}} \frac{d-2}{2} + \frac{d-4}{d-2} \right]$$
(5.160)

Finally we obtain the expression for $\bar{v}(p') \, \delta \Gamma^{\mu} \, u(p)$:

$$\bar{v}(p')\,\delta\Gamma^{\mu}\,u(p) = -i\,g_{s}^{2}\,\frac{\mu^{\epsilon}}{\sqrt{2}}\left(t_{cj}^{A}\,V_{ji}\,t_{ib}^{A}\right)\times\bar{v}(p')\,\gamma^{\mu}\,\frac{(1-\gamma^{5})}{2}\,u(p)\times$$

$$\left(-q^{2}\right)^{d/2-2}\,\frac{i}{(4\,\pi)^{d/2}}\,\frac{1}{\Gamma(1-\epsilon)}\,\frac{\Gamma(2-\epsilon/2)}{1-\epsilon}\Gamma(\epsilon/2)\Gamma(-\epsilon/2)\times$$

$$\left\{-\frac{\epsilon}{2}-2\,\frac{\epsilon}{2}\,\left[-\frac{2}{\epsilon}+\frac{\epsilon-4}{2\,(2-\epsilon)}-\frac{\epsilon}{2-\epsilon}\right]\right\}$$

$$=\frac{\alpha_{s}}{4\,\pi}\left(\frac{4}{3}\right)\left(\frac{4\pi\mu^{2}}{q^{2}}\right)^{\frac{\epsilon}{2}}\times\bar{v}(p')\,\frac{1}{\sqrt{2}}\,\gamma^{\mu}\,\frac{(1-\gamma^{5})}{2}\,u(p)\times$$

$$\frac{\Gamma(1+\frac{\epsilon}{2})\,\Gamma^{2}(1-\frac{\epsilon}{2})}{\Gamma(1-\epsilon)}\left(-\frac{8}{\epsilon^{2}}-\frac{6}{\epsilon}-8+\pi^{2}+\mathcal{O}(\epsilon)\right)$$
(5.161)

5.8 A review of the Standard Model (SM)

5.8.1 Abelian gauge theory: Quantum electrodynamics (QED)

We start with a Lagrangian

$$\mathcal{L}_{0} = \overline{\psi}(x) \left(i\partial \!\!\!/ - m\right) \psi(x) \tag{5.162}$$

which is invariant under a global U(1) symmetry:

$$\psi(x) \rightarrow \psi'(x) = e^{iq\theta}\psi(x)$$

$$\partial_{\mu}\psi(x) \rightarrow \partial_{\mu}\psi'(x) = e^{iq\theta}\partial_{\mu}\psi(x)$$
(5.163)

with spacetime independent group parameter θ . There is a conserved current according to Noether's theorem:

$$J_{\mu}(x) = q \overline{\psi}(x) \gamma_{\mu} \psi(x) \implies \partial^{\mu} J_{\mu}(x) = 0$$
(5.164)

In the case of quantum electrodynamics, the phase invariance is promoted to the level of a local transformation in order to describe the gauge interactions between electrons and photons, *i.e.* the phase θ depends on the spacetime point. So we demand the global U(1) symmetry to local symmetry, this means

$$\theta \to \theta(x)$$
 (5.165)

so

$$\psi(x) \rightarrow \psi'(x) = e^{iq\theta(x)}\psi(x)$$

$$\partial_{\mu}\psi(x) \rightarrow \partial_{\mu}\psi'(x) = e^{iq\theta(x)}\partial_{\mu}\psi(x) + iq e^{iq\theta(x)}\psi(x)\partial_{\mu}\theta(x)$$
(5.166)

To maintain the local gauge invariance, we introduce the covariant derivative.

$$D_{\mu} = \partial_{\mu} + i q A_{\mu}(x) \tag{5.167}$$

in such a way that

$$\psi(x) \rightarrow \psi'(x) = e^{iq\,\theta(x)}\psi(x) = U(x)\psi(x)$$

$$D_{\mu}\psi(x) \rightarrow D'_{\mu}\psi'(x) = e^{iq\,\theta(x)}D_{\mu}\psi(x) = U(x)D_{\mu}\psi(x)$$
(5.168)

i.e. both $\psi(x)$ and $D_{\mu}\psi(x)$ transform the same way under U(1) local symmetry. A^{μ} is the spin 1 gauge field (photon field) and transforms under the local gauge symmetry as

$$A_{\mu}(x) \to A'_{\mu}(x) = A_{\mu}(x) - \partial_{\mu}\theta(x)$$
(5.169)

Note that the transformation property ensures that

- $D_{\mu}\psi(x) \rightarrow D'_{\mu}\psi'(x) = U(x)D_{\mu}\psi(x)$
- gauge invariance of field strength tensor $F_{\mu\nu} = \partial_{\mu}A_{\nu} \partial_{\nu}A_{\mu}$

The commutator of covariant derivatives obeys

$$F_{\mu\nu} = \frac{1}{iq} \left[D_{\mu}, D_{\nu} \right] = \frac{1}{iq} \left[\partial_{\mu} + iqA_{\mu}, \partial_{\nu} + iqA_{\nu} \right] = \partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu}$$
(5.170)

Then obviously the generalized Lagrangian

$$\mathcal{L}_{\psi} = \overline{\psi}(x) \left(i\gamma^{\mu} D_{\mu} - m \right) \psi(x) = \overline{\psi}(x) \left(i\gamma^{\mu} \partial_{\mu} - m \right) \psi(x) - q \,\overline{\psi}(x) \,\gamma_{\mu} \,\psi(x) \,A^{\mu}$$
(5.171)

is invariant under the local gauge transformations. The complete QED Lagrangian has two contributions: matter and gauge field contributions:

$$\mathcal{L} = \mathcal{L}_{\psi} + \mathcal{L}_{\text{gauge}} \tag{5.172}$$

with

$$\mathcal{L}_{\text{gauge}} = -\frac{1}{4} F_{\mu\nu}(x) F^{\mu\nu}(x)$$
 (5.173)

The electron-photon coupling in Eq. (5.171) is called minimal coupling of the photon field A_{μ} to the electromagnetic current $J^{\mu} = q \overline{\psi}(x) \gamma^{\mu} \psi(x)$; and \mathcal{L}_{gauge} cannot contain a term proportional to

 $A_{\mu}A^{\mu}$

since this term is not gauge invariant under Eq. (5.169).

5.8.2 Non-abelian gauge theory

The starting point is a Lagrangian of free or self-interacting fields, *i.e.* symmetric under a global symmetry.

$$\mathcal{L}_{\psi}(\psi, \partial_{\mu}\psi) = \overline{\psi}(x) \left(i\partial \!\!\!/ - m\right)\psi(x) \tag{5.174}$$

where

$$\psi = \begin{pmatrix} \psi_1 \\ \vdots \\ \psi_n \end{pmatrix} = \text{multiplet of a compact Lie group } G$$
(5.175)

The Lagrangian is symmetric under the transformation

$$\psi \to \psi' = U(\theta)\psi, \quad U(\theta) = \exp\{it^A \theta^A\}, \quad U^{\dagger}U = UU^{\dagger} = 1$$
 (5.176)

If U is unitary, the t^A are hermitian matrices, called group generators. θ^A ($A = 1, \dots, N^2 - 1 = \dim G$) are spacetime independent parameters. We can expand the unitary matrix around the unit element of the group G

$$U(\theta) = 1 + i t^A \theta^A + \mathcal{O}(\theta^2)$$
(5.177)

this means that the t^A generate infinitesimal transformation around the unit matrix element of the group. For SU(N) matrix (unitary and det U = 1), there are $N^2 - 1$ traceless, hermitian generators

$$t^A = \frac{\lambda^A}{2} \tag{5.178}$$

The generators for any representation of G satisfy the Lie Algebra relation

$$\left[t^A, t^B\right] = i f^{ABC} t^C \tag{5.179}$$

where the f^{ABC} are called the structure constants of the group G. The starting hypothesis is that Lagrangian is invariant under G

$$\mathcal{L}_{\psi}(\psi, \partial_{\mu}\psi) = \mathcal{L}_{\psi}(\psi', \partial_{\mu}\psi'), \qquad \psi' = U(\theta)\psi$$
(5.180)

Now we promote the global symmetry to local symmetry by gauging the theory, which means that we allow the parameters θ^A to be function of the spacetime coordinates.

$$\theta^A \to \theta^A(x) \implies U \to U(x)$$
 (5.181)

so now we have

$$U(x) = 1 + i t^{A} \theta^{A}(x) + \mathcal{O}(\theta^{2})$$
(5.182)

We obtain a local invariant Lagrangian if we make the substitution

$$\mathcal{L}_{\psi}(\psi, \partial_{\mu}\psi) \to \mathcal{L}_{\psi}(\psi, D_{\mu}\psi) \tag{5.183}$$

with

$$D_{\mu} = \partial_{\mu} + i g A^{A}_{\mu}(x) t^{A} \equiv \partial_{\mu} + i g A_{\mu}(x), \qquad A_{\mu} \equiv \sum_{A} A^{A}_{\mu} t^{A} = A_{\mu} \cdot t$$
(5.184)

where g is the gauge coupling, t^A is the generator of the group G in ψ representation and A^A_{μ} are gauge fields. Here A_{μ} is an $N \times N$ matrix. $\mathcal{L}_{\psi}(\psi, D_{\mu}\psi)$ is local gauge invariant if

$$\psi(x) \rightarrow \psi'(x) = U(x)\psi(x)$$

$$D_{\mu}\psi(x) \rightarrow D'_{\mu}\psi'(x) = U(x)D_{\mu}\psi(x) = U(x)D_{\mu}U^{-1}(x)U(x)\psi(x)$$
(5.185)

i.e. the covariant derivative transforms as

$$D_{\mu} \to U(x) D_{\mu} U^{-1}(x)$$
 (5.186)

implying that

$$A^A_\mu \to A^{\prime A}_\mu = A^A_\mu - f^{ABC} \theta^B A^C_\mu - \frac{1}{g} \partial_\mu \theta^A + \mathcal{O}(\theta^2)$$
(5.187)

We can build the kinetic term for the A^A_μ fields from the field strength:

$$[D_{\mu}, D_{\nu}] = i g t \cdot F_{\mu\nu} = i g t^{A} F_{\mu\nu}^{A} = i g F_{\mu\nu}, \quad F_{\mu\nu} \equiv t^{A} F_{\mu\nu}^{A}$$
(5.188)

Here $F_{\mu\nu}$ is an $N \times N$ matrix and

$$F^A_{\mu\nu} = \partial_\mu A^A_\nu - \partial_\nu A^A_\mu - g f^{ABC} A^B_\mu A^C_\nu$$
(5.189)

which transforms homogeneously under a local gauge transformation

$$F_{\mu\nu} \to F'_{\mu\nu} = U F_{\mu\nu} U^{-1}$$
 (5.190)

We also note that Tr $F_{\mu\nu}$ $F^{\mu\nu}$ is invariant:

$$F^{A}_{\mu\nu}F^{\mu\nu}_{A} \sim \text{Tr} F_{\mu\nu}F^{\mu\nu} \to \text{Tr} UF_{\mu\nu}U^{-1}UF^{\mu\nu}U^{-1} = \text{Tr} F_{\mu\nu}F^{\mu\nu}$$
(5.191)

This is true only for finite dimensional representation of the gauge group. Now we can construct the gauge invariant Lagrangian for gauge and matter fields,

$$\mathcal{L}_{YM} = -\frac{1}{2} \operatorname{Tr} F_{\mu\nu}(x) F^{\mu\nu}(x) + \overline{\psi}(x) \left(i \not\!\!D - m\right) \psi(x)$$
(5.192)

Normalizing the generators t^A as

$$\operatorname{Tr} t^A t^B = \frac{1}{2} \delta^{AB} \tag{5.193}$$

we have $-\frac{1}{2} \operatorname{Tr} F_{\mu\nu} F^{\mu\nu} = -\frac{1}{4} F^A_{\mu\nu} F^{\mu\nu}_A$. As in the abelian case, the fermion/gauge field coupling is of the form

$$\mathcal{L}_{\rm int} \sim g J^{\mu}_A A^A_{\mu} \tag{5.194}$$

where $J^A_{\mu} = \overline{\psi} \gamma_{\mu} t^A \psi$ is the fermionic part of the Noether current. There are some remarks about the non-abelian gauge theory:

- $A^A_\mu A^\mu_A$ are not gauge invariant, this means that gauge bosons of unbroken non-abelian gauge theories are massless.
- We have cubic and quartic gauge boson self-interactions due to $F^A_{\mu\nu}F^{\mu\nu}_A$

$$(\partial A)A^2, \quad A^4$$

- Gauge invariance + renormalizability (absence of higher powers of fields and covariant derivatives in Lagrangian) determine gauge boson/matter couplings and gauge boson self-interactions.
- If $G = SU(3)_C$ and the fermions are in triplets,

$$\psi = \begin{pmatrix} \psi_{\text{red}} \\ \psi_{\text{blue}} \\ \psi_{\text{green}} \end{pmatrix} = \begin{pmatrix} \psi_1 \\ \psi_2 \\ \psi_3 \end{pmatrix}$$
(5.195)

we have the QCD Lagrangian with $3^2 - 1 = 8$ gluons.

5.8.3 Electroweak theory

Gauge boson interactions

The standard electroweak theory [76–78] is based on the more complicated gauge group $SU(2) \times U(1)$. Here we have suppressed the indices L and Y for SU(2) and U(1), which mean lefthanded structure and weak hypercharge, respectively. Essentially, an SU(2) gauge symmetry is applied to left-handed fermion fields only and an independent U(1) gauge symmetry is present in order to incorporate the electric charge Q and unify the weak and electromagnetic interactions in a common gauge structure. Initially the Lagrangian of this model contains three massless gauge bosons A^a_{μ} (a = 1, 2, 3) of SU(2) gauge group and one massless gauge boson, B_{μ} , associated with the U(1) gauge group. The gauge symmetry does not allow any mass term for W and Z bosons. More precisely, local gauge invariance and renormalizability completely determine the kinetic terms for the gauge bosons. The Lagrangian of the gauge bosons is

$$\mathcal{L}_{\text{gauge}} = -\frac{1}{4} F^a_{\mu\nu} F^{\mu\nu}_a - \frac{1}{4} B_{\mu\nu} B^{\mu\nu}$$
(5.196)

The field strength tensors of the SU(2) gauge fields A^a and the U(1) gauge field B are

$$F^{a}_{\mu\nu} = \partial_{\mu}A^{a}_{\nu} - \partial_{\nu}A^{a}_{\mu} + g\epsilon^{abc}A^{b}_{\mu}A^{c}_{\nu}$$

$$B_{\mu\nu} = \partial_{\mu}B_{\nu} - \partial_{\nu}B_{\mu}$$
(5.197)

g being the SU(2) gauge coupling. Note that the vector bosons A^a already have self-interactions because of the non-abelian property of their symmetry group SU(2). This is similar to the fact that gluons carry colour charge in QCD. The coupling of the gauge fields to matter fields is achieved via the covariant derivative

$$D_{\mu}\psi = \left(\partial_{\mu} - i\,g\,T^a\,A^a_{\mu} - i\,g'\,Y_{\psi}\,B_{\mu}\right)\psi \tag{5.198}$$
and the Lagrangian of the interaction is given by

$$\mathcal{L}_{\rm int} = \overline{\psi} \, i \gamma^{\mu} D_{\mu} \, \psi \tag{5.199}$$

g' is the U(1) gauge coupling. The matrices T^a are a representation of the SU(2) weak isospin algebra and Y_{ψ} is the weak hypercharge of the U(1). In order to specify the coupling to matter we have to choose the SU(2) representation T^a and the U(1) gauge charge Y_{ψ} for the matter fields. Here the value of the generator (charge) Y_{ψ} depends on the fermion field. Three group generators T^a correspond to three gauge bosons $A^1_{\mu}, A^2_{\mu}, A^3_{\mu}$; the group generators for gauge doublets are

$$T^a = \frac{\tau^a}{2}, \quad a = 1, 2, 3$$
 (5.200)

and for gauge singlets (e_R^i, ν_R^i)

$$T^a = 0 \tag{5.201}$$

They all satisfy the SU(2) commutation relations:

$$\begin{bmatrix} T^a, T^b \end{bmatrix} = i \,\epsilon^{abc} \, T^c, \qquad \epsilon^{123} = 1 \tag{5.202}$$

and the explicit expression of Pauli matrices are

$$\tau^{1} = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}, \quad \tau^{2} = \begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix}, \quad \tau^{3} = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}$$
(5.203)

Defining

$$W^{\pm}_{\mu} = \frac{A^{1}_{\mu} \mp i A^{2}_{\mu}}{\sqrt{2}}$$
 and $T^{\pm} = T^{1} \pm i T^{2}$ (5.204)

we have

$$T^{a}A^{a}_{\mu} = \frac{1}{\sqrt{2}}T^{+}W^{+}_{\mu} + \frac{1}{\sqrt{2}}T^{-}W^{-}_{\mu} + T^{3}A^{3}_{\mu}$$
(5.205)

where the matrices T^{\pm} and T^{3} satisfy the relations

$$\begin{bmatrix} T^+, T^- \end{bmatrix} = 2 T^3 \begin{bmatrix} T^3, T^{\pm} \end{bmatrix} = \pm T^{\pm}$$
(5.206)

 T^+ and T^- are raising and lowering operators. In the doublet representation of SU(2) we have

$$T^{+} = \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix}, \quad T^{-} = \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix}, \quad T^{3} = \begin{pmatrix} \frac{1}{2} & 0 \\ 0 & -\frac{1}{2} \end{pmatrix}$$
(5.207)

It is worth mentioning that A^3_{μ} and B_{μ} carry identical quantum numbers ($T^3 = 0, Y_{\psi} = 0$), so at the end they will combine to produce two neutral gauge bosons: A_{μ} and Z_{μ} . The neutral weak current was discovered at CERN in 1973, while the Z boson was discovered at CERN in 1983.

The Higgs mechanism: Spontaneous symmetry breaking

Experimentally, the weak gauge bosons are massive, there is only one massless gauge boson in nature, the photon. The Lagrangian Eq. (5.196) describes four massless vector bosons forming a singlet (B_{μ}) and a triplet $(W_{\mu}^{\pm}, A_{\mu}^{3})$. So the model cannot describe the real world. If we add the explicit mass terms for the three weak bosons, then it would violate local gauge invariance and spoil the renormalizability of the theory. Therefore it is necessary to introduce a mechanism of symmetry breaking by which the three weak bosons obtain masses. The mass generation can be implemented through the Higgs mechanism [41–43]: generate mass terms from the kinetic energy term of a scalar doublet field ϕ that undergoes spontaneous symmetry breaking (SSB).

In the standard electroweak theory, the gauge group $SU(2) \times U(1)$ is broken by the Higgs mechanism. Introducing a single complex doublet of scalar Higgs fields

$$\phi = \begin{pmatrix} \phi_1 \\ \phi_2 \end{pmatrix} = \begin{pmatrix} \phi^+ \\ \phi^0 \end{pmatrix}$$
(5.208)

on which the matrices τ^a act. The Higgs Lagrangian is given by

$$\mathcal{L}_{\text{Higgs}} = (D_{\mu}\phi)^{\dagger} D_{\mu}\phi - V(\phi^{\dagger}\phi)$$
(5.209)

which is gauge invariant under local $SU(2) \times U(1)$ transformations. The coupling of the gauge fields to the scalar field is achieved using the covariant derivative

$$D_{\mu}\phi = (\partial_{\mu} - i g T^{a} A^{a}_{\mu} - i g' Y_{\phi} B_{\mu}) \phi, \qquad Y_{\phi} = \frac{1}{2}$$
$$= (\partial_{\mu} - i g \frac{\tau^{a}}{2} A^{a}_{\mu} - i g' \frac{1}{2} B_{\mu}) \phi \qquad (5.210)$$

The Higgs potential is chosen to be of the form

$$V(\phi^{\dagger}\phi) = V_0 + \lambda \left(\phi^{\dagger}\phi\right)^2 - \mu^2 \phi^{\dagger}\phi, \qquad \mu^2, \lambda > 0$$
(5.211)

Note that this potential has a wrong sign for the mass term. As a result, with the parameters μ^2 , $\lambda > 0$, this potential has a classical minimum which is not at $\phi = 0$; instead the potential has minima at

$$|\phi| = \sqrt{\frac{\mu^2}{2\,\lambda}} = \frac{v}{\sqrt{2}} \tag{5.212}$$

All these minimum configurations (ground states) are connected by gauge transformations, that change the phase of the complex field ϕ without altering its modulus. v is called the vacuum expectation value (VEV) of the neutral component of the scalar Higgs doublet. When the system chooses one of the ground states, this ground state is no longer symmetric under the gauge transformation. However the Lagrangian is still gauge invariant under the gauge transformation and all properties connected with it still hold (*e.g.* current conservation). This phenomenon is called spontaneous symmetry breaking. We will discuss the consequences of the Higgs mechanism in more detail in next subsection.

Glashow-Weinberg-Salam theory

We start with the gauge and scalar sector of the theory. The $SU(2) \times U(1)$ gauge invariant Lagrangian is

$$\mathcal{L} = -\frac{1}{4} F^{a}_{\mu\nu} F^{\mu\nu}_{a} - \frac{1}{4} B_{\mu\nu} B^{\mu\nu} + (D_{\mu}\phi)^{\dagger} D_{\mu}\phi - V(\phi^{\dagger}\phi)$$
(5.213)

where the field strength tensors are given by Eq. (5.197); the covariant derivative connecting gauge fields and scalar field is given by Eq. (5.210). We choose the ground state to be

$$A^{a}_{\mu} = B_{\mu} = 0$$

$$<\phi> = \frac{1}{\sqrt{2}} \begin{pmatrix} 0\\v \end{pmatrix} = \phi_{0}$$
(5.214)

Note that only a scalar field can have a VEV. The VEV of a fermion or vector field would break Lorentz invariance. The generators of the gauge group $SU(2) \times U(1)$ are matrices $T^a = \frac{\tau^a}{2}$ and $Y_{\phi} = \frac{1}{2}$. Now we would like to show that the ground state breaks the gauge symmetry. An arbitrary state Φ is invariant under a symmetry operation $\exp(i T^a \theta^a)$ if

$$e^{iT^a\,\theta^a}\Phi = \Phi \tag{5.215}$$

This means that a state is invariant if

$$T^a \Phi = 0 \tag{5.216}$$

For the $SU(2) \times U(1)$ case we have

$$\tau^{1} \phi_{0} \neq 0$$

$$\tau^{2} \phi_{0} \neq 0$$

$$\tau^{3} \phi_{0} \neq 0$$

$$Y_{\phi} \phi_{0} = \frac{1}{2} \phi_{0} \neq 0$$
(5.217)

Here the generators τ^a and Y_{ϕ} correspond to broken generators, the consequence of which is that all the gauge bosons will receive positive masses. However, it may be the case that the generators of the group leave the vacuum (ground state) invariant, in which case the corresponding gauge bosons will remain massless; and the corresponding generators are called unbroken generators. Now we examine the effect of the electric charge operator Q on the vacuum state. The generator Q satisfies

$$Q \phi_0 = (T^3 + Y_\phi) \phi_0 = \begin{pmatrix} 1 & 0 \\ 0 & 0 \end{pmatrix} \phi_0 = 0$$
(5.218)

the electric charge symmetry is unbroken. So there is an unbroken subgroup with a single generator Q: This is the subgroup $U(1)_Q$ of $SU(2) \times U(1)$. This subgroup corresponds to a massless gauge field, which is the electromagnetic field (photon A_{μ}). The choice of the vacuum expectation value in Eq. (5.214) breaks the $SU(2) \times U(1)$ gauge symmetry, since it identifies a specific direction in the internal group space. Now we shall consider small perturbations of the fields around the vacuum,

$$\phi(x) = \begin{pmatrix} \phi^+ \\ \phi^0 \end{pmatrix} = \begin{pmatrix} \phi^+ \\ \frac{1}{\sqrt{2}} \left[v + \chi(x) + i\eta(x) \right] \end{pmatrix} = \frac{1}{\sqrt{2}} \exp\left(\frac{iT^a\theta^a(x)}{v}\right) \begin{pmatrix} 0 \\ v + \chi(x) \end{pmatrix}$$
(5.219)

We still have four real degrees of freedom, (three θ^a and one χ), equivalent to the two complex fields. We can use the unitary gauge in which

$$\phi(x) = \frac{1}{\sqrt{2}} \begin{pmatrix} 0\\ v + \chi(x) \end{pmatrix}$$
(5.220)

In this gauge, the Goldstone fields $\theta^a(x)$ have been rotated away by an SU(2) gauge transformation.

$$\phi(x) \rightarrow \phi'(x) = U(x)\phi(x), \qquad U(x) = \exp\left(-\frac{iT^a\theta^a(x)}{v}\right)$$
$$T \cdot A_\mu \rightarrow T \cdot A'_\mu = UT \cdot A_\mu U^{-1} - \frac{i}{g}\left(\partial_\mu U\right) U^{-1}$$
(5.221)

The $\theta^a(x)$ degrees of freedom no longer appear in the Higgs Lagrangian, they will reappear as the longitudinal modes of the massive gauge bosons. The Higgs boson χ is the only remaining dynamical field. In order to find the mass terms for the gauge bosons and Higgs boson, we need to calculate the quadratic Lagrangian, which means the calculation of the covariant derivative of the field ϕ

$$D_{\mu}\phi = \begin{pmatrix} -\frac{ig}{2\sqrt{2}}(A_{\mu}^{1} - iA_{\mu}^{2})(v + \chi) \\ -\frac{i}{2\sqrt{2}}(g'B_{\mu} - gA_{\mu}^{3})(v + \chi) + \frac{1}{\sqrt{2}}\partial_{\mu}\chi \end{pmatrix}$$

$$= \frac{1}{\sqrt{2}}\begin{pmatrix} 0 \\ \partial_{\mu}\chi \end{pmatrix} - \frac{i}{2}\left(1 + \frac{\chi}{v}\right)\left(\frac{gvW_{\mu}^{+}}{-\sqrt{(g^{2} + g'^{2})/2}vZ_{\mu}}\right)$$
(5.222)

The physical weak bosons are linear combinations of the gauge ones, so we have defined

$$W_{\mu}^{\pm} = \frac{1}{\sqrt{2}} \left(A_{\mu}^{1} \mp i A_{\mu}^{2} \right)$$

$$Z_{\mu} = \frac{1}{\sqrt{g^{2} + g^{\prime 2}}} (g A_{\mu}^{3} - g^{\prime} B_{\mu}) = \cos \theta_{W} A_{\mu}^{3} - \sin \theta_{W} B_{\mu}$$

$$A_{\mu} = \frac{1}{\sqrt{g^{2} + g^{\prime 2}}} (g B_{\mu} + g^{\prime} A_{\mu}^{3}) = \sin \theta_{W} A_{\mu}^{3} + \cos \theta_{W} B_{\mu}$$
(5.223)

where

$$\sin \theta_W = \frac{g'}{\sqrt{g^2 + g'^2}}, \qquad \cos \theta_W = \frac{g}{\sqrt{g^2 + g'^2}}$$
 (5.224)

and it is easy to show that

$$Z_{\mu}^{2} + (A_{\mu})^{2} = (A_{\mu}^{3})^{2} + B_{\mu}^{2}$$
(5.225)

Hence, the covariant derivative to the quadratic part of the Lagrangian is

$$\left[(D_{\mu}\phi)^{\dagger}D^{\mu}\phi \right]^{(2)} = \frac{1}{2}(\partial_{\mu}\chi)^{2} + \left[\frac{g^{2}v^{2}}{4}W_{\mu}^{+}W^{-\mu} + \frac{1}{2}\left(\frac{(g^{2}+g'^{2})v^{2}}{4}\right)Z_{\mu}^{2} \right] \left(1 + \frac{\chi}{v}\right)^{2}$$
(5.226)

To quadratic order, the kinetic term of the vector fields in the quadratic Lagrangian is

$$-\frac{1}{4}\hat{F}^{a}_{\mu\nu}\hat{F}^{\mu\nu}_{a} - \frac{1}{4}B_{\mu\nu}B^{\mu\nu} = -\frac{1}{2}W^{+}_{\mu\nu}W^{\mu\nu-} - \frac{1}{4}\hat{F}^{3}_{\mu\nu}\hat{F}^{\mu\nu}_{3} - \frac{1}{4}B_{\mu\nu}B^{\mu\nu} \\
= -\frac{1}{2}W^{+}_{\mu\nu}W^{\mu\nu-} - \frac{1}{4}Z_{\mu\nu}Z^{\mu\nu} - \frac{1}{4}F_{\mu\nu}F^{\mu\nu}$$
(5.227)

where

$$\hat{F}^{a}_{\mu\nu} = \partial_{\mu}A^{a}_{\nu} - \partial_{\nu}A^{a}_{\mu}
W^{\pm}_{\mu\nu} = \partial_{\mu}W^{\pm}_{\nu} - \partial_{\nu}W^{\pm}_{\mu}
Z_{\mu\nu} = \partial_{\mu}Z_{\nu} - \partial_{\nu}Z_{\mu}
F_{\mu\nu} = \partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu}$$
(5.228)

Thus, the quadratic Lagrangian contains standard kinetic terms for the complex vector field W^{\pm}_{μ} and the real vector fields A_{μ} and Z_{μ} . Collecting all together, we obtain the quadratic Lagrangian

$$\mathcal{L}^{(2)} = -\frac{1}{2} W^{+}_{\mu\nu} W^{\mu\nu-} + M^{2}_{W} W^{+}_{\mu} W^{-\mu} - \frac{1}{4} F_{\mu\nu} F^{\mu\nu} - \frac{1}{4} Z_{\mu\nu} Z^{\mu\nu} + \frac{M^{2}_{Z}}{2} Z_{\mu} Z^{\mu} + \frac{1}{2} (\partial_{\mu} \chi)^{2} - \frac{M^{2}_{\chi}}{2} \chi^{2}$$
(5.229)

We therefore find that the W and Z gauge bosons have acquired masses, given by

$$M_W = \frac{g v}{2}, \quad M_Z = \frac{\sqrt{g^2 + g'^2 v}}{2} = \frac{M_W}{\cos \theta_W}$$
 (5.230)

The photon remains massless because there are no terms quadratic in the field A_{μ} ; the quadratic Lagrangian also describes a massive real scalar field χ (Higgs boson field) with mass given by

$$M_{\chi} = \sqrt{2\,\lambda}\,v\tag{5.231}$$

From the measured value of the Fermi constant

$$\frac{G_F}{\sqrt{2}} = \frac{g^2}{8\,M_W^2} \tag{5.232}$$

the vacuum expectation value of the Higgs field, v, is 246 GeV.

In summary, the Higgs field ϕ acquires a non-zero vacuum expectation value at a particular point on the circle of minima away from the point where $\phi = 0$ and the symmetry is spontaneously broken. Three Goldstone bosons of the four scalar field components get eaten by the gauge bosons to form massive vector bosons (W_{μ}^{\pm}, Z_{μ}) , and a single physical scalar particle remains: The Higgs boson χ with mass given by $M_{\chi} = \sqrt{2\lambda} v$.

There are $WW\chi$ and $ZZ\chi$ couplings from $2\chi/v$ term in Eq. (5.226)

$$\mathcal{L}_{\chi VV} = \frac{2 M_W^2}{v} W_{\mu}^+ W^{-\mu} \chi + \frac{M_Z^2}{v} Z^{\mu} Z_{\mu} \chi = g M_W W_{\mu}^+ W^{-\mu} \chi + \frac{1}{2} \frac{g M_Z}{\cos \theta_W} Z^{\mu} Z_{\mu} \chi \quad (5.233)$$

There are also $\chi\chi WW$ and $\chi\chi ZZ$ couplings from χ^2/v^2 term. An important fact is that Higgs coupling is proportional to mass. Finally we summarize some key points of Glashow-Weinberg-Salam theory:

• To break the symmetry spontaneously, we introduce a scalar Higgs field ϕ in the fundamental representation of SU(2) with non-zero VEV:

$$<\phi>=\phi_0=rac{1}{\sqrt{2}}\left(egin{array}{c}0\\v\end{array}
ight)$$

- Broken symmetry will be applied if the original symmetry were global rather than gauge.
- Let G be a gauge group, the generators of G can be divided into unbroken generators {t_h} and broken generators {t_α}.
- If G is a global symmetry group, then the theory would contain massless Goldstone fields, with number equal to broken generators.
- If G is a gauge group, then the gauge fields corresponding to unbroken generators remain massless, while the gauge fields corresponding to broken generators become massive.
- Goldstone bosons corresponding to broken generators {t_α} disappear from the spectrum; they get eaten by the gauge particles to form massive vector bosons.

Fermion interactions

From experimental facts, charged currents couple only to left-handed fermions and neutral currents couple to a massless photon A_{μ} and a neutral gauge boson Z_{μ} , the gauge group is

 $SU(2) \times U(1)$. For the case of leptons or quarks the left-handed fields are put into doublets; the right-handed fields are all SU(2) singlets

$$\psi_L = \frac{1}{2}(1 - \gamma^5)\psi, \quad \psi_R = \frac{1}{2}(1 + \gamma^5)\psi, \quad \psi = \psi_L + \psi_R$$
(5.234)

It is easy to show

$$\overline{\psi}\,i\,\partial\!\!\!/\,\psi \,=\,\overline{\psi}_L\,i\,\partial\!\!\!/\,\psi_L + \overline{\psi}_R\,i\,\partial\!\!\!/\,\psi_R \tag{5.235}$$

Hence the first generation of left-handed lepton fields is

$$\ell_L = \frac{1}{2} (1 - \gamma^5) \begin{pmatrix} \nu_e \\ e \end{pmatrix} = \begin{pmatrix} \nu_{eL} \\ e_L \end{pmatrix}$$
(5.236)

which is SU(2) doublet, and the first generation of right-handed lepton fields are

$$\nu_{eR} = \frac{1}{2} (1 + \gamma^5) \nu_e, \quad e_R = \frac{1}{2} (1 + \gamma^5) e$$
(5.237)

which form SU(2) singlets. Similarly the quark fields form left-handed doublets

$$q_L = \frac{1}{2}(1-\gamma^5) \begin{pmatrix} u \\ d' \end{pmatrix} = \begin{pmatrix} u_L \\ d'_L \end{pmatrix}, \quad \frac{1}{2}(1-\gamma^5) \begin{pmatrix} c \\ s' \end{pmatrix}, \quad \frac{1}{2}(1-\gamma^5) \begin{pmatrix} t \\ b' \end{pmatrix} \quad (5.238)$$

and the right-handed quark fields are singlets. The primes on the down type quarks will be explained later. All these relations hold for each family. Before symmetry breaking, the coupling of the fermions to the vector bosons is given by

$$\mathcal{L}_{\psi} = \overline{\psi} \, i \, \not\!\!\!D \, \psi = \overline{\psi}_L \, i \, \not\!\!\!D \, \psi_L + \overline{\psi}_R \, i \, \not\!\!\!D \, \psi_R$$

$$= \overline{\psi}_L \, i \gamma^\mu \left(\partial_\mu - i \, g \, T^a \, A^a_\mu - i \, g' \, Y_L \, B_\mu \right) \psi_L + \overline{\psi}_R \, i \gamma^\mu \left(\partial_\mu - i \, g' \, Y_R \, B_\mu \right) \psi_R \qquad (5.239)$$

with

$$Y_{\psi}\psi_L = Y_L\psi_L, \quad Y_{\psi}\psi_R = Y_R\psi_R \tag{5.240}$$

Here ψ denote left- and right-handed quarks and leptons. The U(1) charges, Y_L and Y_R , are chosen to satisfy the relation $Q = T^3 + Y_{\psi}$, so that after symmetry breaking we obtain the correct values of the electric charges Q. Table 5.2 lists all the values of weak isospin and hypercharge for quarks and leptons. Now we focus on one generation of leptons (quarks work the same way)

where the covariant derivative is given by Eq. (5.198) and a right-handed neutrino would have zero coupling both to SU(2) and to U(1), so we have simply omitted this field. We will find that it is also useful to rewrite Eq. (5.198) in terms of the gauge boson mass eigenstates $W^+_{\mu}, W^-_{\mu}, A_{\mu}$ and Z_{μ} fields

$$D_{\mu} = \partial_{\mu} - i \frac{g}{\sqrt{2}} (W_{\mu}^{+} T^{+} + W_{\mu}^{-} T^{-}) - i \frac{g}{\cos \theta_{W}} Z_{\mu} (T^{3} - \sin^{2} \theta_{W} Q) - i e A_{\mu} Q \quad (5.242)$$

where the values of e and the weak SU(2) charge g are related by

$$g = \frac{e}{\sin \theta_W} \tag{5.243}$$

To work out the physical consequences of the fermion-gauge boson couplings, we should write Eq. (5.241) in terms of the vector boson mass eigenstates using the form of the covariant derivative given in Eq. (5.242). Thus the fermion Lagrangian can be divided into three pieces: kinetic term, charged current and neutral current.

$$\mathcal{L}_{\psi} = \mathcal{L}_{\rm kin} + g \left(W_{\mu}^{+} J_{W}^{\mu +} + W_{\mu}^{-} J_{W}^{\mu -} + Z_{\mu} J_{Z}^{\mu} \right) + e A_{\mu} J_{EM}^{\mu}$$
(5.244)

where

$$\mathcal{L}_{kin} = i\overline{\ell}_L \not \partial \ell_L + i\overline{e}_R \not \partial e_R$$

$$\mathcal{L}_{CC} = g \left(W^+_\mu J^{\mu+}_W + W^-_\mu J^{\mu-}_W \right) = \frac{g}{\sqrt{2}} \overline{\ell}_L \left(W^+_\mu T^+ + W^-_\mu T^- \right) \gamma^\mu \ell_L$$

$$= \frac{g}{\sqrt{2}} \left[W^+_\mu \overline{\nu}_{eL} \gamma^\mu e_L + W^-_\mu \overline{e}_L \gamma^\mu \nu_{eL} \right]$$
(5.245)
(5.246)

The neutral current interactions involving Z_{μ} vector boson is

$$\mathcal{L}_{\mathrm{NC}} = g Z_{\mu} J_{Z}^{\mu} = \frac{g}{\cos \theta_{W}} Z_{\mu} \left[\overline{\ell}_{L} \left(T^{3} - \sin^{2} \theta_{W} Q \right) \gamma^{\mu} \ell_{L} + \overline{e}_{R} \left(T^{3} - \sin^{2} \theta_{W} Q \right) \gamma^{\mu} e_{R} \right]$$
$$= \frac{g}{\cos \theta_{W}} Z_{\mu} \left[\overline{\nu}_{eL} \left(\frac{1}{2} \right) \gamma^{\mu} \nu_{eL} + \overline{e}_{L} \left(-\frac{1}{2} + \sin^{2} \theta_{W} \right) \gamma^{\mu} e_{L} + \overline{e}_{R} \left(\sin^{2} \theta_{W} \right) \gamma^{\mu} e_{R} \right]$$
(5.247)

where

$$Q\,\psi_q \,=\, q\,\psi_q \tag{5.248}$$

This procedure works for leptons and also for the quarks, *e.g.* the charged current Lagrangian for quark sector is

$$\frac{g}{\sqrt{2}} \left[W^+_\mu \,\overline{u}_L \,\gamma^\mu \,d_L + W^-_\mu \,\overline{d}_L \,\gamma^\mu \,u_L \right] \tag{5.249}$$

The charged currents of lepton and quark sector for one generation is given by

$$J_W^{\mu+} = \frac{1}{\sqrt{2}} \left[\overline{\nu}_{eL} \gamma^{\mu} e_L + \overline{u}_L \gamma^{\mu} d_L \right]$$

$$J_W^{\mu-} = \frac{1}{\sqrt{2}} \left[\overline{e}_L \gamma^{\mu} \nu_{eL} + \overline{d}_L \gamma^{\mu} u_L \right]$$
(5.250)

Finally we comment that the theoretical motivation for grouping the quarks and leptons as shown in Table 5.2 is that complete families are required for the cancellation of anomalies in the currents which couple to gauge fields. This cancellation shows that Ward identities, which are crucial for the proof of renormalizability of the gauge theory at quantum level, are still validated.

Fermion			T_L^3	Y_L	T_R^3	Y_R	Q
u	С	t	$+\frac{1}{2}$	$+\frac{1}{6}$	0	$+\frac{2}{3}$	$+\frac{2}{3}$
d	S	b	$-\frac{1}{2}$	$+\frac{1}{6}$	0	$-\frac{1}{3}$	$-\frac{1}{3}$
$ u_e$	$ u_{\mu}$	$ u_{ au}$	$+\frac{1}{2}$	$-\frac{1}{2}$	_	_	0
e	μ	τ	$-\frac{1}{2}$	$-\frac{1}{2}$	0	-1	-1

Table 5.2: Weak isospin and hypercharge assignments, they are related through $Q = T^3 + Y_{\psi}$.

Yukawa interactions

A direct fermion mass term

$$m_f \overline{\psi} \,\psi = m_f \left(\overline{\psi}_R \,\psi_L + \overline{\psi}_L \,\psi_R \right) \tag{5.251}$$

is not gauge invariant under SU(2) or U(1) gauge transformation. The Higgs field ϕ can give masses to the fermions via Yukawa interactions with the fermion fields. After spontaneous symmetry breaking, a Yukawa interaction of the form $g_f \overline{f}_L \phi f_R$ leads to a fermion mass, $m_f = g_f v / \sqrt{2}$. The Yukawa Lagrangian is

$$\mathcal{L}_{\text{Yukawa}} = -\Gamma_d^{ij} \bar{q}_L^{\prime i} \phi \, d_R^{\prime j} - \Gamma_d^{ij \star} \bar{d}_R^{\prime i} \phi^{\dagger} q_L^{\prime j} -\Gamma_u^{ij} \bar{q}_L^{\prime i} \phi_c \, u_R^{\prime j} + \text{h.c.} -\Gamma_e^{ij} \bar{\ell}_L^{i} \phi \, e_R^{j} + \text{h.c.}$$
(5.252)

where

$$\phi(x) = \frac{1}{\sqrt{2}} \begin{pmatrix} 0 \\ v + H(x) \end{pmatrix}, \quad \phi_c = i \tau^2 \phi^* = \frac{1}{\sqrt{2}} \begin{pmatrix} v + H(x) \\ 0 \end{pmatrix}$$
(5.253)

and here we have replaced the Higgs boson notation χ with H. q_L and ℓ_L are left-handed doublet fields and d_R , u_R , e_R are right-handed SU(2) singlet fields. The primes for q'_L , d'_R and u'_R mean that they are quark fields that are generic linear combination of the mass eigenstates u and d. Γ_d^{ij} , Γ_u^{ij} and Γ_e^{ij} are 3×3 complex matrices in generation space, spanned by the indices i and j. Note that neutrino masses can be implemented via Γ_{ν}^{ij} term. Since m_{ν} is very small so that we neglect it in the following. $\mathcal{L}_{\text{Yukawa}}$ is Lorentz invariant, gauge invariant and renormalizable, and therefore it can be included in the Lagrangian. In the unitary gauge we have

$$\bar{q}_{L}^{\prime i} \phi \, d_{R}^{\prime j} = \frac{1}{\sqrt{2}} \begin{pmatrix} \bar{u}_{L}^{\prime i} & \bar{d}_{L}^{\prime i} \end{pmatrix} \begin{pmatrix} 0 \\ v+H \end{pmatrix} d_{R}^{\prime j} = \frac{v+H}{\sqrt{2}} \bar{d}_{L}^{\prime i} d_{R}^{\prime j}$$

$$\bar{q}_{L}^{\prime i} \phi_{c} \, u_{R}^{\prime j} = \frac{1}{\sqrt{2}} \begin{pmatrix} \bar{u}_{L}^{\prime i} & \bar{d}_{L}^{\prime i} \end{pmatrix} \begin{pmatrix} v+H \\ 0 \end{pmatrix} u_{R}^{\prime j} = \frac{v+H}{\sqrt{2}} \bar{u}_{L}^{\prime i} u_{R}^{\prime j}$$
(5.254)

so Eq. (5.252) becomes

$$\mathcal{L}_{\text{Yukawa}} = -\Gamma_{d}^{ij} \frac{v+H}{\sqrt{2}} \bar{d}_{L}^{\prime i} d_{R}^{\prime j} - \Gamma_{u}^{ij} \frac{v+H}{\sqrt{2}} \bar{u}_{L}^{\prime i} u_{R}^{\prime j} - \Gamma_{e}^{ij} \frac{v+H}{\sqrt{2}} \bar{e}_{L}^{i} e_{R}^{j} + \text{h.c.}$$

$$= -\left[M_{u}^{ij} \bar{u}_{L}^{\prime i} u_{R}^{\prime j} + M_{d}^{ij} \bar{d}_{L}^{\prime i} d_{R}^{\prime j} + M_{e}^{ij} \bar{e}_{L}^{i} e_{R}^{j} + \text{h.c.} \right] \left(1 + \frac{H}{v} \right)$$
(5.255)

with mass matrices

$$M^{ij} = \Gamma^{ij} \frac{v}{\sqrt{2}} \tag{5.256}$$

Now we would like to diagonalize the mass matrices $M_f^{ij}(f = u, d, e)$, which can be achieved using a bi-unitary transformation U_L^f and U_R^f .

$$f_{L}^{\prime i} = \left(U_{L}^{f}\right)_{ij} f_{L}^{j}, \qquad f_{R}^{\prime i} = \left(U_{R}^{f}\right)_{ij} f_{R}^{j}$$
 (5.257)

with U_L^f and U_R^f chosen such that

$$\left(U_L^f\right)^{\dagger} M_f \left(U_R^f\right) =$$
diagonal (5.258)

and $U_{L/R}^{f}$ must be unitary in order to preserve the form of the kinetic terms in the Lagrangian. We give two examples of diagonalized fermion mass matrices

$$(U_L^u)^{\dagger} M_u U_R^u = \begin{pmatrix} m_u & 0 & 0\\ 0 & m_c & 0\\ 0 & 0 & m_t \end{pmatrix}, \qquad (U_L^d)^{\dagger} M_d U_R^d = \begin{pmatrix} m_d & 0 & 0\\ 0 & m_s & 0\\ 0 & 0 & m_b \end{pmatrix}$$
(5.259)

Hence Yukawa interactions Eq. (5.252) can be rewritten as

$$\mathcal{L}_{\text{Yukawa}} = -\sum_{f',i,j} M_f^{ij} \bar{f}_L^{\prime i} f_R^{\prime j} \left(1 + \frac{H}{v}\right) + \text{h.c.}$$

$$= -\sum_{f,i,j} \bar{f}_L^i \left[\left(U_L^f \right)^{\dagger} M_f \left(U_R^f \right) \right]_{ij} f_R^j \left(1 + \frac{H}{v}\right) + \text{h.c.}$$

$$= -\sum_f m_f \left[\bar{f}_L f_R + \bar{f}_R f_L \right] \left(1 + \frac{H}{v} \right)$$

$$= -\sum_f m_f \bar{f}_f \left(1 + \frac{H}{v} \right) \qquad (5.260)$$

We succeed in producing fermion masses and we got a fermion-antifermion-Higgs coupling which is proportional to the fermion mass. Obviously the Higgs Yukawa couplings are flavour diagonal, this means there is no flavour changing Higgs interactions. Recall that the charged current interaction is of the form

$$g W^+_{\mu} J^{\mu+}_W + \text{h.c.} = \frac{e}{\sqrt{2} \sin \theta_W} W^+_{\mu} \bar{u}^{\prime i}_L \gamma^{\mu} d^{\prime i}_L + \text{h.c.}$$
(5.261)

After the mass diagonalization described previously, this term becomes

$$\frac{e}{\sqrt{2}\sin\theta_W}W^+_{\mu}\bar{u}^i_L \left[(U^u_L)^{\dagger} U^d_L \right]_{ij} \gamma^{\mu} d^j_L + \text{h.c.}$$
(5.262)

We define the Cabibbo-Kobayashi-Maskawa matrix V_{CKM} .

$$V_{CKM} = \left(U_L^u\right)^\dagger \, U_L^d \tag{5.263}$$

The CKM matrix is not diagonal and then it mixes the flavours of the different quarks. It is a unitary matrix $(V_{CKM}^{\dagger} V_{CKM} = 1)$ and the values of its entries must be determined from experiments. The CKM matrix connects the weak eigenstates (d', s', b') and the corresponding mass eigenstates (d, s, b) through

$$\begin{pmatrix} d'\\s'\\b' \end{pmatrix} = \begin{pmatrix} V_{ud} & V_{us} & V_{ub}\\V_{cd} & V_{cs} & V_{cb}\\V_{td} & V_{ts} & V_{tb} \end{pmatrix} \begin{pmatrix} d\\s\\b \end{pmatrix} = V_{\rm CKM} \begin{pmatrix} d\\s\\b \end{pmatrix}$$
(5.264)

For 3×3 CKM matrix, the matrix element can be parameterized by 3 angles and 1 phase, which gives rise to CP violation in Standard Model (SM). Now we look at the neutral current interaction, *e.g.* down type quarks is given by

$$g Z_{\mu} J_{Z}^{\mu} = \frac{e}{\sin \theta_{W} \cos \theta_{W}} \left[\left(-\frac{1}{2} + \frac{1}{3} \sin^{2} \theta_{W} \right) Z_{\mu} \bar{d}_{L}^{\prime i} \gamma^{\mu} d_{L}^{\prime i} + \left(+\frac{1}{3} \sin^{2} \theta_{W} \right) Z_{\mu} \bar{d}_{R}^{\prime i} \gamma^{\mu} d_{R}^{\prime i} \right]$$
(5.265)

After the mass diagonalization we have

$$\frac{e}{\sin\theta_W\cos\theta_W} \left(-\frac{1}{2} + \frac{1}{3}\sin^2\theta_W\right) Z_\mu \bar{d}_L^i \left[\left(U_L^d\right)^\dagger U_L^d\right]_{ij} \gamma^\mu d_L^j + \frac{e}{\sin\theta_W\cos\theta_W} \left(+\frac{1}{3}\sin^2\theta_W\right) Z_\mu \bar{d}_R^i \left[\left(U_R^d\right)^\dagger U_R^d\right]_{ij} \gamma^\mu d_R^j \\
= \frac{e}{\sin\theta_W\cos\theta_W} \left[\left(-\frac{1}{2} + \frac{1}{3}\sin^2\theta_W\right) Z_\mu \bar{d}_L^i \gamma^\mu d_L^i + \left(+\frac{1}{3}\sin^2\theta_W\right) Z_\mu \bar{d}_R^i \gamma^\mu d_R^i\right] (5.266)$$

So the unitary matrices cancel and the Z boson interaction is flavour diagonal also in the mass eigenstate. It works the same way for the other flavours. This mechanism is called the GIM mechanism (Glashow, Iliopoulos and Maiani) [79]. It predicts the existence of the charm quark;

there is then a second doublet (c_L, s'_L) . Hence, the transitions, like $d \to s$ and $s \to d$, cancel precisely. This GIM mechanism generalizes for any number of quark generations.

Finally we summarize the key points of this section.

- The Higgs field ϕ can give masses to the fermions via Yukawa interactions with the fermion fields.
 - \longrightarrow No flavour changing Higgs interactions.
- The flavour is conserved in vertices involving neutral gauge bosons: A_{μ}, Z_{μ} and G_{μ} . \longrightarrow GIM mechanism.
 - \longrightarrow Absence of flavour changing neutral current (FCNC) at the tree level.
 - $\longrightarrow b \rightarrow s\gamma, \cdots$ are loop-induced in SM (high sensitivity to new physics effects).
- The charged current processes mediated by W^{\pm} are flavour violating with the strength of violation given by the SU(2) gauge coupling g and a unitary 3×3 CKM matrix.

5.8.4 The Standard Model of particle physics

The Standard Model (SM) of elementary particle physics is a gauge theory of strong and electroweak interactions. It is based on the following gauge group.

$$SU(3)_C \times SU(2)_L \times U(1)_Y \tag{5.267}$$

The $SU(3)_C$ is the colour group of QCD, while $SU(2)_L \times U(1)_Y$ is the Glashow-Weinberg-Salam electroweak symmetry group, which is spontaneously broken down to $U(1)_Q$, the phase group of the electric charge Q, different from the $U(1)_Y$ of weak hypercharge: $Q = T^3 + Y$, where T^3 is the third component of the weak isospin generator of $SU(2)_L$. The group $SU(3)_C \times U(1)_Q$ is believed to be an exact gauge symmetry of nature. The gauge group Eq. (5.267) contains 12 spin 1 gauge bosons:

- 8 massless gluons of $SU(3)_C$, which are responsible for the strong interactions (QCD).
- 4 gauge bosons of SU(2)_L×U(1)_Y, which are responsible for the electroweak interactions, of which one is massless (photon field: A_μ) and three are massive (W[±] and Z gauge bosons) after spontaneous symmetry breaking (SSB).

These gauge bosons interact with matter fields (coloured quarks and colourless leptons) in a gauge invariant way. The field content is the following:

Gauge sector : Spin = 1

The gauge bosons are spin 1 vector particles belonging to the adjoint representation of the gauge group Eq. (5.267). Their quantum numbers are:

gluons	G^A_μ :	(8, 1, 0)	$SU(3)_C$	g_s	
weak bosons	A^a_μ :	(1, 3, 0)	$SU(2)_L$	g	(5.268)
abelian boson	B_{μ} :	(1, 1, 0)	$U(1)_Y$	g'	

where A runs over the eight colour degrees of freedom of the gluon field G^A_{μ} ($A = 1, \dots, 8$) and a denotes the isospin space (a = 1, 2, 3). In order to avoid confusion we have changed the notation for the strong coupling constant by g_s , which we denoted with g in Chapter 1.

Fermion sector : Spin = $\frac{1}{2}$

In the Standard Model the matter fields are fermions belonging to the fundamental representation of the gauge group Eq. (5.267):

quarks

$$SU(3)_C$$
 $SU(2)_L$
 $U(1)_Y$
 $U(1)_Q$: Q
 $q_L^i =$
 $\begin{pmatrix} u_L \\ d_L \end{pmatrix}$
 $\begin{pmatrix} c_L \\ s_L \end{pmatrix}$
 $\begin{pmatrix} t_L \\ b_L \end{pmatrix}$
 3
 2
 $\frac{1}{6}$
 $\begin{pmatrix} \frac{2}{3} \\ -\frac{1}{3} \end{pmatrix}$
 $u_R^i =$
 u_R
 c_R
 t_R
 $\bar{3}$
 1
 $\frac{2}{3}$
 $\frac{2}{3}$
 $d_R^i =$
 d_R
 s_R
 b_R
 $\bar{3}$
 1
 $-\frac{1}{3}$
 $-\frac{1}{3}$
 $d_R^i =$
 d_R
 s_R
 b_R
 $\bar{3}$
 1
 $-\frac{1}{3}$
 $-\frac{1}{3}$

 leptons
 $\ell_L^i =$
 $\begin{pmatrix} \nu_{eL} \\ e_L \end{pmatrix}$
 $\begin{pmatrix} \nu_{\mu L} \\ \mu_L \end{pmatrix}$
 $\begin{pmatrix} \nu_{\tau L} \\ \tau_L \end{pmatrix}$
 1
 2
 $-\frac{1}{2}$
 $\begin{pmatrix} 0 \\ -1 \end{pmatrix}$
 $e_R^i =$
 e_R
 μ_R
 τ_R
 1
 1
 -1
 -1
 $\nu_R^i =$
 ν_{eR}
 $\nu_{\mu R}$
 $\nu_{\tau R}$
 1
 1
 0
 0

where

$$Q = T^3 + Y \tag{5.269}$$

The Standard Model is described by the following Lagrangian:

$$\mathcal{L} = \mathcal{L}_{gauge} + \mathcal{L}_{Yukawa} + \mathcal{L}_{Higgs}$$
(5.270)

where the \mathcal{L}_{Yukawa} is given by Eq. (5.252), the \mathcal{L}_{Higgs} is given by Eq. (5.209), and the gauge interaction is given by

Here D is a notation for $D = \gamma^{\mu} D_{\mu}$ and *i* denotes three generations. The covariant derivative is given by

$$D_{\mu}\psi = \left(\partial_{\mu} - i\,g\,\frac{\tau^{a}}{2}A_{\mu}^{a} - i\,g'\,Y_{\psi}\,B_{\mu} + i\,g_{s}\,\frac{\lambda^{A}}{2}\,G_{\mu}^{A}\right)\psi$$
(5.272)

which acts on quark fields and lepton fields gives

$$D_{\mu}\ell_{L} = \left(\partial_{\mu} - ig\frac{\tau^{a}}{2}A_{\mu}^{a} - ig'\left(-\frac{1}{2}\right)B_{\mu}\right)\ell_{L}$$

$$D_{\mu}e_{R} = \left(\partial_{\mu} - ig'\left(-1\right)B_{\mu}\right)e_{R}$$

$$D_{\mu}q_{L} = \left(\partial_{\mu} - ig\frac{\tau^{a}}{2}A_{\mu}^{a} - ig'\left(\frac{1}{6}\right)B_{\mu} + ig_{s}\frac{\lambda^{A}}{2}G_{\mu}^{A}\right)q_{L}$$

$$D_{\mu}u_{R} = \left(\partial_{\mu} - ig'\left(\frac{2}{3}\right)B_{\mu} + ig_{s}\frac{\lambda^{A}}{2}G_{\mu}^{A}\right)u_{R}$$

$$D_{\mu}d_{R} = \left(\partial_{\mu} - ig'\left(-\frac{1}{3}\right)B_{\mu} + ig_{s}\frac{\lambda^{A}}{2}G_{\mu}^{A}\right)d_{R}$$
(5.273)

Bibliography

- [1] C. N. Yang and R. L. Mills, Phys. Rev. 96, 191 (1954).
- [2] H. D. Politzer, Phys. Rev. Lett. **30**, 1346 (1973).
- [3] D. J. Gross and F. Wilczek, Phys. Rev. Lett. 30, 1343 (1973).
- [4] R. P. Feynman, Reading 1972, 282p
- [5] J. B. Kogut and L. Susskind, Phys. Rept. 8, 75 (1973).
- [6] G. Altarelli and G. Parisi, Nucl. Phys. B 126, 298 (1977).
- [7] G. P. Lepage and S. J. Brodsky, Phys. Rev. D 22, 2157 (1980).
- [8] J. C. Collins, D. E. Soper and G. Sterman, Adv. Ser. Direct. High Energy Phys. 5, 1 (1988) [arXiv:hep-ph/0409313].
- [9] J. C. Collins and D. E. Soper, Ann. Rev. Nucl. Part. Sci. 37, 383 (1987).
- [10] T. Kinoshita, J. Math. Phys. 3, 650 (1962).
- [11] T. D. Lee and M. Nauenberg, Phys. Rev. 133, B1549 (1964).
- [12] H. Baer, J. Ohnemus and J. F. Owens, Phys. Rev. D 40, 2844 (1989).
- [13] K. Fabricius, I. Schmitt, G. Kramer and G. Schierholz, Z. Phys. C 11, 315 (1981).
- [14] G. Kramer and B. Lampe, Fortsch. Phys. 37, 161 (1989).
- [15] R. K. Ellis, D. A. Ross and A. E. Terrano, Nucl. Phys. B 178, 421 (1981).
- [16] Z. Kunszt and P. Nason, in 'Z Physics at LEP 1', CERN 89-08, vol. 1, p. 373.
- [17] Z. Kunszt and D. E. Soper, Phys. Rev. D 46, 192 (1992).
- [18] S. D. Ellis, Z. Kunszt and D. E. Soper, Phys. Rev. D 40, 2188 (1989).

- [19] S. D. Ellis, Z. Kunszt and D. E. Soper, Phys. Rev. Lett. 69, 1496 (1992).
- [20] S. Frixione, Z. Kunszt and A. Signer, Nucl. Phys. B 467, 399 (1996) [arXiv:hepph/9512328].
- [21] M. L. Mangano, P. Nason and G. Ridolfi, Nucl. Phys. B 373, 295 (1992).
- [22] S. Catani and M. H. Seymour, Nucl. Phys. B 485, 291 (1997) [Erratum-ibid. B 510, 503 (1998)] [arXiv:hep-ph/9605323].
- [23] S. Dittmaier, Nucl. Phys. B 565, 69 (2000) [arXiv:hep-ph/9904440].
- [24] S. Catani, S. Dittmaier, M. H. Seymour and Z. Trocsanyi, Nucl. Phys. B 627, 189 (2002) [arXiv:hep-ph/0201036].
- [25] S. Weinzierl, JHEP 0307, 052 (2003) [arXiv:hep-ph/0306248].
- [26] A. Bassetto, M. Ciafaloni and G. Marchesini, Phys. Rept. 100, 201 (1983).
- [27] Y. L. Dokshitzer, V. A. Khoze, A. H. Mueller and S. I. Troian, *Gif-sur-Yvette, France: Ed. Frontieres (1991) 274 p. (Basics of)*
- [28] Z. Nagy and D. E. Soper, JHEP 0709, 114 (2007) [arXiv:0706.0017 [hep-ph]].
- [29] Z. Nagy and D. E. Soper, JHEP 0803, 030 (2008) [arXiv:0801.1917 [hep-ph]].
- [30] Z. Nagy and D. E. Soper, JHEP 0807, 025 (2008) [arXiv:0805.0216 [hep-ph]].
- [31] D. A. Kosower, Phys. Rev. D 57, 5410 (1998) [arXiv:hep-ph/9710213].
- [32] D. A. Kosower, Phys. Rev. D 71, 045016 (2005) [arXiv:hep-ph/0311272].
- [33] F. A. Berends and W. T. Giele, Nucl. Phys. B 306, 759 (1988).
- [34] M. L. Mangano and S. J. Parke, Phys. Rept. 200, 301 (1991) [arXiv:hep-th/0509223].
- [35] L. J. Dixon, arXiv:hep-ph/9601359.
- [36] S. Catani and M. Grazzini, Nucl. Phys. B 570, 287 (2000) [arXiv:hep-ph/9908523].
- [37] G. Altarelli, R. K. Ellis and G. Martinelli, Nucl. Phys. B 157, 461 (1979).
- [38] G. Altarelli, R. K. Ellis and G. Martinelli, Nucl. Phys. B 143, 521 (1978) [Erratum-ibid. B 146, 544 (1978)].
- [39] G. Altarelli, R. K. Ellis and G. Martinelli, Z. Phys. C 27, 617 (1985).

- [40] G. Altarelli, R. K. Ellis, M. Greco and G. Martinelli, Nucl. Phys. B 246, 12 (1984).
- [41] P. W. Higgs, Phys. Lett. 12, 132 (1964).
- [42] P. W. Higgs, Phys. Rev. Lett. 13, 508 (1964).
- [43] F. Englert and R. Brout, Phys. Rev. Lett. 13, 321 (1964).
- [44] F. Wilczek, Phys. Rev. Lett. 39, 1304 (1977).
- [45] J. R. Ellis, M. K. Gaillard, D. V. Nanopoulos and C. T. Sachrajda, Phys. Lett. B 83, 339 (1979).
- [46] H. M. Georgi, S. L. Glashow, M. E. Machacek and D. V. Nanopoulos, Phys. Rev. Lett. 40, 692 (1978).
- [47] T. G. Rizzo, Phys. Rev. D 22, 178 (1980) [Addendum-ibid. D 22, 1824 (1980)].
- [48] M. A. Shifman, A. I. Vainshtein, M. B. Voloshin and V. I. Zakharov, Sov. J. Nucl. Phys. 30, 711 (1979) [Yad. Fiz. 30, 1368 (1979)].
- [49] A. I. Vainshtein, V. I. Zakharov and M. A. Shifman, Sov. Phys. Usp. 23 (1980) 429 [Usp. Fiz. Nauk 131 (1980) 537].
- [50] M. B. Voloshin, Sov. J. Nucl. Phys. 44, 478 (1986) [Yad. Fiz. 44, 738 (1986)].
- [51] S. Dawson, Nucl. Phys. B **359**, 283 (1991).
- [52] S. Dawson and R. Kauffman, Phys. Rev. D 49, 2298 (1994) [arXiv:hep-ph/9310281].
- [53] A. Djouadi, M. Spira and P. M. Zerwas, Phys. Lett. B 264, 440 (1991).
- [54] D. Graudenz, M. Spira and P. M. Zerwas, Phys. Rev. Lett. 70, 1372 (1993).
- [55] M. Spira, A. Djouadi, D. Graudenz and P. M. Zerwas, Nucl. Phys. B 453, 17 (1995) [arXiv:hep-ph/9504378].
- [56] R. K. Ellis, I. Hinchliffe, M. Soldate and J. J. van der Bij, Nucl. Phys. B 297, 221 (1988).
- [57] I. Hinchliffe and S. F. Novaes, Phys. Rev. D 38, 3475 (1988).
- [58] J. R. Ellis, M. K. Gaillard and D. V. Nanopoulos, Nucl. Phys. B 106, 292 (1976).
- [59] M. A. Shifman, Phys. Rept. 209, 341 (1991) [Sov. Phys. Usp. 32, 289 (1989) UFNAA,157,561-598.1989)].
- [60] E. Braaten and J. P. Leveille, Phys. Rev. D 22, 715 (1980).

- [61] M. Drees and K. i. Hikasa, Phys. Lett. B 240, 455 (1990) [Erratum-ibid. B 262, 497 (1991)].
- [62] S. L. Adler and W. A. Bardeen, Phys. Rev. D 4, 3045 (1971) [Erratum-ibid. D 6, 734 (1972)].
- [63] W. E. Caswell, Phys. Rev. Lett. **33**, 244 (1974).
- [64] D. R. T. Jones, Nucl. Phys. B 75, 531 (1974).
- [65] M. Spira and P. M. Zerwas, Lect. Notes Phys. 512, 161 (1998) [arXiv:hep-ph/9803257].
- [66] M. Spira, Fortsch. Phys. 46, 203 (1998) [arXiv:hep-ph/9705337].
- [67] A. Djouadi, Phys. Rept. 457, 1 (2008) [arXiv:hep-ph/0503172].
- [68] A. Djouadi, M. Spira and P. M. Zerwas, Z. Phys. C 70, 427 (1996) [arXiv:hep-ph/9511344].
- [69] A. Gehrmann-De Ridder, T. Gehrmann and E. W. N. Glover, Phys. Lett. B 612, 49 (2005) [arXiv:hep-ph/0502110].
- [70] K. G. Chetyrkin, B. A. Kniehl and M. Steinhauser, Phys. Rev. Lett. 79, 353 (1997) [arXiv:hep-ph/9705240].
- [71] S. Catani, D. de Florian and M. Grazzini, JHEP 0105, 025 (2001) [arXiv:hep-ph/0102227].
- [72] R. V. Harlander and W. B. Kilgore, Phys. Rev. Lett. 88, 201801 (2002) [arXiv:hepph/0201206].
- [73] R. V. Harlander and W. B. Kilgore, Phys. Rev. D 64, 013015 (2001) [arXiv:hepph/0102241].
- [74] R. V. Harlander, Phys. Lett. B 492, 74 (2000) [arXiv:hep-ph/0007289].
- [75] J. F. Gunion, H. E. Haber, G. L. Kane and S. Dawson, THE HIGGS HUNTER'S GUIDE (1990).
- [76] S. L. Glashow, Nucl. Phys. 22, 579 (1961).
- [77] Salam, A., 1968, in Elementary Particle Theory, Proceedings of the 8th Nobel Symposium, edited by N. Svartholm (Almqvist and Wiksell, Stockholm).
- [78] S. Weinberg, Phys. Rev. Lett. 19, 1264 (1967).
- [79] S. L. Glashow, J. Iliopoulos and L. Maiani, Phys. Rev. D 2, 1285 (1970).
- [80] J. C. Collins, D. E. Soper and G. Sterman, Nucl. Phys. B 261, 104 (1985).

- [81] J. C. Collins, D. E. Soper and G. Sterman, Nucl. Phys. B 250, 199 (1985).
- [82] R. K. Ellis, H. Georgi, M. Machacek, H. D. Politzer and G. G. Ross, Nucl. Phys. B 152, 285 (1979).
- [83] S. Catani, Phys. Lett. B 427, 161 (1998) [arXiv:hep-ph/9802439].
- [84] S. Dittmaier and M. 1. Kramer, Phys. Rev. D 65, 073007 (2002) [arXiv:hep-ph/0109062].
- [85] W. Beenakker, S. Dittmaier, M. Kramer, B. Plumper, M. Spira and P. M. Zerwas, Nucl. Phys. B 653, 151 (2003) [arXiv:hep-ph/0211352].
- [86] T. Robens and C. H. Chung, arXiv:1001.2704 [hep-ph].
- [87] C. H. Chung, M. Kramer and T. Robens, arXiv:1012.4948 [hep-ph].
- [88] T. Robens, C. H. Chung and M. Kramer, arXiv:1105.5327 [hep-ph].
- [89] M. Czakon, M. Krämer and M. Kubocz. Work in progress.
- [90] A. Kanaki and C. G. Papadopoulos, Comput. Phys. Commun. 132, 306 (2000) [arXiv:hepph/0002082].
- [91] A. Ali, J. G. Korner, Z. Kunszt, J. Willrodt, G. Kramer, G. Schierholz and E. Pietarinen, Phys. Lett. B 82, 285 (1979).
- [92] A. Ali, J. G. Korner, Z. Kunszt, E. Pietarinen, G. Kramer, G. Schierholz and J. Willrodt, Nucl. Phys. B 167, 454 (1980).
- [93] K. Fabricius, I. Schmitt, G. Schierholz and G. Kramer, Phys. Lett. B 97, 431 (1980).
- [94] S. Catani and M. H. Seymour, Phys. Lett. B 378, 287 (1996) [arXiv:hep-ph/9602277].
- [95] S. Dittmaier, M. Kramer, Y. Liao, M. Spira and P. M. Zerwas, Phys. Lett. B 478, 247 (2000) [arXiv:hep-ph/0002035].
- [96] S. Dittmaier, M. Kramer, Y. Liao, M. Spira and P. M. Zerwas, Phys. Lett. B 441, 383 (1998) [arXiv:hep-ph/9808433].
- [97] B. A. Kniehl, F. Madricardo and M. Steinhauser, Phys. Rev. D 66, 054016 (2002) [arXiv:hep-ph/0205312].
- [98] A. Gehrmann-De Ridder, T. Gehrmann and G. Heinrich, Nucl. Phys. B **682**, 265 (2004) [arXiv:hep-ph/0311276].
- [99] T. Hahn, Comput. Phys. Commun. 168, 78 (2005) [arXiv:hep-ph/0404043].

- [100] C. Amsler et al. [Particle Data Group], Phys. Lett. B 667, 1 (2008).
- [101] J. Pumplin, D. R. Stump, J. Huston, H. L. Lai, P. M. Nadolsky and W. K. Tung, JHEP 0207, 012 (2002) [arXiv:hep-ph/0201195].
- [102] A. D. Martin, W. J. Stirling, R. S. Thorne and G. Watt, Eur. Phys. J. C 63, 189 (2009) [arXiv:0901.0002 [hep-ph]].
- [103] G. Altarelli, Partons In Quantum Chromodynamics, Phys. Rept. 81, 1 (1982).
- [104] G. Altarelli, A QCD primer, arXiv:hep-ph/0204179.
- [105] S. Willenbrock, QCD CORRECTIONS TO p anti-p → W+ + X: A CASE STUDY, BNL-43793, Jan 1990. 41pp, presented at Theoretical Adv. Summer Inst. (TASI), Boulder, CO, Jun 4-30, 1989. Published in Boulder ASI 1989:323-362.
- [106] R. K. Ellis, An Introduction to the QCD Parton Model, Lectures given at 1987 Theoretical Advanced Study Inst. in Elementary Particle Physics, Santa Fe, N. Mex., Jul 5 - Aug 1, 1987.
- [107] D. E. Soper, Basics of QCD perturbation theory, arXiv:hep-ph/0011256.
- [108] G. Dissertori, I. G. Knowles and M. Schmelling, QCD High energy experiments and theory, *Oxford, UK: Clarendon (2003) 538 p.*
- [109] R. Keith Ellis, W. James Stirling and Bryan R. Webber, QCD and Collider Physics (1996).
- [110] Vernon D. Barger and Roger J. N. Phillips, Collider Physics (1991).
- [111] T. Muta, Foundations of Quantum Chromodynamics: An Introduction to Perturbative Methods in Gauge Theories, World Scientific Lecture Notes in Physics (1998).
- [112] Valery Rubakov, Classical Theory of Gauge Fields, Princeton University Press (2002).
- [113] F. Schwabl, Advanced Quantum Mechanics, Springer (1997).
- [114] F. Mandl and G. Shaw, Quantum Field Theory, Wiley (1993).
- [115] M. Peskin and D. Schroeder, An Introduction to Quantum Field Theory, Addison Wesley (1995).
- [116] L. Ryder, Quantum Field Theory, Cambridge (1996).
- [117] Steven Weinberg, The Quantum Theory of Fields: Volume 1, Foundations.
- [118] Steven Weinberg, The Quantum Theory of Fields: Volume 2, Modern Applications.

- [119] A. Erdelyi, W. Magnus, F. Oberhettinger, F. G. Tricomi, Higher transcendental functions, McGraw-Hill Book Company (1953).
- [120] A. Erdelyi, W. Magnus, F. Oberhettinger, F. G. Tricomi, Tables of integral transforms, McGraw-Hill Book Company (1954).
- [121] O. W. Greenberg, Phys. Rev. Lett. 13, 598 (1964).
- [122] M. Y. Han and Y. Nambu, Phys. Rev. 139, B1006 (1965).

Acknowledgments

Now it is the time to acknowledge those people who have helped me throughout my whole time in Aachen...

First, I would like to thank my supervisor, Prof. Michael Krämer, for providing me an opportunity to work in his research group, and for supervising my PhD thesis. This work would not be completed without his continuous guidance and patience. I had the pleasure of working with him during this exciting adventure... I will always remember the many enlightening suggestions that he gave me and definitely I will remember him being approachable.

Next, I would like to thank Prof. Michal Czakon for correcting my thesis and for pointing out many essential points. I would also like to thank Tania Robens and Zoltan Nagy for the many valuable suggestions, discussions and help with understanding this subject; and I always enjoyed the time we spent working together. I am grateful to Michael Kubocz and Alexander Mück for the many fruitful discussions about the implementation of the codes. I would also like to thank Michael Kubocz, Alexander Mück, Ben O'Leary and Tania Robens for carefully reading the manuscript. Thanks also goes to Anke Bachtenkirch and Johannes Rohrer who have always provided a ready solution for me each time when I had to face a practical problem.

I am further grateful to all the colleagues of the Institute for Theoretical Particle Physics and Cosmology at RWTH Aachen during the last four and half years, for having established a pleasant and encouraging atmosphere from which I have benefitted a lot. I will also remember the people I met during my stay at DESY Hamburg, with whom I spent two pleasant months.

Finally, I am greatly indebted to my parents for their unconditional support and continuous encouragement throughout my PhD studies. I dedicate this thesis to them.

Cheng Han Chung

Institut für Theoretische Physik RWTH Aachen D - 52056 Aachen, Germany
 Phone:
 +49 241 80 27025

 Fax:
 +49 241 80 22187

 Email:
 chenghan@physik.rwth-aachen.de

Personal

Born on October 17, 1976.

Taiwan Citizen.

Languages

Chinese (mother tongue), English (advanced), German (basic knowledge).

Education

B.S. Physics, Chung Yuan University, Taiwan 1997 - 2000.

M.A. Physics, National Tsing Hua University, Taiwan 2000 - 2002.

Ph.D. Physics, RWTH Aachen, Germany 10/2006 - 03/2011. [Date of examination 21 June 2011]

Scholarship

Graduate College: Elementary Particle Physics at the TeV Scale, RWTH Aachen 10/2006 - 06/2009. Full financial support at DESY Hamburg 11/2009 - 12/2009.

Employment

Chung Yuan University 08/2004 - 07/2006.

RWTH Aachen 07/2009 - 03/2010.

RWTH Aachen 03/2011 - 04/2011.

RWTH Aachen 09/2011 - 11/2011.